# CHARACTERISATION OF THE MOLECULAR BASIS OF PROTEIN S ANTICOAGULANT FUNCTION

A thesis submitted to Imperial College London for the degree of Doctor of Philosophy in the Faculty of Medicine

by

## Helena Maria Andersson, BSc, MSc

October 2010

Centre for Haematology Faculty of Medicine Imperial College London Hammersmith Hospital Campus The copyright of this thesis rests with the author and no quotation from it or information derived from it may be published without the prior written consent of the author.

## ABSTRACT

Protein S has an established role in the protein C anticoagulant pathway as a cofactor for anticoagulant protein C (APC) and has also recently been shown to serve as a cofactor enhancing the anticoagulant activity of tissue factor pathway inhibitor (TFPI). Despite its physiological role and clinical importance, the molecular bases of its functions are not fully understood. The aim of my thesis was to clarify the molecular mechanisms involved in the protein S interaction with APC and TFPI. More than 30 point or composite protein S variants were constructed and analysed during this project. These variants spanned the Gla, thrombin sensitive region (TSR), epidermal growth-factor1 (EGF1) and EGF2 domains of protein S. Protein S was expressed in mammalian cells and was purified by chromatography, as required. Protein S was characterised by size, cleavage, multimerisation, y-carboxylation of the Gla domain, binding to phospholipids and to domain specific monoclonal antibodies. Variants were evaluated for their APC and TFPI cofactor activities both by calibrated automated thrombography and in purified FVa inactivation or FXa inhibition assays, respectively. The protein S variant, protein S D95A, with substitution in EGF1 was found to be largely devoid of functional APC cofactor activity and I believe that this residue plays an important role in protein S anticoagulant function. It was ycarboxylated and bound phospholipids and domain specific monoclonal antibodies with an apparent dissociation constant similar to that of wild type protein S. Importantly, protein S D95A enhanced the anticoagulant activity of TFPI, suggesting that distinct residues in protein S mediate its APC and TFPI cofactor activity. Two composite mutants in the protein S EGF1 domain had partially reduced TFPI cofactor activity in plasma. However, none of the more than 30 variants spanning the Gla-TSR-EGF1-EGF2 domain of protein S completely disrupted the protein S cofactor activity towards TFPI. Collectively, these results shed light on the molecular basis of protein S cofactor function and suggest distinct residues in protein S are involved in the binding to APC and to TFPI.

## **DECLARATION OF ORIGINALITY**

I, Helena Maria Andersson, hereby declare that the work presented in this thesis is my own. All work and data analysis of the results were performed by myself, unless otherwise specified in the text.

Helena Maria Andersson

## ACKNOWLEDGEMENT

First of all I would like to thank my supervisors Professor David Lane and Dr Jim Crawley for all their support and help in the last three years, for always having time to discuss things and for allowing me to explore my own ideas. It has been a priviledge to be part of their lab. I would like to thank Professor David Lane, in particular, for critically reading this thesis in such a short time and for constantly supporting my professional development.

I have been very lucky to have the opportunity to collaborate with Dr Suely Rezende. It has been a pleasure to know her and to work with her in the last two years. I thank her for her help and support.

I would like to thank everyone in the Wallenberg Lab in Malmö for their kindness during the two very busy weeks I was there. In particular, I would like to thank Professor Björn Dahlbäck, Sinh and Sofia for their help and advice.

It goes without saying that without all the guys in my lab, my PhD experience wouldn't have been the same. Thanks for all your support, help, jokes and coffee breaks. Agata, thank you for your friendship, for letting me know through that indistinguishable high pitch ultrasound noise that there was someone else, in addition to me, who was struggeling a few benches down the lab, for dragging me to yoga and for all our fun trips. Josefin, thanks for your friendship, help, invaluable scientific discussions and for some excellent team work. Brenda, thanks for your friendship and advice during these three years. Josefin, Brenda and Isabelle, thank you for your support, lunch breaks and girls night out. Tom, you wonderful excuse of a human being, thanks for being supportive and always cracking a joke in the moments of need. Verity and Maria, thanks for your friendship, all the chats and coffee breaks.

Last, but not least, I would like to thank my family for always being so supportive. Tack så mycket för allt ni gjort för mig. Giorgio, thanks for always being there for me and for the last wonderful years spent together. Grazie di tutto amore mio. INDEX

ABSTRACT	3
DECLARATION OF ORIGINALITY	4
ACKNOWLEDGEMENT	5
FIGURE LEGEND	9
TABLE LEGEND	12
ABBREVIATIONS	13
1 INTRODUCTION	17
<ul> <li>Haemostasis</li> <li>1.1.1 Primary haemostasis</li> <li>1.1.2 Secondary haemostasis</li> <li>1.1.2.1 Initiation of coagulation</li> <li>1.1.2.2 Feedback activation</li> <li>1.1.3 Regulation of coagulation by anticoagulant pathways</li> <li>1.1.4 Fibrinolysis</li> <li>1.1.5 The haemostatic balance</li> </ul>	<b>17</b> 18 18 19 20 22 23
1.2       TFPI         1.2.1       TFPI gene         1.2.2       TFPI structure         1.2.3       TFPI function         1.2.3.1       TFPI mediated inhibition of FXa         1.2.3.2       TFPI mediated inhibition of TF/FVIIa	<b>23</b> 23 24 26 26 27
<ul> <li>1.3 Protein C/APC</li> <li>1.3.1 PROC</li> <li>1.3.2 Protein C structure</li> <li>1.3.3 Protein C activation</li> <li>1.3.4 APC function</li> <li>1.3.4.1 APC mediated inactivation of FVa</li> <li>1.3.4.2 APC mediated inactivation of FVIIIa</li> </ul> 1.4 Protein S	28 29 30 32 32 33 <b>34</b>
<ul> <li>1.4.1 PROS1</li> <li>1.4.2 Protein S structure</li> <li>1.4.3 Protein S functions</li> <li>1.4.3.1 Protein S enhances the anticoagulant activity of APC</li> <li>1.4.3.1.1 Protein S enhancement of APC mediated inactivation of FVa</li> <li>1.4.3.1.2 Protein S enhancement of APC mediated inactivation of FVIIIa</li> <li>1.4.3.2 Protein S enhancement of TFPI mediated inactivation of FXa</li> <li>1.4.3.2.1 Protein S enhancement of TFPI mediated inactivation of FXa</li> <li>1.4.3.2.2 Protein S enhancement of TFPI mediated inactivation of TF/FVIIa</li> </ul>	34 35 37 38 38 40 40 40 40 41
<ul> <li>1.5 Domains of protein S which mediate functional interactions</li> <li>1.5.1 Domains of protein S reported to be important for APC cofactor activity</li> <li>1.5.2 Domains of protein S that are potentially important for TFPI cofactor activity</li> </ul>	<b>42</b> 42 43
1.6 Aims of my thesis	43

<u>2 ME</u>	THODOLOGY	44
2.1 0	Seneration of protein S mutant expression vectors	45
2.1.1	Mammalian protein S expression vector and composite mutants generated	45
2.1.2	Generation of protein S mutants by site-directed mutagenesis	46
2.1.3	Transformation of competent cells	47
2.1.4	Generation of glycerol stocks	47
2.1.5	Plasmid miniprep and sequencing	48
2.1.6	Subcloning of the protein S mutants into the pcDNA6/Protein S vector	48
2.1.7	DNA ligation	49
2.1.8	Transformation of competent cells with subcloned mutants	50
2.1.9	DNA maxiprep/megaprep and quantification	50
2.2 E	xpression of protein S	50
2.2.1	Mammalian cell culture	50
2.2.2	Transient transfection of HEK293T cells	51
2.2.3	Stable transfection of HEK293 cells	51
2.2.4	Cryopreservation of mammalian cells	52
2.2.5	Expression and narvesting of WT and variant protein S	52
2.3 P	Protein S purification	53
2.3.1	Protein S purification by FPLC on an anion exchange QFF column and by	
	immunoaffinity chromatography	53
2.3.1	1.1 Partial purification of protein S by FPLC on an anion exchange QFF column	54
2.3.1	1.2 Preparation of the immunoattinity column 1.2 Purification of protoin S by EPLC on an immunoaffinity column	54 54
2.3.	Protein S purification by barium citrate precipitation and by EPI C on an anion	54
2.0.2	exchange DEAE column	55
2.3.2	2.1 Barium citrate precipitation of protein S	55
2.3.2	2.2 Purification of protein S by FPLC on an anion exchange DEAE column	56
2.4 A	nalysis and characterisation of protein S	56
2.4.1	PNGase digestion	56
2.4.2	SDS-PAGE and Native PAGE	56
2.4.3	Silver staining	57
2.4.4	Western Blot	57
2.4.5	Quantification of total protein	58
2.4.6	In-house ELISA with an in-house detection antibody	58
2.4.7	In-house ELISA with commercially available antibodies	59
2.4.8	Preparation of phospholipid vescicles	59
2.4.9	Binding of protein S to phospholipids	60
2.4.10	Binding of protein S to domain specific monocional antibodies	01
2.5 C	AT assay	61
2.5.1	Overview of CAT	61
2.5.2	Evaluation of the protein S-APC cofactor activity	62
2.5.3	Evaluation of the protein S-TFPI cofactor activity	63
2.6 F	Va inactivation assay	63
2.6.1	Overview of the FVa inactivation assay	63
2.6.2	FV activation	64
2.6.3	FVa inactivation	64
2.6.4	Prothrombinase assay	65
2.7 FXa inhibition assav 65		65
2.7.1	Overview of the FXa inhibition assay	65
2.7.2	FXa inhibition assay	66
2.7.3	Kinetic data analysis of FXa inhibition by TFPI in the presence and absence of	
	protein S	66

<u>3 RESULTS</u>	68
<b>3.1 Mutagenesis, expression and purification of protein S</b> 3.1.1 Selection and generation of protein S variants	<b>68</b> 68 71
3.1.3 Purification of protein S by FPLC on an anion exchange QFF column and immunoaffinity column	72
3.1.4 Purification of protein S by barium citrate precipitation and by FPLC on an anion exchange DEAE column	76
<ul> <li>3.2 Characterisation of recombinant protein S</li> <li>3.2.1 Quantification of protein S by ELISA</li> <li>3.2.2 Analysis of the purity of protein S</li> </ul>	<b>79</b> 79 80
<ul><li>3.2.3 Multimerisation of protein S</li><li>3.2.4 Cleavage of protein S</li></ul>	81 82
<ul> <li><b>3.3</b> Evaluation of APC cofactor activity of protein S</li> <li>3.3.1 Preliminary experiments in protein S deficient plasma</li> <li>3.3.2 Evaluation of APC cofactor activity of protein S by CAT</li> <li>3.3.3 Binding of protein S to phospholipid surfaces</li> <li>3.3.4 Binding of protein S to domain specific monoclonal antibodies</li> <li>3.3.5 Protein S enhancement of APC mediated cleavage of FVa at Arg306</li> <li>3.3.6 Discussion</li> </ul>	<b>84</b> 86 94 95 96 98
<ul> <li><b>3.4 Evaluation of TFPI cofactor activity of protein S</b></li> <li>3.4.1 Evaluation of TFPI cofactor activity of protein S by CAT</li> <li>3.4.2 Evaluation of TFPI cofactor activity of protein S in the FXa inhibition assays</li> <li>3.4.2.1 Evaluation of FXa inhibition by WT purified TFPI and protein S</li> <li>3.4.2.2 Evaluation of domains and amino acids of protein S involved in TFPI cofactor activity by use of domain specific monoclonal antibodies and protein S variants</li> <li>3.4.2.3 Kinetic analysis of TFPI mediated inhibition of FXa in the presence and absence of</li> </ul>	<b>106</b> 106 120 120 124
protein S 3.4.3 Discussion	127 133
4 CONCLUDING REMARKS	<u>140</u>
4.1 Protein S cleavage, multimerisation and its binding to phospholipids	140
4.2 APC cofactor activity of protein S; interaction sites and a suggested mechanism	142
4.3 TFPI cofactor activity of protein S; possible interaction sites and mechanism	146
4.4 Protein S direct activity	148
4.5 <i>In vivo</i> regulation of coagulation by protein S	149
APPENDIX	153
REFERENCES	155
PUBLICATIONS ARISING FROM THIS WORK	169
ABSTRACTS LEADING TO ORAL PRESENTATION	169

## FIGURE LEGEND

Figure 1.1 Initiation of the coagulation cascade.	19
Figure 1.2 Propagation of the coagulation cascade.	20
Figure 1.3 Regulation of the coagulation cascade by the anticoagulant pathways.	22
Figure 1.4 Schematic representation of full-length TFPI.	24
Figure 1.5 Distribution of TFPI.	25
Figure 1.6 TFPI binds to and inhibits FXa and TF/FVIIa.	28
Figure 1.7 Schematic structure of protein C.	30
Figure 1.8 Activation of protein C.	31
Figure 1.9 Activation of FV by thrombin and inactivation of FVa by APC.	32
Figure 1.10 Schematic representation of protein S and its domains.	37
Figure 2.1 Overview of the Methodology of this thesis.	44
Figure 2.2 Protein S expression vector.	45
Figure 3.1 Purification of WT protein S by FPLC on a Sepharose HiTrap QFF anion	
exchange column followed by an immunoaffinity column and analysis of elution fraction	ons.
	74
Figure 3.2 Western blot analysis of protein S in the fractions obtained during purificat	ion
on the Sepharose HiTrap QFF and on the immunoaffinity column.	75
Figure 3.3 Purification of barium citrate precipitated WT protein S on an anion exchan	nge
DEAE column and analysis by SDS-PAGE.	78
Figure 3.4 Standard curves of protein S used in the ELISA.	79
Figure 3.5 Silver staining of purified protein S samples.	81
Figure 3.6 Western blot of a native PAGE analysis to identify multimer formation	
occurring during the purification.	82
Figure 3.7 Analysis of protein S cleavage.	83
Figure 3.8 TF initiation of coagulation in protein S deficient plasma.	84
Figure 3.9 Activity of endogenous APC and TFPI in protein S deficient plasma.	85
Figure 3.10 APC and protein S anticoagulant activity in plasma.	87
Figure 3.11 APC cofactor activity of different preparations of protein S in plasma.	88
Figure 3.12 Screening of protein S variants for APC cofactor activity.	89
Figure 3.13 APC cofactor activity in plasma of WT protein S, Face1 and Face2 variar	nts.
	90
Figure 3.14 APC cofactor activity in plasma of WT protein S, protein S D95A, D95N,	
D78A and Q79A variants.	93
Figure 3.15 Binding of protein S to phospholipid surfaces.	94
Figure 3.16 Protein S enhancement of APC mediated cleavage of FVa at Arg306.	97

Figure 3.17 Location of residues important for species specificity in relation to Asp95	ò,
Asp78 and GIn79.	100
Figure 3.18 Conservation of Asp78, GIn79 and Asp95 across species.	101
Figure 3.19 Location of Asp78, Gln79, Asp95 and Face2 within the protein S Gla-TS	R-
EGF1 model.	101
Figure 3.20 EGF domains of protein S and binding to calcium.	102
Figure 3.21 Location of residues found to have reduced APC cofactor activity.	105
Figure 3.22 TFPI anticoagulant activity in the presence or absence of protein S in	
plasma.	107
Figure 3.23 Influence on thrombin generation of polyclonal antibodies against TFPI a	and
against protein S.	108
Figure 3.24 TFPI and protein S anticoagulant activity in plasma in the presence and	
absence of polyclonal antibodies against protein C.	109
Figure 3.25 Thrombin generation in the presence of TFPI and WT protein S, protein	S
Face2, D95A, D95N, D78A or Q79A.	111
Figure 3.26 Change (%) in peak height and lag time of WT protein S, protein S Face	2,
D95A, D95N, D78A and Q79A in comparison with the control.	112
Figure 3.27 Thrombin generation in the presence of TFPI and WT protein S and prot	tein
S EGF1 and EGF2 point variants.	113
Figure 3.28 Thrombin generation in the presence of TFPI and WT protein S and prot	tein
S EGF1 and EGF2 point variants.	114
Figure 3.29 Thrombin generation in the presence of purified TFPI and TFPI in	
concentrated conditioned media in the presence and absence of protein S.	115
Figure 3.30 Protein S enhancement of TFPI at different concentrations of TFPI.	116
Figure 3.31 Thrombin generation in the presence of TFPI and WT protein S and prot	tein
S variants.	118
Figure 3.32 WT protein S, protein S N74A, KTK, NEDM and NNEDM studied by	
thrombin generation in the presence of TFPI or APC.	120
Figure 3.33 Dose-dependent inhibition of FXa by TFPI.	121
Figure 3.34 Dose-dependent enhancement of TFPI activity by protein S.	122
Figure 3.35 Inhibition of FXa by TFPI and protein S in the absence and presence of	
phospholipids.	123
Figure 3.36 Inhibition of protein S cofactor activity towards TFPI by monoclonal domain	ain
specific antibodies.	125
Figure 3.37 TFPI inhibition of FXa in the presence and absence of WT protein S, pro	otein
S Face2 and protein S TSR.	126

Figure 3.38 TFPI inhibition of FXa in the presence and absence of WT protein S, pro	tein
S D95A, protein S N74A, protein S KTK and protein S NNEDM.	127
Figure 3.39 Determination of the $K_m$ of FXa for the chromogenic substrate S-2765.	128
Figure 3.40 FXa inhibition assay started by the addition of FXa or by the addition of	ΓFPI
and S-2765.	129
Figure 3.41 FXa inhibition by TFPI in the presence and absence of WT protein S.	130
Figure 3.42 Change in $V_0$ at increasing concentrations of TFPI in the presence or	
absence of protein S.	131
Figure 3.43 Residue substitutions in the Gla-TSR-EGF1 domains made during this	
project.	139
Figure 4.1 Location of residues in protein S and protein C shown to be important for	APC
cofactor activity of protein S.	146

## TABLE LEGEND

Table 2.1 PCR cycle parameters for Pfu ultra DNA polymerase.	46
Table 2.2 PCR cycle parameters for KOD hot start DNA polymerase.	47
Table 2.3 Enzymatic digestion of WT and mutated pcDNA6/PS vector.	48
Table 2.4 Dephosphorylation of the linearised WT vector.	49
Table 2.5 Ligation of the mutated protein S region into the WT protein S vector.	49
Table 2.6 Composition of silver staining buffers.	57
Table 3.1 Protein S variants produced during this thesis.	69
Table 3.2 N-terminal amino acid sequence of protein S, prothrombin and protein S	
variants.	70
Table 3.3 Binding of protein S to phospholipids and domain specific monoclonal	
antibodies.	95
Table 3.4 Inhibition constants for TFPI inhibition of FXa in the presence or absence	of
protein S.	133
Table 3.5 TFPI cofactor activity of protein S variants in comparison with WT protein	S in
plasma studied by CAT assay.	136

## ABBREVIATIONS

APC	activated protein C
AT	antithromin
AP	activation peptide
Вр	base pair
BSA	bovine serum albumin
C4BP	C4b-binding protein
CAT	calibrated automated thrombography
cDNA	complimentary deoxyribonucleic acid
CAT	calibrated automated thrombography
CIAP	calf intestinal alkaline phosphatase
CMV	cytomegalovirus
СТІ	corn trypsin inhibitor
Ctrl	control
CV	coefficient of variation
DEAE	diethylaminoethyl cellulose anion exchange
DMSO	dimethyl sulfoxide
DNA	deoxyribonucleic acid
dNTP	deoxynucleoside triphosphate
DOPC	1,2-Dioleoyl-sn-glycero-3-phosphocholine
DOPE	1,2-Dioleoyl-sn-glycero-3-phosphoethanolamine
DOPS	1,2-Dioleoyl-sn-glycero-3-phosphoserine
ECM	subendothelial extracellular matrix
EDTA	ethylenediaminetetraacetic acid
EGF	epidermal growth factor
ELISA	enzyme-linked immunosorbent assay
EPCR	endothelial protein C receptor
ER	endoplasmic reticulum
ETP	endogenous thrombin potential
Eq	equation
FCS	foetal calf serum
FPLC	fast protein liquid chromatography
FRET	fluorescence resonance energy transfer
GAG	glycosaminoglycans
Gla	γ-carboxylated glutamic acid
GPI	glycosyl phosphatidylinositol

HEPES	N-2-hydroxyethylpiperazine-N'-2-ethanesulfonic acid
HEK	human embryonic kidney fibroblast
HN	25 mM HEPES pH 7.7, 150 mM NaCl
HNBSACa <sup>2+</sup>	HN, 5 mM CaCl <sub>2</sub> , 5 mg/ml BSA
HRP	horseradish peroxidise
Hyn	β-hydroxyasparagine
К	Kunitz
Kb	kilobase
kDa	kilodaltons
K <sub>d</sub>	dissociation constant
K <sub>d(app)</sub>	apparent dissociation constant
K <sub>i</sub>	initial inhibition constant
K <sub>i</sub> *	final inhibition constant
K <sub>m</sub>	Michaelis-Menten constant
Kobs	apparent rate constant for transition from $V_0$ to $V_s$
LB	Luria-Bertani medium
LDL	low-density lipoproteins
LG	laminin G
mA	milli absorbance
mAU	milli absorbance unit
MEM	minimum essential media
mRNA	messanger ribonucleic acid
MW	molecular weight
OPD	o-phenylenediamine dihydrochloride
PAI	plasminogen activator inhibitor
PBS	phosphate buffered saline
PC	L-a-phosphatidylcholine
PCR	polymerase chain reaction
PE	L-α-phosphatidylethanolamine
PEI	linear polyethylenimine
PS	L-α-phosphatidylserine
QFF	Q sepharose fast flow
SDS-PAGE	sodium dodecyl sulphate polyacrylamide gel
	electrophoresis
SHBG	sex hormone binding globulin
SPR	surface plasmon resonance
SUV	small unilamellar vesicles

TAFI	thrombin activatable fibrinolysis inhibitor
TBE	tris borate EDTA
TF	tissue factor
TFF	tangential flow filtration
TFPI	tissue factor pathway inhibitor
ТМ	thrombomodulin
t-PA	tissue-type plasminogen activator
TSR	thrombin sensitive region
u-PA	urinary plasminogen activator
VWF	von Willebrand factor
V <sub>0</sub>	initial velocity of S-2765 conversion
Vs	final velocity of S-2765 conversion
WT	wild type
Å	Ångström

## Comon abbreviations not spelled out in the text

°C	Celcius degrees
g	gram
I	liter
m	meter
Μ	molar
mol	mole
SD	standard deviation
U	units
V	volt
%	percentage

## Power prefix abbreviations

m	milli
μ	micro
n	nano
р	pico
f	femto

## Amino acid abbreviations

Alanine	Ala	А
Arginine	Arg	R
Asparagine	Asn	Ν
Aspartic acid	Asp	D
Cysteine	Cys	С
Glutamic acid	Glu	Е
Glutamine	Gln	Q
Glycine	Gly	G
Histidine	His	Н
Isoleucine	lle	Ι
Leucine	Leu	L
Lysine	Lys	K
Methionine	Met	М
Phenylalanine	Phe	F
Proline	Pro	Р
Serine	Ser	S
Threonine	Thr	Т
Tryptophan	Trp	W
Tyrosine	Tyr	Y
Valine	Val	V

## **1 INTRODUCTION**

## 1.1 Haemostasis

The cardiovascular system has a vital importance in distributing nutrients and oxygen, while also, removing toxic metabolites. It maintains blood homeostasis by regulating variables such as body temperature, pH, pressure and the concentration of transported molecules.

The integrity of the cardiovascular system is preserved by a highly regulated process called haemostasis. Under normal circumstances, circulating blood is maintained fluid by the physiological anticoagulants acting at the surface of endothelial cells and by the lack of procoagulant stimuli. In case of injury of the blood vessel, there are three major phases that serve to prevent blood loss and enable vessel repair which are all regulated both spatially and temporally, primary haemostasis (or platelet plug formation), secondary haemostasis (or coagulation) and fibrinolysis. Balanced haemostasis results in rapid occlusion of the damaged tissue avoiding bleeding, infiltration of pathogens in the blood and allowing enough time to the injured vasculature to repair. The main components regulating haemostasis are blood vessels, platelets and coagulation factors. These components cooperate, through multiple positive and negative feedback mechanisms, to maintain circulating blood fluid and to stop bleeding in case of injury.

## 1.1.1 Primary haemostasis

Primary haemostasis is the initial response to vascular injury. Following damage of the endothelial cell layer the subendothelial extracellular matrix (ECM) is exposed to the bloodstream. This results in the lack of the anticoagulant endothelial surface and exposure of the adhesive macromolecules of the ECM such as collagen, fibronectin, laminin, thrombospondin and vitronectin. While platelets are able to interact directly with the ECM, this interaction is greatly enhanced by von Willebrand Factor (VWF).<sup>1</sup> VWF is a large multimeric adhesive glycoprotein found in plasma and is secreted by endothelial cells and activated platelets.<sup>2-4</sup> When collagen is exposed to the bloodstream VWF binds to it, and

shear stress induces unravelling of the globular VWF, making its A1 domain accessible to platelets.<sup>5</sup> Circulating platelets bind loosely to VWF through the receptor complex GPIb-V-IX, causing the platelets to roll along the VWF multimer.<sup>6</sup> This rolling enables platelets to come into contact with exposed collagen and to bind to it through their GPVI receptor. Binding of collagen induces intracellular signalling that augments the affinity of cell surface receptors termed integrins for their ligands, causing adherence, aggregation and activation of platelets.<sup>7</sup> Activated platelets release additional VWF resulting in a positive feedback loop that leads ultimately to the formation of a primary loose platelet plug.<sup>8</sup> During activation, the platelet membrane composition is altered exposing negatively charged phospholipids, such as phosphatidylserine, that are confined to the inner phospholipid layer in the inactivated cell.<sup>9</sup> This serves as a surface for the activation and assembly of coagulation factors.

### 1.1.2 Secondary haemostasis

Secondary haemostasis involves coagulation and requires activation of multiple zymogens to active serine proteases, as well as activation of cofactors. The coagulation cascade can be initiated through two different pathways, the contact activation pathway and the tissue factor (TF) pathway. <sup>10</sup> While the physiological relevance of the contact pathway is still debated, the TF pathway is now recognised as the principal physiological initiator of coagulation. The following paragraphs will only consider the TF pathway. Once coagulation is triggered there is a first stage of initiation of coagulation producing only traces of thrombin, which through positive feedback activation leads to propagation of coagulation resulting in more thrombin and the formation of an insoluble fibrin clot.<sup>10</sup>

### 1.1.2.1 Initiation of coagulation

The major initiator of coagulation is TF, a transmembrane protein expressed on the surface of cells at extravascular sites. Following vessel damage, TF is exposed to the blood.<sup>10</sup> While clotting factors with an enzymatic activity usually circulate as inactive zymogens, a small amount (~1%) of factor VII (FVII) circulates in its partially active form, FVIIa.<sup>11</sup> Both FVII and FVIIa are able to bind to TF, but only the TF/FVIIa complex activates FIX and FX (Figure 1.1). In the absence of its cofactor, FVa, FXa converts only a small amount of prothrombin into thrombin.



**Figure 1.1 Initiation of the coagulation cascade.** Circulating FVIIa binds to TF exposed to the bloodstream. The TF/FVIIa complex activates FX and FIX. FXa generates trace amounts of thrombin (FIIa).

#### 1.1.2.2 Feedback activation

The small amount of thrombin that is generated during the initiation phase produces a positive feedback reaction by activating FVIII and FV<sup>12</sup>, the cofactors of FIXa and FXa, respectively (Figure 1.2). The FIXa/FVIIIa complex assembled upon phospholipids is also termed the tenase complex as it activates FX. This leads to an amplification of FXa production bypassing the requirement for TF/FVIIa activity. The FXa/FVa complex on phospholipids, or the prothrombinase complex, converts prothrombin into thrombin up to 278000 times more efficiently than FXa alone.<sup>13-15</sup> Once the damaged tissue and subendothelial cells start to be covered by the platelet plug, it has been proposed that secondary haemostasis is allowed to continue by the adherence of alternatively spliced soluble TF or circulating microparticles containing TF to the surface of activated platelets.<sup>16-18</sup> This results in the propagation of both primary and secondary haemostasis. Assembly of the functional tenase and prothrombinase complex upon negatively charged phospholipids (such as those exposed on activated platelets) depends on domain-specific phospholipid interaction sites such as those provided by the Gla domain, of prothrombin, FVII, FIX, and FX or the C domains of FV and FVIII.<sup>19-22</sup> In addition to spatially confining these complexes, it also results in accelerated formation of the proteases FXa and thrombin.

Once feedback activation of coagulation has taken place, sufficient thrombin is generated for efficiently conversion of fibrinogen into fibrin, (Figure 1.2) which is cross-linked by FXIIIa to form an insoluble fibrin clot.<sup>23</sup> The fibrin clot prevents the loss of blood, the entrance of pathogens into the blood stream and allows the damaged endothelium and subendothelial layers to repair.

Thrombin plays a central role in regulating haemostasis, acting both as a procoagulant enzyme by generating a feedback activation of itself, and as an anticoagulant enzyme by proteolytically activating protein C.<sup>24</sup>



#### Figure 1.2 Propagation of the coagulation cascade.

Thrombin (FIIa) generated during the initiation stage excerts a feedback mechanism activating FVa and FVIIIa, the cofactors of FXa and FIXa respectively. This leads to a massive generation of thrombin which converts fibrinogen into fibrin.

### 1.1.3 Regulation of coagulation by anticoagulant pathways

The procoagulant pathway is tightly regulated by three major anticoagulant pathways, the tissue factor pathway inhibitor (TFPI) pathway, the protein C pathway and the antithrombin (AT) pathway (Figure 1.3).<sup>25</sup> These three pathways are each critical for the regulation of the haemostatic plug. Each pathway targets quite different components of the coagulation cascade regulating haemostasis

both temporally and spatially.

TFPI is a Kunitz type serine protease inhibitor. TFPI inhibits the extrinsic coagulation pathway by inhibiting TF/FVIIa and FXa on the subendothelial cells and on the growing platelet plug. TFPI targets the initiation of coagulation, as once enough thrombin is formed the necessity of TF/FVIIa for generating FXa is by-passed by the FIXa/FVIIIa complex.<sup>26</sup>

The activation of protein C by thrombin complexed to thrombomodulin (TM) on surrounding intact endothelial cells enables activated protein C (APC) to act at the edge of the plug formation inhibiting expansion of fibrin deposition, by inactivating FVa and FVIIIa. As activation of protein C requires the formation of thrombin, APC is probably only generated during the propagation phase.<sup>27</sup> The anticoagulant activity of APC is greatly enhanced by a cofactor, protein S, the central focus of this thesis.

AT is a serine protease inhibitor circulating at a plasma concentration of ~2-5  $\mu$ M.<sup>28</sup> It inhibits multiple serine proteases in the TF pathway (thrombin, FXa, FIXa and FVIIa) by forming a stable 1:1 stoichiometric complex, trapping them in a form in which their active site is not accessible to their usual substrate.<sup>29</sup> The inhibitory activity of AT is greatly enhanced by its binding to heparan sulphate on intact endothelial cells surrounding the tissue damage. Binding of AT to heparan sulphates also results in a conformational change of the former, increasing greatly the rate of inactivation of FXa, FIXa and FVIIa.<sup>30,31</sup> AT efficiently inhibits all of the free serine proteases that escape the immediate proximity of plug formation, assuring the latter remains localised throughout the haemostatic response.<sup>32</sup>



**Figure 1.3 Regulation of the coagulation cascade by the anticoagulant pathways.** The coagulation cascade is tightly regulated by the three main anticoagulant pathways, TFPI, APC and AT. Protein S (PS) acts as a cofactor both for TFPI and APC. Protein S/TFPI inhibits TF/FVIIa and FXa. The protein S/APC complex inactivates FVa and FVIIIa. AT inhibits multiple serine proteases. For simplicity in this diagram only inhibition of thrombin (FIIa) is shown.

### 1.1.4 Fibrinolysis

Fibrinolysis, the final phase of haemostasis, is the process by which the fibrin clot is remodelled and dissolved once the damaged blood vessel has been repaired. Tissue-type plasminogen activator (t-PA) and urinary plasminogen activator (u-PA) are the two major activators of plasminogen.<sup>33,34</sup> Plasminogen is activated to plasmin, the enzyme which degrades fibrin. Under normal circumstances where there is no fibrin, t-PA and plasminogen circulate in the blood. As soon as fibrin is formed, it acts as a template bringing t-PA and plasminogen into proximity, enhancing plasminogen activation ~1000-fold. These proteases are counterbalanced by fibrinolysis inhibitors such as thrombin activatable fibrinolysis inhibitor (TAFI), plasminogen activator inhibitor (PAI)-1 and PAI-2.33,34

#### 1.1.5 The haemostatic balance

Haemostasis represents a fine balance with respect to procoagulant and anticoagulant factors. Coagulation disorders result either in haemorrhage or in thrombosis. Inherited deficiencies in procoagulant factors result in bleeding disorders, the most commonly known are von Willebrand Disease (VWF deficiency), haemophilia A (FVIII deficiency) and haemophilia B (FIX deficiency).<sup>35,36</sup> In contrast, inherited deficiencies in anticoagulant factors result in an increased risk of thrombosis, the most commonly known are AT, protein C and protein S deficiencies as well as FV Leiden (FV resistance to inactivation by activated protein C).<sup>37</sup> Because my thesis is primarily concerned with anticoagulant pathways, I will describe these in more detail in what follows.

## 1.2 <u>TFPI</u>

### 1.2.1 TFPI gene

TFPI is expressed mainly by endothelial cells, but also by vascular smooth muscle cells, megakaryocytes, monocytes, mesangial cells, fibroblasts and cardiomyocytes.<sup>38-40</sup> The human *TFPI* gene is localized on chromosome 2 where it spans approximately 70 kilobase (kb) and consists of 9 exons.<sup>41</sup> The TFPI transcript is alternatively spliced giving origin to two isoforms, TFPI- $\alpha$  and TFPI- $\beta$ . The physiological role of TFPI- $\beta$  has not yet been determined and most results have been derived from in vitro experiments. It is, however, known that TFPI-B contains an alternatively spliced C-terminal domain and associates to endothelial cells through glycosyl phosphatidylinositol (GPI)-anchors, possibly within caveolae.<sup>42</sup> Interestingly, it has been shown to still exert an anticoagulant function by inhibiting FXa and TF/FVIIa.<sup>43</sup> TFPI- $\alpha$  is the most abundant transcript and the one on which my project will focus.<sup>24</sup> TFPI- $\alpha$  is translated into an unprocessed precursor of 304 amino acids with a predicted molecular weight (MW) of 35 kilodaltons (kDa). Cleavage of the 28 amino acid signal peptide and posttranslational modifications, result in mature full-length TFPI protein of 276 amino acids and a MW of 43 kDa. If not indicated otherwise in this thesis the term TFPI will hereafter refer to the  $\alpha$ -isoform.

### 1.2.2 TFPI structure

The full-length TFPI protein derived from the TFPI-  $\alpha$  transcript has a negatively charged N-terminal region (amino acids 1-12), 3 Kunitz domains (K1, amino acids 13-78; K2, amino acids 92-150; K3, amino acids 182-241) separated by 2 linker regions and a positively charged C-terminal region (amino acids 242-273) (Figure 1.4).44 It is N-linked glycosylated at Asn117 and Asn167 and O-linked glycosylated at Ser174 and Thr175. Truncated forms of TFPI derived from the TFPI- α transcript are also present in plasma. Truncation usually results in TFPI variants lacking their C-terminal tail and often also part of their K3 domain.<sup>45</sup> The physiological origin of these truncated forms is not known but it has been shown in vitro that TFPI can be cleaved by proteases that TFPI may encounter in vivo, thrombin, plasmin, neutrophil elastase such as and some matrix metalloproteinases.46-49



Figure 1.4 Schematic representation of full-length TFPI.

The N-terminal region contains negatively charged residues (yellow), while the C-terminal region contains positively charged residues (red). N-linked glycosylations are shown as black squares, O-linked glycosylations as gray circles. P1 residues in the respective Kunitz domains are indicated by arrows (Figure from Crawley and Lane, 2008).<sup>26</sup> The usually reported sequence contains a Y in position 115, G in position 121 and L in position 197.

The majority of TFPI (~80%) is associated with the endothelial cell surface (Figure 1.5). It has been reported that the mechanism of this association is mainly

(~75%) through an indirect binding to GPI in the caveolae.<sup>42,50-53</sup> As TFPI does not contain a GPI anchor sequence it has been hypothesised that TFPI is associated to the cell surface indirectly through a GPI binding protein.<sup>54</sup> The remaining amount (~25%) of TFPI associated to the endothelial cell surface does so through a direct interaction with cell surface glycosaminoglycans (GAG) mediated by its C-terminal tail.<sup>55</sup> This pool of TFPI is released following heparin injection augmenting 2-4-fold the amount of circulating full-length TFPI.<sup>40,56-59</sup> The remaining TFPI (~20%) is either located inside platelets or circulates at a plasma concentration of ~2.5 nM (Figure 1.5).55 The small amount (~5%) of plasma TFPI localised inside platelets is in storage granules that are distinct from  $\alpha$ -granules.<sup>60</sup> This TFPI can be released following dual stimulation of platelets with collagen and thrombin which leads to platelet activation.<sup>61-63</sup> Of plasma TFPI, only ~10% is in its full-length 43 kDa form and represents the fraction of TFPI immediately available to counterbalance the procoagulant response. A small amount (~5%) of TFPI circulates as free truncated TFPI. The remaining plasma TFPI (~80%) is bound to lipoproteins.<sup>64,65</sup>





### 1.2.3 TFPI function

TFPI binds to and directly inhibits FXa and, in a FXa-dependent way, inhibits the TF/FVIIa complex by forming a quaternary, inactive FXa/TFPI/TF/FVIIa complex.<sup>66-68</sup> By inhibiting TF/FVIIa, TFPI inhibits the initiation phase of coagulation. The need for TFPI to bind and to inhibit FXa to effectively inhibit TF/FVIIa ensures that the TFPI pathway only regulates coagulation once it has started.<sup>69</sup>

## 1.2.3.1 TFPI mediated inhibition of FXa

FX is composed of a Gla domain responsible for its binding to phospholipids, two epidermal growth factor (EGF) domains required for protein-protein interaction and a serine protease domain responsible for the enzymatic activity of FXa. FX circulates as an inactive zymogen at a plasma concentration of ~136 nM and is activated following proteolysis by TF/FVIIa or by FVIIIa/FIXa.<sup>70</sup>

TFPI is a slow, tight-binding competitive inhibitor. TFPI binds FXa in a 1:1 stoichiometry, occupying its active site through its P1 residue, Arg107, in the K2 domain.<sup>66</sup> TFPI reversibly inhibits FXa by binding FXa in its active site. TFPI acts as a competitive inhibitor preventing FXa from binding to and activating prothrombin. It has been shown that FXa is able to hydrolyse TFPI after its P1 residue in the K3 domain, however, the physiological significance of this still needs to be evaluated.<sup>71</sup>

The kinetic mechanism of TFPI mediated inhibition of FXa is described as a two step reaction (Equation (Eq) 1).<sup>72</sup>



The first step describes the rapid formation of a loose initial FXa/TFPI complex. This is followed by a slower step resulting in formation of a final tight-binding of the enzyme to the inhibitor (FXa/TFPI\* complex). Tight-binding is characterised by inhibition occuring at concentrations of inhibitor close to that of the enzyme. The first step is described by the inhibition constant  $K_i$  ( $K_i$ =[FXa]\*[TFPI]/[FXa/TFPI]) and the overall reaction by the inhibition constant  $K_i^*$  ( $K_i^*$ =[FXa]\*[TFPI]/([FXa/TFPI]+[FXa/TFPI\*])).

It has been shown that both phospholipids and FVa can enhance TFPI mediated inhibition of FXa.<sup>72,73</sup> Phospholipids are likely to co-localise FXa and TFPI on its surface, as FXa binds phospholipids through its Gla domain and it has been suggested that the positively charged C-terminal tail of TFPI can interact with negatively charged phospholipid surfaces and cell surfaces.<sup>74-76</sup> TFPI inhibition of FXa in the prothrombinase complex is, however, inhibited by the presence of sub-physiological concentrations of prothrombin.<sup>73,77</sup> It is likely that this is a result of prothrombin competing for the active site of FXa.

It has been shown that TFPI truncated, either after its K3 domain (lacking the K3 domain and the C-terminal tail) or lacking the C-terminal tail, is not as potent inhibitor of FXa as full-length TFPI.<sup>78-80</sup> It has been suggested that the K3 domain and C-terminal tail of TFPI might interact with phospholipids and cell surfaces,<sup>74-76</sup> and that the C-terminal tail of TFPI might interact with the Gla domain of FXa<sup>79,80</sup> possibly partially explaining the reduced activity of truncated TFPI. More recently, it was shown that protein S acts as a cofactor towards full-length TFPI mediated inhibition of FXa but not towards TFPI lacking its K3 and C-terminal domain.<sup>81</sup>

## 1.2.3.2 TFPI mediated inhibition of TF/FVIIa

TFPI inhibits TF/FVIIa in a FXa dependent way.<sup>66-68</sup> As described in section 1.2.3.1 TFPI is a slow, tight-binding competitive inhibitor. It reversibly inhibits TF/FVIIa by binding the active site of FVIIa through its P1 residue (Lys36) in the K1 domain.<sup>66</sup> It has been shown that the intact GIa domain of FXa is required for TFPI mediated inhibition of TF/FVIIa.<sup>82,83</sup> One study used a plate binding assay and showed that FXa enhanced the binding of TFPI to phospholipids.<sup>76</sup> It has been suggested that the GIa domain and EGF1 domain of FXa are important for interactions with the TF/FVIIa complex, potentially explaining its importance in TFPI mediated inhibition of TF/FVIIa.<sup>84,85</sup>

Inhibition of TF/FVIIa is usually described as a two-step process, in which TFPI first binds to FXa (this being the rate-limiting step) and then inhibits TF/FVIIa by binding to FVIIa (Figure 1.6 left). Kinetic studies, however, favour the binding of

TFPI to FXa that is already bound to TF/FVIIa (Figure 1.6 right).<sup>68</sup> The resulting inactive quaternary TFPI/FXa/TF/FVIIa complex leads to inhibition of the initial procoagulant stimuli. Although the K3 domain and C-terminal tail of TFPI are not absolute requirements for inhibition of FXa/TF/FVIIa, their presence appreciably enhances the TFPI anticoagulant activity.<sup>78,79</sup>



#### Figure 1.6 TFPI binds to and inhibits FXa and TF/FVIIa.

Inhibition of TF/FVIIa is frequently described as a two-stage process (left) in which TFPI first binds FXa and then the TFPI/FXa complex binds and inhibits TF/FVIIa. However, kinetic studies favour a model whereby TFPI binds to newly activated FX prior its release from the TF/FVIIa/FXa complex (right). In both cases the resulting inactive quaternary complex is TFPI/FXa/TF/FVIIa.

## 1.3 Protein C/APC

### 1.3.1 PROC

Protein C is a plasma protein that is synthesised and secreted by hepatocytes. *PROC*, the gene that encodes protein C, is localised on chromosome 2 and contains 9 exons.<sup>86</sup> Translation of the messanger ribonucleic acid (mRNA) results in an immature protein of 461 amino acids.<sup>87</sup> Post-transcriptional modifications include  $\gamma$ -carboxylation,  $\beta$ -hydroxylation, glycosylation and removal of the Lys156-Arg157 dipeptide by proteolysis which results in conversion of protein C from a single chain to a two-chain protein. Following removal of the pro- and pre-peptide protein C is secreted as a mature zymogen of 419 amino acids with a MW of 62 kDa.

#### **1.3.2 Protein C structure**

Protein C circulates as a zymogen at a plasma concentration of ~65 nM. It is composed of a Gla domain, two EGF domains, an activation peptide (AP) and a serine protease domain (Figure 1.7).<sup>88</sup> The Gla domain (amino acids 1-45) contains 9 y-carboxylated Glu (Gla) residues (Gla6, 7, 14, 16, 19, 20, 25, 26, and 29).<sup>87,89</sup> These are essential for coordination of Ca<sup>2+</sup> ions and binding to negatively charged phospholipids surfaces.<sup>19</sup> It has also been shown that Leu8, which is part of the hydrophobic  $\omega$ -loop, is essential for binding to the endothelial protein C receptor (EPCR).<sup>90</sup> In addition, substitution of Asp36/Leu38/Ala39 results in a APC variant that can not be enhanced by protein S. The two EGF domains are predicted to be important for protein-protein interactions but no definitive role has so far been reported in the literature. EGF1 (amino acids 46-92) is β-hydroxylated at Asp71 and this residue in conjunction with Asp46 and Asp48 coordinates a Ca<sup>2+</sup> ion.<sup>91</sup> A study using a Fab fragment against this region suggested that the EGF1 of protein C might be important for interactions with protein S.<sup>91</sup> EGF2 (amino acids 93-137) is glycosylated at Asn97 and this has been reported to be important for effective secretion of protein C.<sup>92</sup> The serine protease domain (amino acids 138-419) is composed by a catalytic triad (His211, Asp257 and Ser360), and contains loop 225 and loop 70 involved in Na<sup>+</sup> and Ca<sup>2+</sup> binding, respectively.<sup>93</sup> It has been shown that binding of one ion to its loop increases the affinity of the other ion.93 The serine protease domain also contains loop 37, loop 60 and loop 148. Basic residues in loop 37, loop 70 and loop 148 are believed to form an exosite for binding to FVa and FVIIIa.<sup>94-98</sup>



#### Figure 1.7 Schematic structure of protein C.

The complete model of zymogen protein C was made by combining the crystal structure of the Gla-domainless APC with the Gla domain modelled from the crystal structure of the TF/FVIIa complex.<sup>88</sup> The AP was introduced by homology modelling. The serine protease domain is coloured in red, with the catalytic triad showed by the darker red spheres. The EGF2 is colured in dark blue, the EGF1 in cyan and the Gla domain in orange. The 9 Gla residues in the Gla domain are shown as sticks and coordinate calcium ions (yellow spheres). The Leu8 residue is shown in olive green and the Asp36/Leu38/Ala39 residues in the bright orange spheres.

### 1.3.3 Protein C activation

Cleavage of the negatively charged AP (Asp158-Arg169) of protein C by thrombin results in activation of protein C. Activated protein C is a two chain protein composed of a light (Gla-EGF1-EGF2) and heavy (AP and serine protease domain) chain connected through a disulphide bond (Cys141 and Cys277). Protein C is activated only once a sufficient amount of thrombin is generated. Thrombin that escapes the site of injury can bind to TM on intact endothelial cells adjacent to the site of vascular injury. TM is composed of a C-terminal cytoplasmic tail, a transmembrane region, a serine-threonine rich domain, 6 EGF domains and an N-terminal type C lectin-like domain.<sup>99</sup> Thrombin binds TM in the EGF5 and EGF6 domains through its exosite I.<sup>100,101</sup> It has also been shown that a chondroitin sulfate moiety of TM can interact with the exosite II of thrombin.<sup>102</sup>

As exosites I and II of thrombin are required for the activation of procoagulant factors FVa and FVIIIa,<sup>12,103-106</sup> binding of thrombin to TM results in a loss of its procoagulant functions. Instead, the specificity of thrombin is altered and the anticoagulant properties of thrombin are enhanced. The ability of thrombin to activate protein C is enhanced by 2000-fold following binding to TM.<sup>107</sup> Thrombin activates protein C by cleaving after Arg169 removing its AP. TM also binds the serine protease domain in the loops 37, 60, 70 and 148 of protein C through its EGF4 domain (Figure 1.8).<sup>107-109</sup> The activation of protein C is appreciably enhanced (more than 5-fold) by the presence of EPCR which binds the Leu8 in the Gla domain of protein C localising protein C to the endothelial surface close to the thrombin/TM complex.<sup>90,110-113</sup> This increases the local concentration of protein C available for activation.



#### Figure 1.8 Activation of protein C.

Thrombin (light purple) binds to EGF5 and EGF6 of TM through its excosite I domain. EPCR (dark purple) binds the Gla domain of protein C resulting in the alignment of protein C with the thrombin/TM complex. EGF4 of TM is able to interact with loops 37, 60, 70 and 148 (not shown) of protein C bringing the AP of protein C in proximity of the active site of thrombin. Cleavage after Arg169 of protein C by thrombin activates protein C. Figure modified from Dahlbäck and Villoutreix.<sup>114</sup>

### 1.3.4 APC function

APC downregulates coagulation by proteolytical inactivation of FVa and FVIIIa, the non-enzymatic cofactors of FXa and FIXa, respectively.<sup>27</sup> FV and FVIII have a highly homologous domain structure containing an N-terminal A1 and A2 domain, a B domain, and a C-terminal A3 domain followed by a C1 and C2 domain. The C2 domains possibly also the C1 domains are responsible for FV and FVIII binding to phospholipids.<sup>20-22</sup> The A2 and A3 domains of FVa and FVIIIa are important for the interaction with FXa and FIXa respectively.<sup>115-118</sup>

### 1.3.4.1 APC mediated inactivation of FVa

Activation of FV occurs by the cleavage of peptide bonds in the B domain by either thrombin or FXa (Figure 1.9). This leads to dissociation of the B domain resulting in a two-chain molecule. The heavy chain (A1-A2 amino acids 1-709) and the light chain (A3-C1-C2 amino acids 1546-2196) remain associated by a non covalent complex between the A1 and A3 domains. While the association between the domains was believed to be through a high affinity Ca<sup>2+</sup> interaction this has recently been questioned.<sup>20</sup> FVa can be inactivated by APC by proteloytic cleavage. APC can cleave FVa after Arg306, Arg506 and Arg679 (Figure 1.9).<sup>119</sup>





Thrombin activates FV by proteolytic cleavage in its B domain. The heavy and light chains of FVa are associated through a high affinity binding to a calcium ion. APC inactivates FVa by proteolytic cleavage after Arg306, Arg506 and Arg679. Structure/figure of FV based on Kane *et al.* 1986, Kane *et al.* 1987, Jenny *et al.* and Laffan.<sup>120-123</sup>

While the physiological relevance of Arg679 cleavage is still being debated it is widely accepted that APC inactivates FVa in plasma by cleavage at Arg506 and Arg306. Cleavage at Arg506 is enhanced and cleavage at Arg306 dependent of phospholipids, with phosphatidylserine upon the presence and phosphatidylethanolamine both being important for the stimulation of APC mediated inactivation of FVa.<sup>124-128</sup> In FVa variants created for *in vitro* studies additional cleavage sites (Lys309, Arg313, Arg316, Arg317 and Arg505) have been identified.<sup>129,130</sup> However, the physiological role of these cleavage sites is questionable. Following cleavage after Arg506 and Arg306 the A2 domain of FVa dissociates resulting in complete inactivation of FVa. Inactivation of FVa is biphasic.<sup>128,131,132</sup> This is a consequence of the different rates of APC cleavage after Arg506 and Arg306. Arg506 is the first and kinetically most favoured cleavage site of APC and results in an intermediate FVa form which have been reported to have approximately 40% of intact FVa cofactor activity.<sup>128</sup> The second cleavage of FVa by APC occurs after Arg306 but is ~20-fold slower than the first cleavage at Arg506.<sup>128</sup> It has been shown that cleavage after Arg506, but to a minor extent also cleavage after Arg306, results in a significant decrease of FVa affinity for FXa and prothrombin.<sup>128,133,134</sup> It is unresolved whether complete loss of FVa activity occurs following Arg506 and Arg306 cleavage (before dissociation of the A2 domain) or only after dissociation of the A2 domain.<sup>133,135</sup>

It is known that FVa is protected from APC mediated proteolytic inactivation following binding to FXa<sup>136-139</sup> and prothrombin.<sup>140-142</sup> It has been shown that binding of FXa to FVa specifically inhibits APC cleavage after Arg506 by approximately 20-fold<sup>137,138</sup> and it has been recently suggested that FXa can stimulate APC cleavage at Arg306.<sup>137</sup> Binding of prothrombin to FVa has been shown to inhibit both APC mediated cleavage at Arg306 and Arg506 of FVa, however prothrombin did not inhibit APC mediated inactivation of FVIIIa.<sup>141,142</sup>

### 1.3.4.2 APC mediated inactivation of FVIIIa

FVIII circulates in plasma bound to VWF which act as a carrier protein increasing the half life of FVIII.<sup>143-145</sup> FVIII is activated following proteolytical cleavage by thrombin or FXa in the B domain. Only after activation FVIIIa dissociates from VWF.<sup>146,147</sup> Similarly to FV activation, activation of FVIII results in a two chain

molecule with a light and heavy chain. However, an additional cleavage between the A1 and A2 domains by thrombin leads to a weak association between the A1 and A3 domain which are linked by a Cu<sup>2+</sup> ion.<sup>148-150</sup> Binding to Ca<sup>2+</sup> or Mn<sup>2+</sup> is also required to yield active FVIIIa.<sup>150-152</sup> FVIIIa is unstable and spontaneous dissociation of the A2 domain results in loss of activity.<sup>153,154</sup> Incorporation of FVIIIa into the tenase complex stabilises FVIIIa and also protects it from APC mediated inactivation.<sup>155,156</sup> APC inactivates FVIIIa by proteolytical cleavage after Arg336 and Arg562. Cleavage occurs preferentially at Arg336 and thereafter at Arg562, resulting in dissociation of the A2 domain and complete loss of function of FVIIIa.<sup>157,158</sup>

## 1.4 Protein S

### 1.4.1 PROS1

Protein S is encoded by the *PROS1* gene on chromosome 3 (p11.1), which spans more than 100 kb and gives rise to an mRNA of 3292 bases.<sup>159,160</sup> Following transcription the 676 amino acid long protein is directed to the endoplasmic reticulum (ER) by its hydrophobic signal peptide. Here, protein S goes through several post-transcriptional modifications, including  $\gamma$ -carboxylation, glycosylation and  $\beta$ -hydroxylation.  $\gamma$ -carboxylation is carried out by the vitamin K-dependent  $\gamma$ carboxylase which recognises and binds to the propeptide of protein S. Prior to secretion both the signal peptide and propeptide are proteolytically removed generating a mature protein of 635 amino acids with a reported molecular weight of 69-77 kDa.<sup>161,162</sup>

Protein S is synthesised and secreted into the blood stream mainly by hepatocytes but also by endothelial cells, megakaryocytes and vascular smooth muscle cells.<sup>163-166</sup> The resulting concentration of protein S in plasma is ~0.35  $\mu$ M.<sup>167</sup> Of plasma protein S, 60% circulates as a non-covalent, high-affinity complex with C4b-binding protein (C4BP) while the remaining 40% is free.<sup>168</sup>

#### 1.4.2 Protein S structure

The mature protein S is composed of an N-terminal Gla domain (amino acids 1-46), a thrombin sensitive region (TSR, amino acids 47-75), four EGF domains (EGF1, amino acids 76-116; EGF2 amino acids 117-160; EGF3, amino acids 161-202; EGF4, amino acids 203-242) and a C-terminal Sex Hormone Binding Globulin-like domain (SHBG) (amino acids 243-635) containing two laminin G (LG)-type domains (Figure 1.10).<sup>169,170</sup>

The N-terminal GIa domain of protein S contains one disulphide bond (Cys17-Cys22) and 11 glutamic acid residues (Glu 6, 7, 14, 16, 19, 20, 25, 26, 29, 32 and 36) that are  $\gamma$ -carboxylated by the vitamin K-dependent  $\gamma$ -carboxylase in the ER.<sup>171</sup> The resulting N-terminal GIa residues coordinate 8 calcium ions which induce an appropriate conformational change in protein S for activity.<sup>19,171,172</sup> Folding of the GIa domain enables protein S to bind to phospholipids through the insertion of its hydrophobic  $\omega$ -loop into the membrane and ionic interactions between positively charged calcium ions and the negatively charged phospholipid surface.<sup>19,171,172</sup> Binding of protein S to phospholipid surfaces localises protein S to activated endothelial cells and activated platelet surfaces and is required for both its anticoagulant and non-anticoagulant activity (see section 1.4.3).

The TSR is formed by two α-helixes brought in close contact through a disulphide bond between Cys47 and Cys72 and is a domain unique to protein S. Its name derives from the observed cleavage of this region by thrombin. While cleavage at Arg49 and Arg70 by thrombin has been observed *in vitro*, this only occurs in the absence of Ca<sup>2+</sup> and phospholipids, suggesting that this is not of physiological importance.<sup>173</sup> FXa has also been shown to be able to cleave the TSR at Arg60. Recently, it was shown that the TSR domain of protein S could bind to FXa in the presence of phospholipids.<sup>174</sup> Although FXa cleavage is both Ca<sup>2+</sup> and phospholipid dependent, it is unlikely to occur *in vivo* due to its slow kinetics and the inability of FXa to cleave protein S when bound to its cofactor FVa or when in presence of prothrombin.<sup>173</sup> It is, however, interesting to notice that cleavage of the TSR has been observed to inhibit the APC-dependent cofactor activity of protein S.<sup>175-177</sup> The effect of cleavage upon the protein S APC-independent activity is still not clear, but it has been suggested that it is abolished only

following cleavage after both Arg49 and Arg70.<sup>176</sup> It has been reported that approximately 10% of plasma protein S circulate in its cleaved form.<sup>178</sup>

The EGF1 domain of protein S contains 3 disulphide bonds and is  $\beta$ -hydroxylated at Asp95.<sup>171</sup> This domain has to date not been found to bind Ca<sup>2+</sup> and  $\beta$ -hydroxylation is thought not to be a requisite for protein S function.<sup>179</sup>

The EGF2-4 domains all contain 3 disulphide bonds each and a  $\beta$ -hydroxyasparagine residue (Hyn 136, Hyn 178 and Hyn 217). Each domain also contains a high affinity Ca<sup>2+</sup> binding site<sup>180</sup> (dissociation constant (K<sub>d</sub>) nM range), which is thought to include the modified Asn residue. Binding of Ca<sup>2+</sup> to these domains have been shown to be important for the function of protein S. In 2006 Rezende and colleagues demonstrated that the abolition of binding of Ca<sup>2+</sup> to EGF4 impaired both the structure and function of EGF1.<sup>181</sup> It has also been suggested that a naturally occurring deletion mutation in EGF4 (deletion I203D204) alters the conformation and function of the Gla domain,<sup>182</sup> while deletion or replacement of the entire EGF2 domain results in a significant loss of activity.<sup>183</sup>

The C-terminal SHBG domain of protein S represents more than half of its amino acids and is kept in its globular state through 3 disulphide bonds. It contains 3 potential N-linked glycans at Asn458, Asn468 and Asn489 and is also predicted to contain Ca<sup>2+</sup> binding sites.<sup>184,185</sup> As anticipated, 60% of protein S circulates as a non-covalent, high-affinity complex with C4BP. This interaction is mediated through the SHBG domain of protein S.<sup>184,186</sup> C4BP is a 570 kDa multimeric protein consisting of 6 or 7 identical  $\alpha$ -chains with one or no  $\beta$ -chain held together by disulphide bonds. It is this  $\beta$ -chain which contains the binding site for protein S.<sup>187</sup> The net plasma concentration of C4BP is ~260 nM. Of this only ~80% (~208 nM) has the  $\beta$ -chain and is able to bind to protein S with a 1:1 stoichiometry.<sup>187</sup> The amount of unbound or free circulating protein S is equal to the molar excess of protein S over  $\beta$ -chain containing C4BP (~145 nM) and represents the pool of protein S with major anticoagulant function.<sup>188</sup>

Multimerisation of protein S has been documented in the literature. Protein S is generally thought to exist in plasma in its monomeric form (although some have reported multimeric forms of protein S in plasma) and multimerisation has been attributed at least in part to *in vitro* manipulation.<sup>189-192</sup>

36




Protein S contains a N-terminal Gla domain that is responsible for binding to phospholipid surfaces. This is followed by a TSR domain, 4 EGF domains and a SHBG domain. The figure in the box has been created from the Gla-TSR-EGF1 model of protein S.<sup>193</sup> The Gla domain is coloured in yellow, with the 11 Gla residues shown as sticks and the 8 calcium ions shown as orange spheres in the box at the right. The  $\omega$ -loop of protein S can be seen underneath the calcium ions. The TSR is shown in green and the EGF1 domain in cyan.

#### 1.4.3 Protein S functions

Protein S is a multifunctional protein involved in coagulation, the classical complement pathway and apoptotic clearance.<sup>194-196</sup> It is also thought to stimulate vascular injury repair by acting as a mitogen towards smooth muscle cells and has recently been found to have a neuroprotective effect during ischemic brain injury in mice.<sup>197-200</sup>

The most studied protein S function is its anticoagulant activity, which is also the focus of my thesis. Protein S has an important role *in vivo*, as is shown clinically by infants with complete deficiency, who suffer purpura fulminans despite treatment of anticoagulants, and by heterozygous carriers of *PROS1* gene deletions and point mutations, who are at enhanced risk of venous thromboembolism.<sup>201-203</sup> Hereditary protein S deficiency is an uncommon autosomal dominant disorder. It is classified into type I, type II and type III

deficiencies. Type I and type III deficiencies are defined as quantitative deficiencies of total and free protein S, respectively, and correspond to ~96% of all protein S deficiencies.<sup>202</sup> The remaining ~4% are represented by type II deficiencies. This subtype is classified as a qualitative deficiency.<sup>202</sup> Acquired protein S deficiencies are more common and occur as a consequence of oral anticoagulant therapy, oral contraception, liver disease, nephritic syndrome and pregnancy.<sup>202,204</sup> Protein S deficiencies, inherited or acquired, both lead to an increased risk of thrombotic events.<sup>205</sup> The importance of protein S has also been demonstrated in murine knockouts, which fail to survive development.<sup>206,207</sup>

Protein S is best known for its anticoagulant activity towards APC. Protein S was discovered in 1977 by Richard DiScipio and was 3 years later shown to be the cofactor of APC.<sup>208</sup> Protein S was, however, also suggested to have an APC-independent activity in 1988.<sup>209</sup> In the early 1990's it was demonstrated that protein S could bind to FVa,<sup>116,210-212</sup> FXa<sup>174,213,214</sup> and FVIIIa<sup>215</sup> inhibiting both the tenase and prothrombinase complex in an APC-independent manner. It was shown that the observed inhibition of the tenase and prothrombinase complex was due to competition of protein S multimers for binding to limiting phospholipid surfaces.<sup>191,192,216,217</sup> However, it was later shown that protein S had an APC-independent activity in plasma also when saturating amounts of phospholipids were present, but which could be demonstrated only at low concentrations of TF.<sup>218</sup>

#### 1.4.3.1 Protein S enhances the anticoagulant activity of APC

#### 1.4.3.1.1 Protein S enhancement of APC mediated inactivation of FVa

Protein S enhances the APC mediated inactivation of FVa in a phospholipid dependent manner.<sup>124,208,219</sup> It does so by selectively enhancing by approximately 20-fold the slow cleavage at Arg306 of FVa.<sup>125,138</sup> The use of FV variants FV R306Q/R679Q and FV R506Q/R679Q, that can be cleaved only in Arg506 and Arg306, respectively, has been useful to characterise the effect of protein S and other components on the individual cleavage sites.<sup>125,137,220</sup> Through the use of such variants it has recently been suggested that protein S can also enhance ~5-fold APC mediated cleavage at Arg506.<sup>125</sup>

Protein S has been proposed to enhance APC mediated inactivation of FVa in 3 distinct ways. It has been shown to enhance APC binding to phospholipid surfaces, platelets and endothelial cells,<sup>127,221,222</sup> consequently localising APC in close proximity to the membrane bound FVa and FVIIIa. This proposal is supported by the low affinity of APC for phospholipids. While the apparent dissociation constant (K<sub>d(app)</sub>) of protein S for phospholipids is between 4-15  $nM^{182,223-225}$  the K<sub>d(app)</sub> of APC for phospholipids is around 2-7  $\mu$ M,<sup>226,227</sup> which is well outside the range of the physiological plasma concentration of APC (~65 nM). The second mechanism by which protein S is thought to enhance APC anticoagulant activity is through inducing an allosteric conformational change in APC.<sup>228,229</sup> Yegneswaran et al. used fluorescence resonance energy transfer (FRET) and showed that the active site of APC was localised 94 Ångström (Å) above the membrane surface. However, in the presence of protein S the active site of APC was relocated to 84 Å from the membrane surface.<sup>229</sup> It has been suggested that this shift in the active site of APC in the presence of protein S facilitates the preference of APC cleavage from Arg506 to Arg306. As described in section 1.3.4.1, FXa and prothrombin are able to protect FVa from APC mediated cleavage. The third mechanism by which protein S is thought to enhance APC mediated inactivation of FVa is through overcoming the protective effect of FVa.<sup>137,219</sup> It is still being debated whether protein S competes with the binding of FXa to FVa or not.<sup>116,137,138</sup> Depending on this, protein S could either act by competing with FXa binding<sup>116</sup> to FVa. Alternatively, protein S could enhance cleavage of free FVa resulting in depletion of free FVa and dissociation of FXa/FVa complexes.<sup>137</sup> An other study showed that the binding sites of protein S and FXa on FVa are distinct and that protein S is likely to enhance APC mediated cleavage at Arg306 of FVa even when the latter is bound to FXa.<sup>138</sup> A recent paper suggests that not only protein S but also FV can act as a cofactor for APC mediated inactivation of FVa.<sup>230</sup>

While it was previously thought that only free protein S could act as a cofactor for APC, recent work shows that protein S bound to C4BP is able to enhance cleavage of FVa.<sup>231</sup> However, while cleavage after Arg306 by this complex is enhanced 10-fold (compared to ~20-fold by free protein S), cleavage after Arg506 is inhibited 3-4-fold (compared to no or little influence of free protein S).<sup>231</sup> Consequently, bound protein S enhances APC cleavage of FVa, but is 6-8-fold

less effective that free protein S, potentially explaining why in prior experiments the addition of C4BP resulted in a decreased APC activity.

#### 1.4.3.1.2 Protein S enhancement of APC mediated inactivation of FVIIIa

It has been shown that protein S can enhance both the slow cleavage at Arg562 by ~3-fold and cleavage at Arg336 by ~2-fold. <sup>232,233</sup> However, FV cleaved at Arg506 also acts in synergy with protein S as a cofactor enhancing ~10-fold APC mediated inactivation of FVIIIa.<sup>157,233,234</sup> Both free protein S and protein S bound to C4BP act as cofactors for APC mediated inactivation of FVIIIa.<sup>235</sup> However, it has been reported that while free protein S acts in synergy with FV in the APC mediated inactivation of FVIIIa, bound protein S does not.<sup>235</sup>

It is believed that protein S exerts its cofactor activity through increasing the binding of APC to phospholipid surfaces, platelets and endothelial cells, as described in section 1.4.3.1.1. <sup>127,221,222</sup>

Similar to FVa in the prothrombinase complex, when FVIIIa is assembled in the tenase complex cleavage is protected by FIXa. Protein S can overcome the protective action of FIXa by competing with its binding to FVIIIa.<sup>236</sup>

#### 1.4.3.2 Protein S enhances the anticoagulant activity of TFPI

#### 1.4.3.2.1 Protein S enhancement of TFPI mediated inactivation of FXa

In 2006 Hackeng *et al.* showed that protein S enhances the anticoagulant activity of TFPI, potentially providing a molecular explanation behind the APC-independent activity observed in plasma.<sup>81</sup> Since it had been shown that the APC-independent activity of protein S was visualised only at low concentrations of TF,<sup>218</sup> their investigation hypothesised that protein S had an effect on TF mediated activation of FX. Hackeng *et al.* analysed thrombin generation by calibrated automated thrombography (CAT). They showed that in normal plasma, in the presence of polyclonal antibodies against APC, antibodies against protein S increased thrombin generation. The same effect was, however, not observed in TFPI depleted plasma, suggesting the anticoagulant activity of protein S was dependent upon the presence of TFPI. This was confirmed by supplementing the plasma with TFPI. They also performed a FXa inhibition assay. This was a

purified assay in which FXa hydrolysis of a chromogenic substrate was measured over time. It was observed that protein S could potentiate the TFPI mediated inhibition of FXa by enhancing the formation of the initial FXa/TFPI complex (see section 1.2.3.1). Little effect was seen on the transition from the loose FXa/TFPI complex to the tightly bound FXa/TFPI\* complex. The effect of protein S was phospholipid dependent and resulted in a 10-fold reduction of the K<sub>i</sub> (see section 1.2.3.1), from 4.4 nM to 0.5 nM, with little effect on the K<sub>i</sub>\*.<sup>81</sup> Ndonwi and Broze confirmed that protein S enhances TFPI inhibition of FXa.<sup>237</sup> Hackeng *et al.* also showed that protein S was not able to enhance the activity of TFPI lacking its K3 domain and C-terminal tail, and that protein S bound to C4BP only exerted 60% of the activity of free protein S.<sup>81</sup> Ndonwi *et al.* recently showed that the K3 domain of TFPI and in particular its P1 residue were necessary for protein S binding and cofactor activity towards TFPI.<sup>238</sup>

Rosendaal and colleagues observed that higher plasma concentrations of protein S resulted in a higher anticoagulant activity of TFPI.<sup>239</sup> Recently, it has been shown that the impaired protein S enhancement of TFPI is potentially an important component in patients with protein S deficiency.<sup>240</sup> Interestingly, it has been suggested that protein S binds to TFPI in plasma and that this might account for the observed decrease in TFPI antigen levels in protein S deficient patients.<sup>241</sup> The detailed molecular mechanisms of the TFPI-dependent protein S anticoagulant activity were unknown and form an important part of the investigation carried out in my thesis.

Even though the APC-independent activity of protein S now has a plausible explanation (it acts as a cofactor of TFPI), Heeb and colleagues have reported that protein S also has a "direct" activity, independent of both APC and TFPI, exerted by direct binding to and inhibition of FXa and TF/FVIIa.<sup>189,190,242</sup>

#### 1.4.3.2.2 Protein S enhancement of TFPI mediated inactivation of TF/FVIIa

Studying the protein S cofactor activity on TFPI mediated inactivation of TF/FVIIa is complicated by the difficulty in distinguishing between enhanced TF/FVIIa inhibition and enhanced inhibition of FXa. While it was initially suggested that protein S could enhance TFPI inhibition of TF/FVIIa activation of FX,<sup>81</sup> Ndonwi and Broze suggested that this was not the case.<sup>237</sup> The latter support the

proposal that TFPI predominantly binds to FXa already bound to TF/FVIIa, rather than the previously suggested two-step mechanism by which TFPI first binds FXa and FXa/TFPI subsequently binds TF/FVIIa (see section 1.2.3.2). Consequently, Ndonwi and Broze suggested the major role of protein S enhancement of TFPI was to down-regulate any FXa that escapes TF/FVIIa/FXa inhibition by TFPI.

#### 1.5 Domains of protein S which mediate functional interactions

While the functions of the domains of protein S have been extensively studied, the APC cofactor function and the recently discovered TFPI cofactor activity of protein S remain poorly understood at the molecular level. A number of basic and clinical studies have confirmed that alteration of protein S can disrupt phospholipid binding and thereby impair function of protein S.<sup>182,223</sup> Consequently, variants with impaired cofactor function need to be evaluated for their ability to bind to phospholipids and those that clearly alter phospholipid binding can be discounted from further investigation. A search of the literature has suggested the domains of protein S that are most likely involved in APC and TFPI cofactor activity (see sections 1.5.1 and 1.5.2 below).

# 1.5.1 Domains of protein S reported to be important for APC cofactor activity

Saller *et al.* prepared protein S/prothrombin chimeras and on the basis of functional studies supported by molecular modelling, identified a cluster of protein S Gla domain residues (termed Face2) which they proposed would provide an APC interaction site.<sup>224</sup> Additional studies have focused on the possible role of TSR and EGF domains. Several studies have reported that cleavage in the TSR of protein S affects its APC-dependent and APC-independent cofactor activity.<sup>175-177</sup> Dahlbäck *et al.* used domain specific monoclonal antibodies directed against protein S and their Fab fragments and suggested that the TSR and EGF1 domain of protein S have also been reported to be responsible for APC species specificity.<sup>244</sup> By using EGF-like modules Stenberg *et al.* concluded the EGF1 module to be crucial for interaction with APC.<sup>245</sup> Hackeng *et al.* demonstrated that

the isolated EGF1 domain of protein S could inhibit full-length protein S anticoagulant enhancing function for APC: direct binding of the domain to APC was also demonstrated.<sup>246</sup> Mille-Baker *et al.* prepared EGF2 domain deletion and substitution variants of protein S and demonstrated reduced APC anticoagulant cofactor activity, in the presence of normal phospholipid binding.<sup>183</sup> Collectively, these investigators have suggested Gla, TSR, EGF1 and EGF2 domains may each contain recognition elements on protein S needed for APC cofactor function.

### 1.5.2 Domains of protein S that are potentially important for TFPI cofactor activity

Work presented in preliminary form at the ISTH congress in Geneva by Hackeng *et al.*, who had carryied out an *in silico* molecular docking experiments of protein S and TFPI, suggested that the TSR of protein S might be involved in the interaction with TFPI. Recent reports suggest that free protein S, but not protein S bound to C4BP, form a reversible interaction with free full-length TFPI in plasma.<sup>241</sup> In addition, free protein S seems to act more efficiently as a cofactor for TFPI than protein S in complex with C4BP.<sup>81</sup> Collectively, these results would suggest that the N-terminal domains of protein S are more likely to be involved in an interaction with TFPI than the C-terminal domains, which can be bound to C4BP.

#### 1.6 Aims of my thesis

The aims of my project were to determine the domains and residues of protein S involved in the interaction with both APC and TFPI. Following selection of the domains, over 30 point or composite variants in these domains were made. Wild type (WT) protein S and protein S variants were expressed in human embryonic kidney fibroblast (HEK) cells and purified as required. The functional cofactor activity of protein S variants towards APC and TFPI anticoagulant activities was evaluated and compared to that of WT protein S in various functional assays including calibrated automated thrombography (CAT), FVa inactivation assay and FXa inhibition assay.

### 2 METHODOLOGY

A schematic overview of the sequence of methods used during this project and the sections in which they are described is shown in Figure 2.1.



Figure 2.1 Overview of the Methodology of this thesis.

#### 2.1 Generation of protein S mutant expression vectors

# 2.1.1 Mammalian protein S expression vector and composite mutants generated

A pcDNA6/protein S vector (Figure 2.2) containing the complimentary deoxyribonucleic acid (cDNA) for the WT protein S sequence had previously been generated in our lab by Dr S.M. Rezende (Universidade Federal de Minas Gerais, Brazil).<sup>247</sup> The V5 epitope and the polyhistidine tag were excised from the pcDNA6/V5-HisB vector (Invitrogen) by digestion with the Nhel restriction enzyme (New England Biolabs). The cDNA of protein S, amplified by polymerase chain reaction (PCR) with primers containing the restriction site for Nhel, was digested with the restriction enzyme and ligated into the vector. The resulting pcDNA6/protein S vector contains the SV40 origin of replication, cytomegalovirus (CMV) and T7 promoters, and resistance to ampicillin and blasticidine. The plasmid was used as a template to produce single point mutations by PCR mediated site-directed mutagenesis. Newly produced cDNA with point mutations were subsequently used as a template to introduce additional mutations resulting in single and composite mutants.



#### Figure 2.2 Protein S expression vector.

The cDNA for protein S has been inserted into the vector through ligation of the Nhel digested ends of the insert to the vector. The vector contains resistance to ampicillin for selection of transformed bacteria and the CMV promoter for expression in mammalian cells. The restriction sites BsmBI and BspEI are used for subcloning of the mutants generated (see section 2.1.6). bp; base pair.

#### 2.1.2 Generation of protein S mutants by site-directed mutagenesis

Complementary primer pairs with the desired base pair substitutions (for amino acid residues mutated see Table 3.1, for primer sequences see Appendix) were manufactured by Thermo Scientific. PCR reactions were performed using Stratagene's QuickChange XL Site-Directed Mutagenesis Kit. In detail, final concentrations of 100 ng of protein S vector, 65 ng of forward primer, 65 ng of reverse primer, 0.5  $\mu$ l deoxynucleoside triphosphate (dNTP) mix, 1X reaction buffer, 1.5  $\mu$ l of QuikSolution and 1.25 U of high fidelity *Pfu ultra* deoxyribonucleic acid (DNA) polymerase were used in a final volume of 25  $\mu$ l.

Alternatively, mutagenesis was performed with the KOD hot start DNA polymerase (Novagen). In detail, final concentrations of 100 ng of dsDNA template, 0.3  $\mu$ M of forward primer, 0.3  $\mu$ M of reverse primer, 0.2 mM dNTPs, 0.5  $\mu$ I DMSO, 1.5 mM MgSO<sub>4</sub>, 1X reaction buffer and 0.5 U KOD hot start DNA polymerase were used in a final volume of 25  $\mu$ I.

Cycling parameters for *Pfu ultra* and KOD hot start DNA polymerase are outlined in Table 2.1 and Table 2.2 respectively.

After PCR, parental dsDNA template was digested with 5 U of DpnI at 37°C for 1 hour before transforming bacteria.

Stage	Phase	Cycles	Temperature (°C)	Time
Denaturation		1	95	2 minutes
	Denaturation		95	50 seconds
Amplification	Annealing	18	60	50 seconds
	Extension		68	1 min/kb plasmid
Final extension		1	68	7 minutes

 Table 2.1 PCR cycle parameters for *Pfu ultra* DNA polymerase.

PCR cycle parameters used to generate protein S mutants by site-directed mutagenesis. Min; minute.

Stage	Phase	Cycles	Temperature (°C)	Time	
Denaturation		1	95	5 minutes	
	Denaturation		94	15 seconds	
Amplification	Annealing	18	60	30 seconds	
	Extension		72	20 sec/kb plasmid	
Final extension		1	72	5 minutes	

#### Table 2.2 PCR cycle parameters for KOD hot start DNA polymerase.

PCR cycle parameters used to generate protein S mutants by site-directed mutagenesis. Sec; seconds.

#### 2.1.3 Transformation of competent cells

Competent cells (XL-10 gold ultracompetent cells from Stratagene, One shot TOP10 competent cells from Invitrogen or NEB Turbo competent cells from New England Biolabs) were transformed with the mutated protein S vector according to the manufacturer's instructions. In detail, 45  $\mu$ l of cells were thawed on ice. XL-10 gold ultracompetent cells were incubated with 2  $\mu$ l of  $\beta$ -mercaptoethanol for 10 minutes, gently swirling the tube every 2 minutes. Competent cells were incubated with 2  $\mu$ l of Dpnl treated DNA; the tube was swirled and incubated on ice for 30 minutes. Cells were heat-pulsed in a water-bath at 42°C for 30 seconds and placed on ice for an additional 2 minutes. Transformed bacteria were grown in 0.25-0.5 ml of preheated S.O.C. medium at 37°C for 1 hour with shaking. Bacteria were subsequently plated out on an LB-agar (Invitrogen) plate containing 100  $\mu$ g/ml ampicillin and incubated at 37°C over night. Ampicillin-resistant clones were picked and grown in 5 ml LB-broth (Invitrogen) with 100  $\mu$ g/ml ampicillin over night at 37°C with shaking.

#### 2.1.4 Generation of glycerol stocks

Of the 5 ml of bacteria in section 2.1.3, 700  $\mu$ l were mixed with 300  $\mu$ l of autoclaved 50% glycerol solution. The resulting mix was vortexed and stored at - 80°C. This could be partially thawed when additional transformed bacteria were

required. The remaining 4.3 ml of bacteria were used to extract plasmid DNA by miniprep.

#### 2.1.5 Plasmid miniprep and sequencing

Plasmid DNA was extracted with the Qiagen Miniprep kit according to instructions. Mutations were confirmed by sequencing the protein S vector with the T7 primer 5'-TAA TAC GAC TCA CTA TAG GG-3' (MRC Genomics Core Facility).

### 2.1.6 Subcloning of the protein S mutants into the pcDNA6/Protein S vector

Prior to mammalian expression, the mutated cDNA was sequenced with additional primers spanning the whole cDNA of protein S to make sure no unwanted mutations had been introduced (see Appendix for sequencing primers). Alternatively, the cDNA region containing the mutation was subcloned into the original WT vector. To perform such subcloning, the vector with the protein S mutation (40  $\mu$ l, ~16  $\mu$ g) and the WT vector (1  $\mu$ g) were digested with BsmBI (New England Biolabs) at 55°C for 2 hours and subsequently with BspEI (New England Biolabs) at 37°C for 2 hours 30 minutes in buffer 3 (New England Biolabs) in a total volume of respectively 50 and 25  $\mu$ l (Table 2.3).

	Digestion of:					
	Protein S mutant vector	WT vector				
DNA	40 µl (16 µg)	0.77 µl (1µg)				
BsmBI (10 U/µI)	1 µl	0.2 µl				
BspEl (10 U/µl)	1 µl	1 µl				
Buffer 3	5 µl	2.5 µl				
Water	3 μΙ	20.53 µl				
Total volume	50 µl	25 µl				

Table 2.3 Enzymatic digestion of WT and mutated pcDNA6/PS vector.

Digested DNA products were run on a 1% agarose gel containing SYBRSafe in tris borate ethylenediaminetetraacetic acid (TBE) buffer. Samples and marker (1 kb or 100 bp) were electrophoresed for 20-40 minutes at 80 V. The DNA bands of interest were excised and purified using the QIAquick Gel extraction kit (QIAGEN) according to the manufacturer's instructions.

The linearised pcDNA6/Protein S vector was dephosphorylated with 0.01 U of Calf Intestinal Alkaline Phosphatase (CIAP) as described in Table 2.4, according to the manufacturer's instructions (Invitrogen). The reaction was performed at 37°C for at least 1 hour and the dephosphorylated vector was subsequently purified using QIAquick PCR purification Kit protocol (QIAGEN).

Dephosphorylation							
DNA (~0.48 pmol)	25 µl						
10X dephosphorylation buffer	4 μΙ						
Water	10 µl						
CIAP (0.01 U/µI)	1 µl						
Total volume	40 µl						

Table 2.4 Dephosphorylation of the linearised WT vector.

#### 2.1.7 DNA ligation

Approximately 60 fmol of the DNA fragment containing the mutation were ligated into 20 fmol of the dephosphorylated vector (molar ratio insert:vector, 3:1) with 0.1 U of T4 DNA ligase in a total volume of 20  $\mu$ l, at 23-26°C for 1 hour as described in Table 2.5.

Ligation reaction					
Vector	3 µl (20 fmol)				
Insert	0.5 µl (60 fmol)				
5X ligase buffer	4 µl				
T4 ligase (0.1 U/µI)	1 µl				
Water	11.5 µl				
Total volume	20 µl				

Table 2.5 Ligation of the mutated protein S region into the WT protein S vector.

#### 2.1.8 Transformation of competent cells with subcloned mutants

The ligation product was used to transform One shot TOP10 (Invitrogen) or NEB turbo chemically competent cells (New England Biolabs) according to the manufacturer's instructions. Transformed bacteria were plated out on LB-agar/ampicillin plates over night at 37°C. Ampicillin resistant-colonies were picked and grown in 5 ml Luria-Bertani medium (LB) broth over night. The plasmid was extracted with the Qiagen Miniprep kit and sequenced (for primers see Appendix) as outlined previously (MRC Genomics Core Facility). Clones were sequenced across both restriction sites to confirm correct ligation.

#### 2.1.9 DNA maxiprep/megaprep and quantification

Bacteria containing the successfully mutated protein S vector were cultured on a large scale (200-500 ml) to obtain sufficient amounts of plasmid (extracted with a Qiagen Maxiprep or Megaprep kit) for transfection of mammalian cells.

The DNA was quantified with the Quant-IT<sup>™</sup> nucleic acid Assay Kit (Invitrogen) according to manufacturer's instructions or with a NanoDrop spectrophotometer by measuring the absorbance at 260 nm.

#### 2.2 Expression of protein S

#### 2.2.1 Mammalian cell culture

HEK293T and HEK293 cells (ATCC) were used respectively for transient and stable transfections. Cells were cultured at  $37^{\circ}$ C, 5% CO<sub>2</sub>, in complete media; minimum essential media (MEM, Invitrogen) supplemented with 10% foetal calf serum (FCS) (Biosera), 2 mM L-glutamine (Invitrogen), 50000 U penicillin/ 50000 µg streptomycin and 1X non-essential amino acids (Invitrogen), and were generally grown in T175 flasks or T175 triple flasks. When confluent, cells were washed with phosphate buffered saline (PBS; 10 mM phosphate buffer, 2.7 mM potassium chloride, 137 mM sodium chloride) (5 ml/T175 flask) split 1:3 using 1 ml trypsin/T175 flask to detach cells and grown to confluence again in 3 days. Vitamin K (10 µg/ml) was added to complete MEM media, at least 24 hours prior

expression of protein S. This allows cells to take up vitamin K which is essential for the  $\gamma$ -carboxylation of protein S. Protein S was expressed in OptiMEM I (Invitrogen) reduced serum-medium supplemented with 10 µg/ml vitamin K and 100 mg/L CaCl<sub>2</sub> for 3 days.

#### 2.2.2 Transient transfection of HEK293T cells

Protein S was transiently expressed in HEK293T cells. Cells were cultured in T175 triple flasks and grown in complete media. When cells reached ~60-70% confluency, the media was changed and supplemented with 10  $\mu$ g/ml vitamin K. Cells were grown in the vitamin K containing media for at least 24 hours before being transfected. On the day of transfection, for one T175-triple flask, 257  $\mu$ l of linear Polyethylenimine (PEI) (Polysciences,Inc) (1 mg/ml) were diluted into 3.75 ml of 0.15 M autoclaved NaCl. The resulting PEI solution was added drop wise to 3.75 ml of 0.15 M autoclaved NaCl containing 114  $\mu$ g DNA and the mix was incubated at room temperature for 15-20 minutes. The DNA-PEI solution was added to 75 ml OptiMEM supplemented with 10  $\mu$ g/ml vitamin K and 100 mg/L CaCl<sub>2</sub>. Each T175-triple flask was washed with PBS and cells were grown in the 82.5 ml of OptiMEM containing the transfection solution. Media containing the secreted protein S was harvested after 3 days.

#### 2.2.3 Stable transfection of HEK293 cells

A heterogeneous stable HEK293 cell line expressing WT protein S had already been generated in our lab by Dr S.M. Rezende (Universidade Federal de Minas Gerais, Brazil). Heterogeneous stable cell lines expressing protein S variants of interest were produced during this project. HEK293 cells were seeded in a 6 well plate in complete media and were allowed to reach 85-95% confluence before being transfected. Ten µl of Lipofectamine 2000 (Invitrogen) was mixed with 250 µl OptiMEM (GIBCO) and incubated for 5 minutes at room temperature. Following incubation, 4 µg of DNA pre-diluted into 250 µl OptiMEM were added drop-wise to the Lipofectamine 2000 solution. The resulting DNA-Lipofectamine 2000 mix was incubated for 5 minutes at room temperature before being added to the cells previously washed with PBS. Cells were incubated at 37°C, 5% CO<sub>2</sub> for 4-6

hours, after which the media was replaced by complete MEM media. After 24-48 hours cells were split 1:10 and seeded in complete MEM media containing 5  $\mu$ g/ml Blasticidine-HCI (Invitrogen). Stably transfected HEK293 cells were selected by culturing the cells in selective media at a confluence below 60% for 4 weeks. A stable cell line expressing full-length TFPI was also set up in HEK293 cells. These stably transfected cells were selected for with 500  $\mu$ g/ml G418 as the expression vector for TFPI contained the neomycin resistance cassette. TFPI in conditioned media was concentrated, quantified (by my colleague Verity Hockey) and used in the plasma assay (see section 2.5.3).

#### 2.2.4 Cryopreservation of mammalian cells

HEK293T, HEK293 and protein S stable cell lines were cryopreserved in liquid nitrogen. Once cells reached confluence in a T175 flask they were washed with PBS and trypsinised with 1 ml of trypsin. Trypsin was neutralised by adding 10-20 ml complete media and cells were centrifuged at 1200 rpm for 5-10 minutes. The supernatant was discarded and cells were resuspended in 2 ml of cold complete media. Two ml of cold complete media with 10% dimethyl sulfoxide (DMSO) was added dropwise to the cells and the solution was mixed. Cells were aliquoted in 1 ml fractions in cryovials and stored in a cryo freezing container (Nalgene) at -80°C for a minimum of 3 hours and maximum over night prior to being transferred into liquid nitrogen.

To recover cryopreserved cells, vials were transferred from the liquid nitrogen to ice and then thawed at 37°C. Cells were seeded into a T75 flask with 20-30 ml of complete media to dilute out the amount of DMSO.

#### 2.2.5 Expression and harvesting of WT and variant protein S

Both stable and transiently transfected cells were cultured in triple T175 flasks containing 75 ml of OptiMEM (GIBCO) supplemented with 10  $\mu$ g/ml vitamin K and 100 mg/L CaCl<sub>2</sub>. After 3-4 days the media was harvested, centrifuged and filtered to remove cell debris, and concentrated using a labscale tangential flow filtration (TFF) system (Millipore) with 10 kDa cut off membrane. The same procedure was

repeated for non-transfected HEK293T cells to obtain media, not containing protein S, to be used as a negative control in functional assays.

Stable cell lines were, in addition to being cultured in triple flasks, also cultured in CELLine adhere 1000 two-compartment bioreactors (IBS integra biosciences) according to the manufacturer's instructions. The cell compartment contained 15 ml OptiMEM, 10  $\mu$ g/ml vitamin K, 100 mg/L CaCl<sub>2</sub> and the media compartment contained 1 L complete MEM media supplemented with vitamin K. Media was changed weekly and OptiMEM from the cell compartment, containing protein S, was harvested.

Media samples to be used in functional assays were dialysed with at least 3 changes of buffer in 3-5 L of 20 mM Tris-HCl pH 7.5, 140 mM NaCl, for 1 hour each time at room temperature. Later during my project media samples (protein S variants 74A, KTK, NEDM and NNEDM) were dialysed in 20 mM Tris-HCl pH 7.5, 140 mM NaCl, 3 mM CaCl<sub>2</sub>. Samples were concentrated by Amicon Ultra centrifugal filter devices with a cut off of 50 kDa (Millipore), aliquoted and stored at -80°C.

### 2.3 Protein S purification

# 2.3.1 Protein S purification by FPLC on an anion exchange QFF column and by immunoaffinity chromatography

Protein S was purified by a two step method. Fast protein liquid chromatography (FPLC) was performed using an ÄKTApurifier UPC-10 (GE healthcare) and the Unicorn 5.1 (Build 407) software, strategy version 1.00 and consisted of a passage over an anion exchange Q Sepharose Fast Flow (QFF) column (section 2.3.1.1) followed by a second passage over an immunoimmunoaffinity column (section 2.3.1.3). This purification procedure had previously been optimised by Sofia Carlsson (in Professor Björn Dahlbäck's lab, Lund University, Malmö, Sweden).

# 2.3.1.1 Partial purification of protein S by FPLC on an anion exchange QFF column

Concentrated and dialysed conditioned media containing protein S supplemented with 4 mM ethylenediaminetetraacetic acid (EDTA) was applied to an anionic 5 mL of HiTrap QFF column (GE Healthcare) equilibrated with 20 mM Tris-HCI (pH 7.4), 100 mM NaCl, 4 mM EDTA, and 5 mM benzamidine-HCI (Sigma-Aldrich). The column was washed with 20 mM Tris-HCI (pH 7.4), 100 mM NaCl, and 5 mM benzamidine-HCI to first elute weakly bound proteins. Protein S was eluted from the column with 20 mM Tris-HCI (pH 7.4), 0.5 M NaCl, and 5 mM benzamidine-HCI. The Sepharose HiTrap<sup>™</sup> QFF anion exchange column was stripped of more tightly bound proteins with 20 mM Tris-HCI, pH 7.4, 1 M NaCl and 5 mM Benzamidine-HCI.

To prevent cross-contamination between protein S preparations, the column was cleaned with 2 M NaCl in a reversed flow direction for 15 minutes and 1 M NaOH for 1-2 hours. The column was then stored in 20% ethanol.

#### 2.3.1.2 Preparation of the immunoaffinity column

An immunoaffinity column for protein S was made with 2.5 mg of monoclonal mouse anti-human protein S MK21 antibody.<sup>243</sup> All monoclonal domain specific antibodies described in this thesis were kindly provided by Professor Björn Dahlbäck and Sofia Carlsson, Lund University, Malmö, Sweden and have been described in the literature.<sup>243</sup> The MK21 antibody is a calcium dependent antibody specific for the Gla domain of protein S. The antibody was thoroughly dialysed in the coupling buffer (0.2 M NaHCO<sub>3</sub>, 0.5 M NaCl, pH 8.3). The antibody was coupled to a 1 ml HiTrap NHS-activated HP column (GE healthcare) according to manufacturer's instructions and any excess active groups were deactivated with ethanolamine and acetate. The immunoaffinity column was stored in 20 mM Tris-HCl pH 7.5, 150 mM NaCl, 0.1% sodium azide.

#### 2.3.1.3 Purification of protein S by FPLC on an immunoaffinity column

To lower the ionic strength (from 500 mM NaCl to ~166 mM NaCl) and add CaCl<sub>2</sub>, two volumes of 7.5 mM CaCl<sub>2</sub> were added to the elution fraction from the anion

exchange QFF column, before injecting the sample on the immunoaffinity column. The fraction eluted from the QFF column containing protein S was applied to the equilibrated column (equilibration buffer: 50 mM Tris-HCl pH 7.5, 150 mM NaCl, 2 mM CaCl<sub>2</sub>). After washing with 1 M NaCl, the ionic strength was lowered by passing 3 column volumes of equilibration buffer over the column prior to elution of protein S with 50 mM Tris-HCl (pH 7.5), 150 mM NaCl, and 10 mM EDTA. An excess of CaCl<sub>2</sub> was present in the tubes collecting the eluted protein S to neutralise the EDTA. The column was regenerated by stripping with 0.1 M glycine-HCl pH 2.7.

Samples to be used in functional assays were dialysed and concentrated as described in section 2.2.5.

# 2.3.2 Protein S purification by barium citrate precipitation and by FPLC on an anion exchange DEAE column

#### 2.3.2.1 Barium citrate precipitation of protein S

Protein S was partially purified by barium citrate precipitation as previously described.<sup>248</sup> The conditioned media containing protein S that had previously been concentrated to 50 ml in the TFF (see section 2.2.5) was divided into two 50 ml tubes, each containing 25 ml media, and incubated on ice. 1.13 ml of a 0.5 M stock of trisodium citrate was added to each tube containing concentrated conditioned media and incubated on ice for 10 minutes. 1.13 ml of a 1 M stock BaCl<sub>2</sub> was added to each tube and the media was vortexed and incubated on ice for 1 hour. The sample was centrifuged at 5000 g for 30 minutes and the pellet that formed was resuspended in 25 ml of 100 mM BaCl<sub>2</sub>, 100 mM NaCl and incubated on ice for 1 hour. The sample was spun as before and the pellet was resuspended in 15-50 ml of 20 mM Tris-HCl pH 7.5, 150 mM NaCl, 150 mM trisodium citrate.

Barium citrate precipitated protein S was dialysed in 20 mM Tris-HCl pH 7.5, 140 mM NaCl for at least 1 hour at room temperature before dialysing it at 4°C.

#### 2.3.2.2 Purification of protein S by FPLC on an anion exchange DEAE column

The purification of protein S on a diethylaminoethyl cellulose anion exchange (DEAE) column was optimised by my colleague Dr Josefin Ahnström. Briefly, a 5 ml DEAE column (GE healthcare) was equilibrated with 20 mM Tris-HCl pH 7.5, 150 mM NaCl. Barium citrate precipitated protein S was loaded onto the column at 1-1.5 ml/minute and protein S was eluted with a CaCl<sub>2</sub> gradient from 0-20 mM. The anion exchange DEAE column was stripped of more tightly bound proteins with 20 mM Tris-HCl pH 7.5, 50 mM CaCl<sub>2</sub>, 1 M NaCl.

The column was cleaned and stored as described for the QFF column in section 2.3.1.1.

#### 2.4 Analysis and characterisation of protein S

#### 2.4.1 PNGase digestion

To determine whether the double band observed when running protein S on sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS-PAGE) was due to different glycosylation or cleavage, protein S was diluted to 8 ng/ul and a total of 10 µl were digested with 250 U of PNGase at 37°C over night.

#### 2.4.2 SDS-PAGE and Native PAGE

SDS-PAGE under reducing (30% β-mercaptoethanol) and non-reducing conditions was performed to assess the purity of the different protein S preparations. Samples were heated at 70-90°C for 5-10 minutes. Electrophoresis was performed on 10% polyacrylamide cast gels, 4-12% or 10% precasted NuPAGE® Novex® Bis-Tris gels (Invitrogen) at 200 V for 30-40 minutes.

Samples in a native state were run on NativePAGE<sup>™</sup> Novex<sup>®</sup> 3-12% precast Bis-Tris gels (Invitrogen), at 150 V for 90 minutes. Anode and cathode buffer were supplied by Invitrogen.

Gels were washed 3 x 5 minutes with water, stained for 1 hour with blue safe protein stain (Thermo scientific) and then destained in  $ddH_2O$ , stained by silver staining (see section 2.4.3) or used for Western blot analysis (see section 2.4.4).

#### 2.4.3 Silver staining

Electrophoresis was performed on SDS-PAGE gels as described in section 2.4.2. During all steps in the silver staining procedure (Table 2.6) volumes of 50 ml/gel were used and incubations were performed at RT with shaking.

Proteins were fixed by incubating the gel for 30 minutes in 50% methanol, 10% glacial acetic acid (buffer A). An additional incubation in 5% methanol (buffer B) was performed for 15 minutes. The gel was washed with water 3 x 5 minutes and incubated with sodium thiosulphate (buffer C) for 2 minutes. After a second wash, 3 x 30 seconds, the gel was stained with Silver Nitrate (buffer D). The gel was washed 3 x 1 minute and developed with buffer E for up to 10 minutes. The colour development was stopped with EDTA (buffer F).

Buffer	Composition				
A	50% CH <sub>3</sub> OH, 10% CH <sub>3</sub> COOH				
В	5% CH₃OH				
C (made up fresh)	1.26 mM Na <sub>2</sub> S <sub>2</sub> O <sub>3</sub> *5H <sub>2</sub> O				
D	169.9 mM AgNO <sub>3</sub>				
E (per 50 ml, made up fresh)	283 mM Na <sub>2</sub> CO <sub>3</sub> , 25 $\mu l$ of 37% HCOH, 1 ml solution C				
F	37.6 mM Na <sub>2</sub> EDTA				

Table 2.6 Composition of silver staining buffers.

#### 2.4.4 Western Blot

Western blots were performed by transferring proteins separated in unstained polyacrylamide gels (section 2.4.2) to a Hybond-ECL nitrocellulose membrane (Amersham Biosciences) or a PVDF-plus membrane (Osmonics INC) for SDS-PAGE and Native PAGE, respectively. Proteins were transferred to the hybond-ECL nitrocellulose membrane in a transfer buffer composed of 25 mM tris base, 190 mM glycine, 20% methanol. The PVDF-plus membrane was preincubated in methanol for 30 seconds, washed in water and immerged in the supplied transfer buffer (Invitrogen).

The transfer, for both types of membranes, was performed at 30 V for 60-75 minutes. Membranes were blocked with 4% milk for 40-60 minutes. Protein S was

detected with 20 ml of 0.588  $\mu$ g/ml polyclonal rabbit anti-protein S antibody (DAKO) for 40-60 minutes. Following a 3 x 5 minutes wash with PBS, 0.1% Tween, membranes were incubated with 20 ml of 0.175  $\mu$ g/ml peroxidise conjugated polyclonal goat anti-rabbit antibody (DAKO or Sigma Aldrich) for 40-60 minutes. The membrane was washed 4 x 5 minutes and developed with the chemiluminescent horseradish peroxidise (HRP) substrate Immobilon (Millipore) and Amersham Hyperfilm ECL (GE healthcare). Alternatively, to detect only  $\gamma$ -carboxylated protein S, 5  $\mu$ g/ml (20 ml) of a monoclonal mouse antibody directed against  $\gamma$ -carboxyglutamic residues (American Diagnostica INC) was used as a primary antibody and 20 ml of 2.352  $\mu$ g/ml polyclonal goat anti-mouse IgG HRP labelled antibody as a secondary antibody (DAKO).

#### 2.4.5 Quantification of total protein

To quantify the total protein concentration in protein S preparations the absorbance at 280 nm was read in a NanoDrop spectrophotometer. Beer-Lambert law for concentration c=A/  $\epsilon$  L with an extinction coefficient  $\epsilon$  1% 9.5 was used.<sup>161</sup> The total protein concentration was then compared with the Enzyme-linked immunosorbent assay (ELISA) results to determine the % purity of protein S.

#### 2.4.6 In-house ELISA with an in-house detection antibody

Protein S concentrations in conditioned media were determined using an inhouse ELISA. A polyclonal rabbit anti-protein S antibody (1  $\mu$ g/ml, DAKO) was immobilised onto 96-well Nunc Maxisorp microplates in 50 mM sodium carbonate buffer pH 9.6 at 4°C over night. All incubations were carried out in a plate shaker at 37°C for 1 hour unless stated otherwise. Washing steps were performed in triplicate with 250  $\mu$ l PBS 0.1% Tween between each step. Wells were blocked with PBS containing 3% bovine serum albumin (BSA, Sigma or Santa Cruz) for 2 hours. A standard curve, 0-2 nM protein S, was made by either diluting normal plasma or plasma purified protein S (Enzyme Research Laboratories). Normal plasma is known to contain 25  $\mu$ g/ml protein S. The reported MW of protein S ranges from 69-77 kDa.<sup>161,162</sup> In this thesis a MW of 69000 Da was assumed and

the concentration of protein S in normal plasma was calculated to be 362 nM. Protein S samples, 100  $\mu$ l, were incubated in the wells for 1 hour. Bound protein S was detected by 4 nM of a mouse monoclonal antibody directed against the EGF3-4 domain of protein S (MK55) followed by 8 nM of an HRP-conjugated goat anti-mouse antibody (DAKO). PBS 0.5% BSA was used as buffer in which to dilute protein S and the antibodies. The plate was developed with 100  $\mu$ l/well chromogenic substrate o-phenylenediamine dihydrochloride (OPD, Sigma), the enzymatic reaction was stopped with 50  $\mu$ l/well 2 M H<sub>2</sub>SO<sub>4</sub> and the absorbance was read at 492 nm. Data was analysed with GraphPad Prism 4.03 and data points were fitted to a sigmoidal dose response curve. Intra ELISA variability was determined by calculating the coefficient of variation (CV) between the same sample diluted to 2-3 different final concentrations in the same ELISA. Inter ELISA variability was determined by calculating the CV obtained for the same sample in (3-10) different ELISA.

#### 2.4.7 In-house ELISA with commercially available antibodies

An in-house ELISA with commercially available antibodies was also developed to avoid dependence upon the availability of the MK55 antibody generously supplied by Professor Björn Dahlbäck. The assay was set up as described for the in-house ELISA in section 2.4.6 with the exception of 20 mM Tris-HCl pH 7.5, 140 nM NaCl, 3 mM CaCl<sub>2</sub> being used as a buffer and that protein S was detected with 4 nM of the mouse monoclonal antibody AHPS-5092 (Haematologic Technologies Inc).

#### 2.4.8 Preparation of phospholipid vescicles

Activated platelets or cells exposing negatively charged phospholipids such as phosphatidylserine are required *in vivo* for coagulation to occur. In *in vitro* assays these are often replaced by phospholipid vescicles. Phospholipid vescicles were prepared during this project to be used as a procoagulant surface in functional assays and to assess protein S binding to phospholipids.

All phospholipids were from Avanti Polar Lipids Inc. Phospholipid mixtures in chloroform were prepared and the chloroform evaporated under a nitrogen

stream. The phospholipids were resuspended in 20 mM Tris-HCl pH 7.5, 140 mM NaCl for the plasma and FXa inhibition assay or in 25 mM HEPES (N-2hydroxyethylpiperazine-N'-2-ethanesulfonic acid) pH 7.7, 150 mM NaCl (HN) for the FVa inactivation assays. Unilamellar phospholipids vesicles were obtained either by sonicating the phospholipids in an ice container at amplitude 22% for 7 minutes or by extruding them as previously described.<sup>141</sup> Briefly, extrusion was performed by passing the phospholipid mixture through a 100 nm polycarbonate Synthetic phospholipids 1,2-Dioleoyl-sn-glycero-3membrane 19 times. phosphocholine (DOPC), 1,2-Dioleoyl-sn-glycero-3-phosphoserine (DOPS), and 1,2-Dioleoyl-sn-glycero-3-phosphoethanolamine (DOPE) were used in the plasma assay, the plate binding assay and the FXa inhibition assay. Natural (PS, phospholipids L-α-phosphatidylserine brain extract), L-αphosphatidylethanolamine (PE, egg extract) and L- $\alpha$ -phosphatidylcholine (PC, egg extract) were used in the FVa inactivation and prothrombinase assays.

#### 2.4.9 Binding of protein S to phospholipids

As binding of protein S is a prerequisite for its function, binding of protein S and protein S variants to phospholipids was assessed.

Phospholipids vesicles (DOPS/DOPC/DOPE, 20:60:20), 25 µg/ml, were coated onto 96-well Nunc Maxisorp microplates in 50 mM sodium carbonate buffer pH 9.6 at 4°C over night. Washing steps were performed in triplicate with 250 µl/well 20 mM Tris-HCl pH 7.5, 140 mM NaCl, 5 mM CaCl<sub>2</sub>, 0.3% BSA between each step and incubations were carried out in a plate shaker at 37°C. The same buffer was used to dilute protein S and the antibody in. Wells were blocked with 20 mM Tris-HCl pH 7.5, 140 mM NaCl, 5 mM CaCl<sub>2</sub>, 3% BSA for 2 hours. Protein S or protein S variants, 0-120 nM, partially purified by barium citrate precipitation, were incubated in the wells in duplicate for 2 hours and were detected with an HRP-conjugated anti-protein S antibody (Affinity Biologicals) for 45 minutes. Controls were performed by assessing binding of protein S to phospholipids in the presence of 10 mM EDTA and by assessing aspecific binding to the plate in the absence of phospholipids and presence of CaCl<sub>2</sub>. The plate was developed, stopped and read as previously described for the ELISA. Data was analysed with GraphPad Prism 4.03 and curves were fitted to a one site binding hyperbola. The

 $K_{d(app)}$  for protein S binding to the phospholipids were derived and that of protein S variants was compared with that of WT protein S.

#### 2.4.10 Binding of protein S to domain specific monoclonal antibodies

To assess the correct folding of protein S variants, their binding affinity to domain specific monoclonal antibodies was compared to that of WT protein S.

Domain specific monoclonal antibodies MK21 (Gla domain), MK54 (EGF1 domain) and MK61 (SHBG domain), 1  $\mu$ g/ml, were coated on 96-well Nunc Maxisorp microplates as described for the ELISA. Washing steps were performed in triplicate with 250  $\mu$ l 20 mM Tris-HCl pH 7.5, 140 mM NaCl, 3 mM CaCl<sub>2</sub>, 0.1% Tween between each step. Incubations were carried out in a plate shaker at 37°C for 1 hour unless stated otherwise. Wells were blocked with 200  $\mu$ l 20 mM Tris-HCl pH 7.5, 140 mM NaCl, 3 mM CaCl<sub>2</sub>, 3% BSA for 2 hours. Protein S, 0-160 nM was incubated in the plate for 1 hour and was detected by a rabbit polyclonal antibody against protein S followed by a goat anti-rabbit HRP-conjugated antibody. Protein S and the antibody were diluted in a Tris-HCl pH 7.5, 140 mM NaCl, 3 mM CaCl<sub>2</sub>, 0.5% BSA buffer. The plate was developed, stopped and read as previously described for the ELISA. Data was analysed with GraphPad Prism 4.03 and data was fitted to a one site binding hyperbola. The K<sub>d(app)</sub> for protein S binding to the monoclonal antibodies were derived and that of protein S variants was compared with that of WT protein S.

### 2.5 CAT assay

#### 2.5.1 Overview of CAT

CAT is an assay that allows measurement of thrombin generation. It was used during this project to evaluate the different cofactor activity of protein S and protein S variants toward APC and TFPI. It is a plasma-based assay in which a coagulant response is initiated by the addition of TF, CaCl<sub>2</sub> and phospholipids and the amount of thrombin generated is measured through the cleavage of its fluorogenic substrate over time and quantified by reference to a standard calibrator. Phospholipids are added into the reaction to provide a surface on which coagulation can occur once triggered by TF. As the plasma is citrated and most coagulation factors are calcium dependent the reaction does not start until calcium is added into the reaction. Calcium is added together with Z-Gly-Gly-Arg-AMC (Bachem), a fluorogenic substrate of thrombin. Thrombin generation is described and quantified by the lag time (which approximates the clotting time), peak height (maximal velocity of thrombin generation) and the endogenous thrombin potential (ETP) which is the total amount of thrombin generated over time and corresponds to the area under the curve.

#### 2.5.2 Evaluation of the protein S-APC cofactor activity

Protein S deficient plasma (Affinity Biologicals), 80  $\mu$ L, was incubated with 65  $\mu$ g of corn trypsin inhibitor (CTI, Haematologic Technologies Inc) per ml of plasma to inhibit contact activation, 50  $\mu$ M phospholipid vesicles (DOPS/DOPC/DOPE, 20:60:20), 1 pM TF (Dade Innovin; Dade Behring), 4 to 16 nM APC (Enzyme Research Laboratories) with 0-120 nM protein S, in a final volume of 100  $\mu$ L (all concentrations are final). A polyclonal anti-TFPI antibody (Haematologic Technologies Inc), 100 nM, was used to inhibit any protein S cofactor activity toward TFPI. Polyclonal antibodies against protein S (1500 nM, DAKO) and protein C (130 nM, Sigma-Aldrich or Haematologic Technologies Inc) were used as further controls. All conditions were performed in duplicate and assays were performed at least twice. Thrombin generation was assessed using a Fluoroscan Ascent FL Plate Reader (Thermo Lab System) in combination with the Thrombinoscope software (SYNAPSE BV).

Thrombin generation was initiated by automatic dispensation of 20  $\mu$ L of 2.5 mM Z-GlyArg-AMC-HCl (Bachem), 2.5% Me<sub>2</sub>SO, 20 mM Tris-HCl (pH 7.5), 60 mg/mL BSA, 100 mM CaCl<sub>2</sub> into each well. The reaction was performed at 37°C and measurements were performed with an excitation and emission wavelength of 390 nm and 460 nm, respectively. To compare the cofactor activity of protein S and protein S variants towards APC, the changes in peak height and ETP were evaluated.

#### 2.5.3 Evaluation of the protein S-TFPI cofactor activity

To evaluate the ability of protein S to enhance the anticoagulant activity of TFPI, protein S deficient plasma (Affinity Biologicals, Hyphen Biomed or Enzyme Research Laboratories) was incubated with CTI, phospholipids, TF (as described in section 2.5.2), 0-10 nM TFPI (either purified, kind gift of the Chemoserotheraputic Institute Kaketsuken, or in concentrated conditioned media) and 0-120 nM protein S or protein S variants in a final volume of 100 µL (all concentrations are final). A polyclonal antibody against protein C (130 nM, Sigma-Aldrich or Haematologic Technologies Inc) was included, as required, to inhibit protein S cofactor activity towards any protein C activated during the assay. To show the specificity of the system polyclonal antibodies against protein S (1500 nM, DAKO) or polyclonal antibodies against TFPI (100 nM, Haematologic Technologies Inc) were used. All conditions were performed in duplicate and assays were performed at least twice. Thrombin generation was assessed as described in section 2.5.2. To compare the cofactor activity of protein S and protein S variants towards TFPI, the changes in peak and lag time were evaluated.

#### 2.6 FVa inactivation assay

#### 2.6.1 Overview of the FVa inactivation assay

The FVa inactivation assay is an assay with pure components that was used during this project to assess the ability of protein S and protein S variants to enhance APC mediated FVa inactivation. FVa inactivation by APC is indirectly measured by performing a prothrombinase assay. The prothrombinase assay relies on the amount of FVa that has not been inactivated by APC and its ability to form a complex with FXa and activate prothrombin. The amount of thrombin generated is then measured by the rate at which it cleaves its chromogenic substrate S-2238 (Chromogenix).

#### 2.6.2 FV activation

To obtain a source of FVa to assess APC mediated FVa inactivation, FV was first activated. A previously prepared and reported<sup>125</sup> double variant of FV, FV R506Q/R679Q, was activated with 0.5 U/ml human thrombin at 37°C for 10 minutes. This variant can not be cleaved in Arg506 and Arg679 allowing evaluation of cleavage at Arg306 by APC which is the site mainly enhanced by protein S (see section 1.4.3.1.1). The reaction was stopped by the addition of hirudin (5 U/ml, final concentration). The activated FVa variant was used as a substrate for APC to determine the ability of protein S to enhance cleavage of Arg306 of FVa by APC.

#### 2.6.3 FVa inactivation

The FVa inactivation was performed either by evaluating the cofactor activity of increasing amounts of protein S (0-100 nM) at 10 minutes or by evaluating the cofactor activity of a fixed concentration (50 nM) of protein S at different timepoints (0-20 minutes).

The highly purified protein S preparations, 0-100 nM, were incubated with 0.5 nM APC, 25  $\mu$ M phospholipids vesicles (PS/PC/PE, 10:70:20) and 0.8 nM FVa R506Q/R679Q in HN, 5 mM CaCl<sub>2</sub>, 5 mg/ml BSA (HNBSACa<sup>2+</sup>, only high purity grade BSA from Sigma A7030 was used) in a total volume of 50  $\mu$ l (all concentrations are final). The solution was incubated at 37°C for 10 minutes and the reaction was stopped by performing a 1:25-fold dilution in ice cold HNBSACa<sup>2+</sup>. The remaining FVa activity was measured in a prothrombinase assay.

Time course experiments were performed to derive the rate constant of FVa cleavage. APC, 0.5 and 3 nM, and 50 nM protein S were used and aliquots were quenched at intervals between 0-20 minutes. The remaining FVa activity was measured in a prothrombinase assay. Controls containing FVa R506Q/R679Q in the absence of APC either in the presence or absence of protein S and protein S variants were performed. To calculate the pseudo-first order rate constant for

APC mediated cleavage at Arg306, the previously reported equation (Eq2)<sup>141</sup> was used:

Eq2.  $Va_t = Va_0^* e^{-k306^*t} + C^* Va_0^* (1 - e^{-k306^*t})$ 

 $Va_t$  represents the cofactor activity determined at time *t*,  $Va_0$  is the cofactor activity determined at time point 0, *C* is the remaining procoagulant cofactor activity of FVa after cleavage at position 306, and *k*306 is the rate constant of cleavage at position 306.

#### 2.6.4 Prothrombinase assay

The amount of FVa inactivated was indirectly measured by performing a prothrombinase assay. An aliquot, 25  $\mu$ l, of the FVa inactivation reaction was incubated with phospholipid vesicles (PS/PC 10:90) and FXa in the presence of CaCl<sub>2</sub>. The reaction was initiated by the addition of prothrombin, in a final volume of 125  $\mu$ l. The buffer contained HN and 0.5 mg/ml ovalbumin. Final concentrations were 50  $\mu$ M PS/PC, 5 nM FXa and 0.5  $\mu$ M prothrombin. The solution was incubated at 37°C for 2 minutes and the reaction was terminated by an 8-fold dilution in 50 mM Tris-HCl pH 7.9, 100 mM NaCl, 20 mM EDTA, 1% PEG 6000. The amount of thrombin generated was measured by cleavage of its chromogenic substrate S-2338 (Chromogenix) at 405 nm for 15 minutes at 30 seconds intervals. The cofactor activity of protein S and protein S variants towards APC was compared.

#### 2.7 FXa inhibition assay

#### 2.7.1 Overview of the FXa inhibition assay

The FXa inhibition assay was performed to assess the different cofactor activity of protein S and protein S variants towards TFPI mediated inhibition of FXa. In the FXa inhibition assay the ability of FXa to cleave its chromogenic substrate S-2765 is measured over time. The ability of TFPI to inhibit FXa and of protein S to enhance the TFPI anticoagulant activity is measured as a change in the rate of

substrate proteolysis. The cleavage of the chromogenic substrate S-2765 was measured for up to 1 hour at 1 minute intervals.

#### 2.7.2 FXa inhibition assay

Purified human FXa (final concentration of 0.5 nM; Hyphen Biomed) was added to wells already containing the chromogenic substrate S-2765 (200  $\mu$ M; Chromogenix), phospholipids (10  $\mu$ M), CaCl<sub>2</sub> (5 mM), in the presence or absence of varying concentrations of purified TFPI (0-10 nM, kind gift of the Chemoserotheraputic Institute Kaketsuken) and protein S or protein S variants (0-160 nM). The reactions were set up in a buffer containing 20 mM Tris-HCl buffer pH 7.5, 140 mM NaCl, 5 mg/ml BSA (only high purity grade BSA from Sigma A7030 was used), with a final volume/well of 100  $\mu$ l. The hydrolysis of S-2765 by FXa was followed over time by measuring the A<sub>405</sub> at regular intervals for up to 1 hour at room temperature. A control was obtained by incubating FXa and S-2765 in the absence of TFPI, in the presence or absence of WT or variant protein S. A blank was obtained by including TFPI and S-2765 but not FXa. All conditions were performed in duplicate and assays were repeated at least twice.

### 2.7.3 Kinetic data analysis of FXa inhibition by TFPI in the presence and absence of protein S

To calculate the inhibition constant of TFPI for FXa in the presence and absence of protein S, 0-8 nM TFPI  $\pm$  100 nM protein S were used. The assay was either set up as described in section 2.7.2 by starting the assay with the addition of FXa or, alternatively, FXa was present in the wells and the reaction was started with the addition of TFPI and S-2765.

Data was fitted with the previously described equation (Eq3):<sup>249</sup>

Eq3.  $A_t = A_0 + V_s t + (V_0 - V_s) \{1 - exp(-k_{obs}t)\}/k_{obs}$ 

Where t represents time,  $A_t$  represents the absorbance at time point t,  $A_0$  the absorbance at time zero,  $V_0$  the initial velocity,  $V_s$  the final steady state velocity of

S-2765 conversion and  $k_{obs}$  the apparent rate constant for the transition from  $V_0$  to  $V_{s}.$ 

The maximum initial velocity  $V_{max}$  (FXa conversion of S-2765 in the absence of TFPI) was divided by the initial velocities at each time point and the resulting  $V_{max}/V_0$  value was plotted against the concentration of TFPI. Values were fitted with a linear regression and the x intercept was used to calculate the K<sub>i</sub> value with the following formula (Eq4):

Eq4.  $K_i = -x_{intercept}/(1+[S]/K_m)$ 

Where [S] represent the concentration of the chromogenic substrate and  $K_m$ , the Michaelis-Menten constant, the concentration of substrate that results in half maximal cleavage by FXa.

### 3 RESULTS

The results of this thesis are divided into 4 sections. The first (section 3.1) describes the mutagenesis, expression and purification of protein S. The protein S obtained either in concentrated conditioned media or purified was then quantified by ELISA and analysed for multimeric content and cleavage (section 3.2). In the final two sections, 3.3 and 3.4, the cofactor activity of protein S and protein S variants towards APC and TFPI respectively are evaluated.

#### 3.1 Mutagenesis, expression and purification of protein S

#### 3.1.1 Selection and generation of protein S variants

Based on the domains of protein S known from the literature to be important for APC and TFPI cofactor activity (outlined in section 1.5.1 and 1.5.2) protein S variants with substitutions in the Gla, TSR, EGF1 and EGF2 were produced. Composite variants in these N-terminal domains produced during this thesis were; protein S Face1, Face2, Gla1, TSR, NEDM, NNEDM and KTK. Twelve additional alanine point variants in the EGF1 domain and 12 alanine point variants in the EGF2 domain of protein S were made available through collaboration with Dr S.M. Rezende (Universidade Federal de Minas Gerais, Brazil). Additional point variants expressed during this project are protein S N74A, D78A, Q79A and D95N resulting in a total of 35 variants covering 49 amino acid residues of protein S.

Table 3.1 lists the variants produced during this project and indicates in which domain the substitutions have been introduced. Single point variants were named according to the residue that was mutated. The name given to the composite variant is shown in Table 3.1.

Table 3.2 shows the N-terminal amino acid residues of protein S highlighting those that have been mutated during this project. Variants were tested both for their APC cofactor activity and their TFPI cofactor activity.

Domain	Name of variant	Residues mutated				
Gla	Face1	N33S/P35T/E36A/Y39V				
Gla	Face2	L21T/ N23S/K24Y/R28F/D34S/Y41W/L45T				
Gla/EGF1	Gla1	D38N/V46G/K97Q				
TSR/EGF1	TSR	R49Q/R60Q/D68N/R70Q/D78N				
TSR		N74A				
EGF1	NEDM	N86A/E87Q/D88A/M91A				
EGF1	NNEDM	N74A/N86A/E87Q/D88A/M91A				
EGF1	КТК	K94A/T103A/K105A				
EGF1 (from Dr S.M. Rezende)		K94,D95,K97,S99,T101,T103, K105,P106,W108,Q109,E111,K112				
EGF1		D78A,Q79A,D95N				
EGF2 (from Dr S.M. Rezende)		D135,N136,T137,Y141,H142,S144, K146,S153,N154,K155,K156,D157				

Table 3.1 Protein S variants produced during this thesis.Single or point variants of protein S were produced by site-directed mutagenesis targetingresidues located in the Gla domain, TSR domain, EGF1 or EGF2 domain.

Aa nr	30	) 40	) 9	50 60	) 7(	0 80	) 9(	) 1	00 11	.0 1	20
	GLA DOMAIN		TSR 1	DOMAIN		EGF1 I	DOMAIN ——			EGF2	Domain
WT protein S	LCNKEEAREV	FENDPETDYF	YPKYLVCLRS	FQTGLFTAAR	QSTNAYPDLR	SCVNAIPDQC	SPLPCNEDGY	MSCKDGKASF	TCTCKPGWQG	EKCEFDINEC	:
Proth WT	TCSYEEAFEA	LESSTATDVF	<b>wa</b> kyta								
Face1	LCNKEEAREV	FE <mark>S</mark> D <b>TA</b> TD <mark>V</mark> F	YPKYLVCLRS	FQTGLFTAAR	QSTNAYPDLR	SCVNAIPDQC	SPLPCNEDGY	MSCKDGKASF	TCTCKPGWQG	EKCEFDINEC	
Face2	TC <mark>SY</mark> EEA <mark>F</mark> EV	FEN <mark>S</mark> PETDYF	<mark>w</mark> pky <mark>t</mark> vclrs	FQTGLFTAAR	QSTNAYPDLR	SCVNAIPDQC	SPLPCNEDGY	MSCKDGKASF	TCTCKPGWQG	EKCEFDINEC	
<mark>Gla1</mark>	LCNKEEAREV	FENDPET <mark>N</mark> YF	YPKYL <mark>g</mark> CL <b>r</b> S	FQTGLFTAA <b>R</b>	QSTNAYP <b>D</b> LR	SCVNAIPDQC	SPLPCNEDGY	MSCKDG <mark>Q</mark> ASF	TCTCKPGWQG	EKCEFDINEC	
TSR	LCNKEEAREV	FENDPETDYF	ypkylvcl <mark>q</mark> s	FQTGLFTAA <mark>Q</mark>	QSTNAYP <mark>N</mark> L <mark>Q</mark>	SCVNAIP <mark>N</mark> QC	SPLPCNEDGY	MSCKDGKASF	TCTCKPGWQG	EKCEFDINEC	
TSR pm	LCNKEEAREV	FENDPETDYF	YPKYLVCL <b>R</b> S	FQTGLFTAA <b>r</b>	QSTNAYP <b>D</b> LR	scv <mark>a</mark> aip <b>d</b> qc	SPLPCNEDGY	MSCKDGKASF	TCTCKPGWQG	EKCEFDINEC	
NEDM	LCNKEEAREV	FENDPETDYF	YPKYLVCLRS	FQTGLFTAAR	QSTNAYPDLR	SCVNAIPDQC	SPLPC <mark>AQN</mark> GY	<mark>a</mark> sckdgkasf	TCTCKPGWQG	EKCEFDINEC	
ктк	LCNKEEAREV	FENDPETDYF	YPKYLVCLRS	FQTGLFTAAR	QSTNAYPDLR	SCVNAIPDQC	SPLPCNEDGY	MSC <mark>A</mark> DGKASF	TC <mark>A</mark> C <mark>A</mark> PGWQG	EKCEFDINEC	
EGF1 pm	LCNKEEAREV	FENDPETDYF	YPKYLVCLRS	FQTGLFTAAR	QSTNAYPDLR	scvnaip <mark>aa</mark> c	SPLPCAQNGY	MSC <mark>AA</mark> G <mark>A</mark> AAF	<mark>acacaa</mark> g <mark>aa</mark> g	<mark>AA</mark> CEFDINEC	
Aa nr	130	) 140	) 19	50 16	50						
	EGF2 DOMAIN	v ——— v		——————————————————————————————————————							
WT protein S	KDPSNINGGC	SQICDNTPGS	YHCSCKNGFV	MLSNKKDCKD							
EGF2 pm	KDPSNINGGC	sqic <mark>aaa</mark> pgs	<mark>aa</mark> c <mark>a</mark> cangfv	ML <mark>AAAAA</mark> CKD							

#### Table 3.2 N-terminal amino acid sequence of protein S, prothrombin and protein S variants.

Amino acid residues of the Gla domain of protein S that are not conserved in prothrombin were swapped for those of prothrombin in the composite variants Face1 and Face2. No mutations in the first 20 amino acid residues of protein S were made in an attempt to not disrupt phospholipids binding. Gla1 and NEDM were made by mutating residues into similar non charged residues. TSR was created by substituting charged residues in the TSR domain with polar residues. KTK and all EGF1 and 2 point variants were mutated to alanine. NNEDM (not shown) were made by introducing the N74A substitution into NEDM. Domains are labelled below the Aa nr. Amino acids in the TSR and EGF2 domain are in italics in WT protein S. Aa nr, amino acid number; Proth, prothrombin; pm, point variants.

Protein S Face1 and Face2 variants, previously evaluated for their APC cofactor activity by Saller *et al.*<sup>224</sup> are chimeras of the human protein S Gla domain and the human prothrombin Gla domain. These were produced in our lab by Ms Y. Yu by inverse PCR (KOD Hot Start DNA Polymerase Kit, Novagen) and site-directed mutagenesis (QuikChange site-directed mutagenesis Kit, Stratagene).

To identify residues in the Gla and EGF1 domain that might interact with APC or TFPI, these domains were aligned with homologous domains of other coagulation factors. Non conserved amino acids were selected and clusters of amino acids with surface exposed R groups were selected. This was done by using the structural protein S model spanning the Gla domain to the EGF1 domain created by Villoutreix *et al.*.<sup>193</sup>

As the TSR is unique to protein S, no swap substitutions could be made. Hence, charged residues in the TSR were substituted with the most similar non-charged amino acid (R to Q and D to N), to minimize the risk of adverse conformational changes, resulting in the generation of the TSR variant.

Dr S.M. Rezende (Universidade Federal de Minas Gerais, Brazil) mutated residues between the third/fourth and fifth/sixth Cys residues of EGF1 and EGF2 as these regions on the EGF-like domains are likely to be involved in protein interaction as suggested by (i) the highly divergent sequence of the B region; (ii) the association of dysfunctional proteins with natural *PROS1* mutations in these regions and (iii) the location of many mutations in these regions associated with dysfunctional proteins in genes coding for other EGF-containing coagulation proteins, such as FVII, FIX and protein C.<sup>202,250</sup>

All mutations were confirmed by sequencing. Subcloned fragments were also sequenced across the ligation sites to confirm correct insertion.

#### 3.1.2 Expression of protein S

Protein S was expressed by transient or stable transfection of HEK293T or HEK293 cells, respectively. The amount of protein S harvested from HEK293 cells over time did not result in a decreased expression according to ELISA results, suggesting that all cells were stably transfected after the initial 4 weeks selection with blasticidine.

WT protein S expressed approximately at 1.5-2  $\mu$ g/ml, as determined by ELISA, in triple flasks and on average at 60  $\mu$ g/ml in the CELLine adhere 1000 twocompartment bioreactor, with expression levels increasing over time in the latter. The high concentration obtained in the bioreactor is a consequence of harvesting the media after a week rather than after 3 days and of the bioreactor concentrating protein S in the cell compartment. Expression levels of the variants were not formally assessed against that of WT protein S in a dedicated experiment where number of cells and transfection efficiency were taken into account. However, the great majority of variant protein S preparations expressed well (1-2  $\mu$ g/ml), with the notable exceptions of protein S T108A and protein S N136A, which expressed at lower than 0.5  $\mu$ g/ml. Protein S expressed from transient transfections yielded up to 10 times less than protein S expressed from a stable cell line. Harvested protein S media was either dialysed, concentrated and assayed directly, or purified by FPLC.

### 3.1.3 Purification of protein S by FPLC on an anion exchange QFF column and immunoaffinity column

Harvested and concentrated media was supplemented with EDTA and protein S was partially purified on a Sepharose HiTrap QFF anion exchange column (elution buffer: 20 mM Tris-HCl pH 7.4, 0.5 M NaCl, 5 mM Benzamidine). Figure 3.1A (left chromatogram) illustrates this purification step on WT protein S with the milli absorbance unit (mAU) at 280 nm displayed against the volume of liquid flowing through the column. As expected, it shows that large amounts of protein (mAU values of ~260) are present in the flowthrough, although these are diluted in a large volume (~20 ml in this representative chromatogram). A high and narrow elution peak was obtained, while very small washing and stripping peaks could be observed. Fractions of the flowthrough, washing, elution and stripping peaks from the anion exchange chromatography were analysed by SDS-PAGE. Proteins in the respective fractions were stained with Blue Safe protein stain (Figure 3.1B). Protein S was specifically detected by an anti-protein S antibody on a Western blot (Figure 3.1C). The stained SDS-PAGE gel shows a large amount of protein is present both in the flowthrough and elution fraction (Figure 3.1, Panel B, first 4 lanes after the marker). The bands of protein in the flowthrough fraction
are quite faint as proteins were diluted into a large volume (flowthrough volume of ~20 ml). A large protein band is present in the elution fraction that migrates with the same mobility as commercially available plasma purified protein S (last lane of gel, labelled as Ctrl). Western blot analysis (Figure 3.1, Panel C, first 4 lanes) confirmed that protein S was present in the elution fraction. A faint band of higher molecular weight was also present in the elution fractions. A higher MW band was sometimes visualised on gels when high amounts of purified protein S was loaded and is likely to correspond to aspecific binding of the secondary antibody. Protein S was also detected in the flowthrough of the QFF anion exchange column using Western Blot (Figure 3.1C). According to ELISA results, the amount of protein S lost in the flowthrough and washing peak varied from batch to batch but was generally between 5-15%, while the majority of protein S (~80%) was found in the elution fraction. This is compatible with what can be seen in the elution fraction of the Western blot (Figure 3.1C). Protein S content and total amount of protein in the elution peak were measured, respectively, by ELISA and total protein assay and suggested a mean purity of ~25%.

To obtain high purity protein S the elution fraction from the Sepharose HiTrap QFF anion exchange column was applied on an immunoaffinity column (elution buffer: 50 mM Tris-HCl pH 7.5, 150 mM NaCl, 10 mM EDTA). A representative chromatogram is shown in Figure 3.1A (right chromatogram). Fractions were collected and the flowthrough, washing peak, elution peak and stripping peak were analysed by SDS-PAGE. The proteins were visualised by Blue Safe protein staining (Figure 3.1B, lane 5-8 after the marker); protein S was specifically detected by polyclonal anti-protein S antibodies following Western blot (Figure 3.1C, last four lanes). The elution fraction from the immunoaffinity column in the stained SDS-PAGE gel (Figure 3.1B) shows a single band migrating with the same mobility as plasma purified protein S (last lane of gel, labelled as Ctrl). Western blot analysis (Figure 3.1C) confirms that this is protein S, suggesting pure protein S could successfully be isolated following this two step purification procedure. It is observed that the majority of contaminants visualised in the elution fraction of the QFF anion exchange column (Figure 3.1B) are present in the flowthrough of the immunoaffinity column. Protein S was, however, also present in the flowthrough fraction and washing fraction of the immunoaffinity column (Figure 3.1C). It should be noted that ten fold more in terms of volume of

the washing fraction was loaded in comparison with the other fractions to determine whether this contained any protein S. The high amount of protein S in the flowthrough fraction partially explains why the absorbance on the chromatogram of this fraction is so high in comparison with that of the elution fraction (considering ~25% of the proteins loaded onto the immunoaffinity column should be protein S). However, the high absorbance in the flowthrough in comparison to the elution peak is also due to the presence of Benzamidine in the loaded sample (see section 2.3.1.1) as this absorbs at 280 nm. Benzamidine is, on the other hand, not present in the elution buffer.



Figure 3.1 Purification of WT protein S by FPLC on a Sepharose HiTrap QFF anion exchange column followed by an immunoaffinity column and analysis of elution fractions. (A) Purification of protein S on sepharose QFF anion exchange purification (left chromatogram) followed by immunoaffinity purification (right chromatogram). The 280 nm absorbance (blue line) is plotted against the elution volume. The vertical green lines represent a change of buffer. Total protein content in the fractions was detected by Blue safe protein stain of an SDS-PAGE (B) and protein S was specifically detected with a polyclonal antibody against protein S (C). F, flowthrough; W, washing; E, elution; S, stripping.

To evaluate why so much protein S was present in the flowthrough fraction from the immunoaffinity column, y-carboxylation of protein S was firstly assessed as the monoclonal MK21 antibody used for the purification recognises the Gla domain of protein S. Importantly, when fractions from the Sepharose HiTrap QFF anion exchange column were analysed by Western blot with a monoclonal anti-Gla antibody (recognising only y-carboxylated protein S), only protein S in the elution fraction was recognised by both antibodies (compare Figure 3.2, detection with monoclonal anti-Gla antibody with Figure 3.1C, detection with polyclonal antibody). This suggests that, non y-carboxylated protein S is in the flowthrough, indicating that this method selectively purifies  $\gamma$ -carboxylated protein S. This was thought to be of considerable importance, as only y-carboxylated protein S is functional. A consequence of this is that all protein S loaded onto the immunoaffinity column was  $\gamma$ -carboxylated. The finding that protein S in the elution and flowthrough fractions from the immunoaffinity column was ycarboxylated was confirmed by recognition by the monoclonal anti-Gla antibody (Figure 3.1C, detection with polyclonal antibody and Figure 3.2, detection with monoclonal anti-Gla antibody). Ten fold more of the washing fraction was loaded in the gel where protein S was detected with the polyclonal antibody in comparison with the one in which it was detected with the monoclonal antibody recognising only y-carboxylated Gla domain. However, when ten fold more of the washing fraction was loaded (not shown) the washing fraction was recognised by the anti-Gla antibody.



## Figure 3.2 Western blot analysis of protein S in the fractions obtained during purification on the Sepharose HiTrap QFF and on the immunoaffinity column.

Flowthrough (F), washing (W), elution (E) and stripping (S) peaks of, respectively, Sepharose HiTrap<sup>TM</sup> QFF purification (lane 1-4) and affinity purification (lane 5-8) were blotted with a monoclonal anti-Gla antibody recognising only  $\gamma$ -carboxylated Gla domains.

The finding of a single band in the elution fraction of the immunoaffinity column (Figure 3.1B), together with its identification as protein S by Western blot (Figure 3.1C), demonstrates that pure protein S could successfully be isolated using this two-step purification method. The results of Figure 3.2 indicate that this purified protein S was y-carboxylated. However, a large part of y-carboxylated protein S was also present in the flowthrough of the immunoaffinity column. To investigate this further, a similar sample containing protein S was applied to the immunoaffinity column continuously over night at a greatly reduced flow rate of 0.5 ml/minute to increase contact time. However, this did not appreciably increase the amount of protein S in the elution fraction. When the flowthrough fraction was reapplied to the immunoaffinity column, additional protein S could be purified and eluted with EDTA (results not shown), suggesting the protein S in the first flowthrough was more likely associated with low binding capacity of the immunoaffinity column. Consequently, the flowthrough was reapplied to the column until all protein S was purified and was no longer in the flowthrough. All fractions from the elution peak from the immunoaffinity column containing protein S were pooled, dialysed and concentrated on spin columns. This purification procedure was used during the early part of my work to purify WT protein S, protein S D95A, protein S Face2 and protein S TSR.

# 3.1.4 Purification of protein S by barium citrate precipitation and by FPLC on an anion exchange DEAE column

Due to the low binding capacity of the immunoaffinity column, an alternative purification method for protein S was developed. WT protein S and protein S N74A, protein S KTK, protein S NNEDM variants and some additional protein S D95A were purified according to this alternative protocol.

For this, protein S was first partially purified from concentrated conditioned media by barium citrate precipitation. This favours precipitation of protein S that is γcarboxylated and allows removal of any non γ-carboxylated protein S.<sup>248</sup> Following barium citrate precipitation, protein S was then further purified by FPLC on a DEAE anion exchange column. This was optimised during the last year of my work by my colleague Dr Josefin Ahnström. A representative chromatogram is shown in Figure 3.3A where WT protein S was eluted with a calcium gradient of 0-20 mM CaCl<sub>2</sub>. Conditioned media, barium citrate precipitated protein S and the various fractions eluted from the DEAE anion exchange column were run on SDS-PAGE and analysed by silver staining or by Western blot (Figure 3.3B and C, respectively). The silver stained gel shows that barium citrate precipitation removes most of the contaminants present in the concentrated conditioned media (Figure 3.3B). The purity of the barium citrate precipitated protein S, however, varied. Protein S expressed in high amounts from stably expressed cells was relatively pure following barium citrate precipitation (as can be seen in Figure 3.3B), while barium citrate precipitated protein S from transiently transfected cells contained more contaminants (results not shown). The lower band contaminant seen in the silver stain in the barium citrate precipitated material is partially removed following passage on the DEAE anion exchange column and can be seen in the flowthrough fraction (Figure 3.3B). The higher band in the barium citrate precipitated sample and the main band in the elution fractions were identified as protein S on the Western blot (Figure 3.3C). Fractions from the elution peak that contained protein S and trace amounts of contaminants were pooled, concentrated and used in functional assays. An overall yield, as determined by ELISA, after barium citrate precipitation and purification on a DEAE anion exchange column, of 10-30% was obtained.



## Figure 3.3 Purification of barium citrate precipitated WT protein S on an anion exchange DEAE column and analysis by SDS-PAGE.

(A) The 280 nm absorbance (blue line) is plotted against the ml of liquid passing through the column. The green lines represent a change of buffer; equilibration buffer (20 mM Tris-HCl pH 7.5, 150 mM NaCl), elution buffer (20 mM Tris HCl pH 7.5, 150 mM NaCl, 0-20 mM CaCl<sub>2</sub>) and stripping buffer (20 mM Tris HCl pH 7.5, 1 M NaCl, 50 mM CaCl<sub>2</sub>). Peaks were collected and analysed. Total protein was detected by silver staining (B) and protein S was detected by a polyclonal antibody (C). Small fractions from the elution peak were collected separately and run on the gel. CM; conditioned media, bcp; barium citrate precipitated protein S, F; flowthrough, S; stripping.

## 3.2 Characterisation of recombinant protein S

## 3.2.1 Quantification of protein S by ELISA

The concentrations of purified protein S and protein S in conditioned media were determined by ELISA. A standard curve of 0-2 nM protein S was fitted to a sigmoidal dose-response curve and samples were diluted to ensure they fell in the linear part of the curve (generally between 0.25-1.5 nM protein S). The plasma purified protein S gave a higher absorbance reading than the corresponding concentration of protein S in normal plasma when detected with the monoclonal antibodies (Figure 3.4). This could potentially be due to slightly different amounts of protein S in the two standard preparations, to plasma protein S being monomeric and purified protein S multimeric (see section 3.2.3) or due to the fact that ~60% of plasma protein S is bound to C4BP and is less recognised by the monoclonal antibody.





When protein S in normal plasma (362 nM) was analysed by Western blot alongside plasma purified protein S and detected by a polyclonal antibody, bands of similar intensity were obtained suggesting that the reason behind the difference in absorbance observed between plasma purified protein S and protein S in media was not due to different concentrations of protein S. When my quantified recombinant protein S preparations (pure or in conditioned media) were analysed by Western blot, bands of equal intensity of purified protein S and protein S in conditioned media were only obtained if the concentration of pure protein S was determined with the plasma purified protein S standard and if protein S in conditioned media was determined with the normal plasma standard (results not shown). Since the monoclonal antibody MK55 recognises the EGF3-4 domains and the difference in detection of plasma protein S and purified plasma protein S by the monoclonal antibody AHPS 9052 is not 60% (difference between total and free protein S) it is unlikely that the differences in detection are due to the fraction of bound protein S in plasma. Multimerisation was analysed in section 3.2.3 and might account for the differences in detection between plasma protein S and purified protein S.

The concentration of purified protein S was, therefore, determined with the plasma purified protein S standard curve; that of protein S in conditioned media was determined with the normal plasma standard curve. Intra and inter ELISA variability varied, with purified protein S preparations giving higher CV values than protein S in conditioned media. Generally, the intra-assay CV was less than 10% for purified protein S and less than 5% for protein S in media. The mean inter-assay CV was 11% with samples being assessed by ELISA a minimum of 3 times.

### 3.2.2 Analysis of the purity of protein S

SDS-PAGE and silver staining of purified protein S preparations were performed to assess purity. WT protein S and protein S variants analysed under non-reducing conditions, irrespective of the purification method, migrated predominantly as a single band (Figure 3.5) of identical size to commercially available plasma purified protein S (see Figure 3.1B). A band of lower molecular weight than protein S was observed in some gels. This was, however, very faint in comparison with the protein S band and could only be visualised following long development of the silver stain. The concentration of purified protein S samples was assessed by absorbance at 280 nm, which corroborated the concentrations obtained by ELISA, suggesting that the protein S preparations were suitably pure for the *in vitro* experiments used in this project.



#### Figure 3.5 Silver staining of purified protein S samples.

WT and variant protein S samples (100 ng/well) were stained by silver staining. WT1; WT protein S purified by QFF anion exchange chromatography followed by immunoaffinity chromatography. WT2; WT protein S purified by barium citrate precipitation followed by DEAE anion exchange chromatography.

## 3.2.3 Multimerisation of protein S

Multimerisation of protein S has been well documented in the literature and it has been reported that multimeric protein S binds to phospholipids with higher affinity than monomeric protein S.<sup>189-192,251</sup>

To assess multimer formation of WT protein S, a Western blot was performed on a native PAGE gel. Protein S in dialysed concentrated conditioned media appeared to be monomeric (Figure 3.6A,B). Protein S partially purified by QFF anion exchange chromatography was mainly monomeric but also contained some dimers (Figure 3.6A). However, protein S partially purified by barium citrate precipitation (Figure 3.6B), fully purified protein S (Figure 3.6A) and plasma purified protein S (Figure 3.6B) all exhibited appreciable multimerisation. These results are consistent with previous reports.<sup>191,192</sup> It was difficult to obtain high quality Western blots showing protein S multimers. Nevertheless, no major difference in multimer formation between barium citrate precipitated protein S, fully purified protein S (independently of the purification protocol) and plasma purified protein S was observed.

To allow for potential differences in activity between monomeric and multimeric protein S, both protein S in concentrated conditioned media (monomeric) and purified protein S (multimeric) was used, where possible, in the functional assays.



Figure 3.6 Western blot of a native PAGE analysis to identify multimer formation occurring during the purification.

Protein S in dialysed concentrated conditioned media (lanes 1 and 4) is predominantely monomeric (A,B). Protein S partially purified by QFF anion exchange column (2) exhibited some dimers while its subsequent passage over the immunoaffinity column lead to multimerisation (3) (A). Barium citrate precipitated protein S (5) and plasma purified protein S (6) were both multimeric (B).

## 3.2.4 Cleavage of protein S

Cleavage of protein S has previously been reported to occur after one or more of the 3 arginine residues (Arg49, Arg60, Arg70) in the TSR domain. Once cleaved, the disulphide bond between Cys47 and Cys72 keeps the Gla domain and the EGF1 domain together. However, under reducing conditions the disulphide bond is broken and the protein S fragment spanning from the cleavage site in the TSR to the SHBG domain is observed below the band of full-length protein S. The effect of protein S cleavage upon protein S function remains to be fully defined.<sup>175-177</sup>

When partially purified protein S was analysed by Western blot under reducing conditions a doublet was observed. To investigate whether the two closely migrating bands were due to variable glycosylation or cleavage of protein S, partially purified protein S samples were incubated with or without PNGase. Western blot analysis (Figure 3.7) was performed both with a polyclonal anti protein S antibody (Panel A) and a monoclonal anti-Gla antibody (Panel B) under reducing conditions. Results in Panel A show that the double band was conserved after WT protein S and protein S Face2 are digested with PNGase,

suggesting that the difference in size is not due to differences in glycosylation. Protein S TSR was present only as one band in the blot revealed using a polyclonal antibody. Protein S TSR has all the arginines (after which protein S is cleaved) substituted to Gln. When protein S was detected with the anti-Gla antibody (Panel B) only one band was present. According to the MW, this band corresponds to the higher band of the duplets detected for WT and Face2 with the polyclonal antibody. The isolated Gla domain is not detected by the antibodies, probably because too small (~4.5 kDa) and present in too low amounts. This is consistent with previously reported results in the literature.<sup>252</sup> Together, these results suggest the doublet band observed under reducing conditions is due to cleavage of protein S in the TSR domain rather than variable glycosylation. Blots of protein S in conditioned media suggests that cleavage occurs during the course of protein S expression (results not shown).

To account for any differences in activity due to cleavage of protein S, regular checks on every preparation of interest were made to ensure cleavage of the protein S variants was no more than that of WT protein S.



#### Figure 3.7 Analysis of protein S cleavage.

Protein S incubated (+) and not incubated (-) with PNGase was analysed by SDS-PAGE under reducing conditions and detected either by a polyclonal anti-protein S antibody (A) or a monoclonal anti-Gla antibody (B).

## 3.3 Evaluation of APC cofactor activity of protein S

## 3.3.1 Preliminary experiments in protein S deficient plasma

Before analysing the APC and TFPI dependent cofactor activity of protein S in plasma, TF concentration was titrated in the CAT assay for optimal analysis of the activity of endogenous protein C and TFPI in the plasma. Protein S deficient plasma is more procoagulant than normal plasma and it has been reported that only low amounts of TF (~1.4 pM) allow visualisation of the TFPI cofactor activity of protein S.<sup>218</sup> To establish the concentration of TF to use, I performed a titration of TF from 0.0625 to 1 pM (Figure 3.8). This resulted in a prolongation of the lag time but thrombin peak height and ETP did not vary significantly. However, variability between the replicates (not showed) increased at lower TF concentration. TF concentration at 1 pM was chosen to trigger coagulation in all my plasma assays (unless otherwise stated) as duplicates were good and this is compatible with what is currently in use in the literature to trigger coagulation in plasma.<sup>240,241,253,254</sup> The stock concentration of recombinant TF in Innovin was considered to be 6000 pM, as estimated by P. Giesen in the Thrombogram guide of Synapse b.v. Hence, a 1:6000 dilution of the TF stock was performed to obtain 1 pM final concentration in the assay.





A titration of TF (0.0625-1 pM) was performed in protein S deficient plasma from Affinity Biologicals supplemented with 50  $\mu$ M phospholipids and 65  $\mu$ g CTI per ml plasma in a final volume of 100  $\mu$ l to assess the amount of thrombin generated.

To evaluate the activity of both endogenous protein C and TFPI in the protein S deficient plasma, polyclonal antibodies against APC (150 nM) and polyclonal antibodies against TFPI (100 nM) were used. When these were added to the plasma, no change in thrombin generation was observed (Figure 3.9). Addition of 120 nM plasma purified protein S to protein S deficient plasma had no effect on thrombin generation (Figure 3.9). Results suggest that APC and TFPI in protein S deficient plasma under these experimental conditions do not appreciably regulate thrombin generation, either in the presence or absence of protein S. The lack of APC influence is not surprising, considering that protein C does not circulate in its active form and that EPCR and TM (absent in these experiments) are important for physiological activation of protein C by thrombin. The lack of influence of TFPI was attributed to co-depletion of protein S and TFPI from the plasma during its preparation by the manufacturer (see section 3.4.1). Consequently, APC or TFPI were supplemented to the plasma preparations to evaluate cofactor activities of protein S. APC from Enzyme Research Laboratories was used unless otherwise stated.



**Figure 3.9 Activity of endogenous APC and TFPI in protein S deficient plasma.** Thrombin generation was performed in protein S deficient plasma in the presence and absence of 130 nM polyclonal antibodies against protein C (Sigma-Aldrich), 100 nM polyclonal antibodies against TFPI (Haematologic Technologies Inc) and in the presence of 120 nM protein S. Ab; antibody. PC, protein C.

## 3.3.2 Evaluation of APC cofactor activity of protein S by CAT

To assess the APC cofactor activity of protein S, thrombin generation was measured by CAT using protein S deficient plasma from Affinity Biologicals supplemented with recombinant WT protein S and its variants in conditioned media. Under the conditions of the assay, in the absence of added protein S, APC (0-10 nM) did not influence thrombin peak height, ETP, lag time or time to peak, in protein S deficient plasma (Figure 3.10, Panel A). When (0-120 nM) WT protein S was introduced into the assay in the presence of APC, dose-dependent changes were observed, primarily on reduction of the peak height and of ETP values (Figure 3.10, Panel B). The extent of inhibition of peak height and ETP was dependent both on the concentration of APC and protein S. For example, the higher the APC concentration, the less protein S was necessary to completely inhibit thrombin generation. Under these experimental conditions, 10 nM APC and 120 nM protein S reduced both the peak height and ETP by ~84%. To verify that those effects were a direct consequence of APC cofactor activity of protein S, polyclonal antibodies against either protein S (Figure 3.10, Panel C) or protein C (Figure 3.10, Panel D) were used. As seen in Figure 3.10 these completely inhibited the anticoagulant APC-protein S activity.



#### Figure 3.10 APC and protein S anticoagulant activity in plasma.

Thrombin generation was performed in protein S deficient plasma containing 100 nM inhibitory antibodies against TFPI. Up to 10 nM APC had no effect on thrombin generation in the absence of protein S; all concentrations generate lines that are superimposable (A). Following addition of 120 nM protein S (at 0-10 nM APC) an APC dose-dependent effect was observed (B). The upper single line represents 0-10 nM APC in the absence of protein S. Protein S in the presence of no or 2.5 nM APC generated lines that were superimposable. Conditions used are noted adjacent to the peak heights to which they refer. The anticoagulant activity of 10 nM APC and 120 nM protein S was inhibited by polyclonal antibodies against protein S (1.5  $\mu$ M, DAKO) ( $\Box$ ) (C) or against protein C (130 nM, Sigma-Aldrich) (x) (D). PS, protein S; PC, protein C. Representative experiments are shown (n=3).

To compare the anticoagulant activity of recombinant WT protein S in concentrated conditioned media with that of protein S purified by different techniques, the former was compared to protein S purified first by anion exchange chromatography and thereafter by immunoaffinity chromatography, or alternatively by barium citrate precipitation followed by anion exchange chromatography (see sections 2.3.1 and 2.3.2). All protein S preparations showed a similar inhibition of thrombin generation as can be seen in Figure 3.11. As protein S in concentrated conditioned media showed a similar activity to purified protein S, and furthermore also considering that protein S in media is essentially monomeric, conditioned media samples rather than fully purified protein S samples were preferably used to evaluate thrombin generation by CAT. In some experiments where purified protein S was used to assess APC cofactor activity, these preparations were purified by anion exchange chromatography followed by immunoaffinity chromatography.



Figure 3.11 APC cofactor activity of different preparations of protein S in plasma.

The anticoagulant activity of protein S in conditioned media was compared with that of protein S purified either by anion exchange chromatography followed by affinity purification (purified protein S 1, dotted lines) or protein S purified by barium citrate precipitation followed by anion exchange chromatography (purified protein S 2, dashed lines), 90 nM, in the presence of 4 nM APC (Xigris). APC from Xigris was used in this experiment and was more active than the commercial APC preparation mostly used in this thesis (Enzyme Research Laboratories).

Having optimised the CAT assay for APC-dependent protein S cofactor activity, I then aimed to screen my panel of protein S variants to identify those with a functional deficit. The screening of the protein S variants for their APC cofactor activity was performed at a high concentration of APC (16 nM) to achieve almost complete inhibition of thrombin generation in the presence of 100 nM WT protein

S (Figure 3.12). Figure 3.12 shows the % peak height obtained for each protein S variant in comparison with the peak height obtained in the absence of protein S. When % ETP values were plotted instead of % peak height a very similar graph was obtained (result not shown). In total, 28 variants were screened for their APC cofactor activity. As previously reported,<sup>224</sup> the cofactor activities of Face1 and of Face2 for APC were reduced. Importantly, these screening experiments identified a single point variant in EGF1 of protein S, D95A, which had a severely impaired APC cofactor activity that had not previously been implicated in protein S function. Under all conditions used (4, 9 or 16 nM APC, 60, 90 or 100 nM protein S), the reduction in APC cofactor activity of protein S D95A was greater than that observed for the previously reported variant with 7 amino acids exchanged (prothrombin residue swap), Face2.



Figure 3.12 Screening of protein S variants for APC cofactor activity.

The APC cofactor activity of protein S was evaluated at 16 nM APC and 100 nM protein S by CAT. The peak height in the absence of protein S was set to 100%. A high concentration of APC, leading to almost complete inhibition of thrombin generation with 100 nM WT protein S, was chosen specifically for screening purposes as this allows widening of the assay window at which variants with reduced APC cofactor activity are visualised.

As Saller *et al.*<sup>225</sup> did not investigate the APC cofactor activity of protein S Face1 and Face2 variants in the CAT assay, I titrated both variants alongside WT protein S (0-120 nM) in the presence of 16 nM APC (Figure 3.13). Protein S Face1 (Panel B) had almost completely abolished APC cofactor activity while the cofactor activity of Face2 (Panel C) was partially reduced (Figure 3.13). These results from the CAT assay, corroborate the results obtained by Saller *et al.*<sup>225</sup> using clotting assays. Saller *et al.*, however, showed a 9-fold reduction in protein S Face1 binding to phospholipids. They could therefore not conclude whether the residues comprising the variant were important or not for a cofactor interaction with APC. Protein S Face2, however, bound phospholipids and domain specific monoclonal antibodies with similar  $K_{d(app)}$  values to those of WT protein S.<sup>224</sup>



**Figure 3.13 APC cofactor activity in plasma of WT protein S, Face1 and Face2 variants.** Thrombin generation was measured in protein S deficient plasma supplemented with 16 nM APC, and 0-120 nM WT protein S (A), protein S Face1 (B) or protein S Face2 (C). Legends are positioned adjacent to the peak heights to which they refer.

To further evaluate the APC cofactor activity of protein S D95A, a titration (0-120 nM) of the variant (Figure 3.14, Panel B) was performed alongside WT protein S (Figure 3.14, Panel A), using 9 nM APC. Protein S D95A had a severely impaired APC cofactor activity, a finding replicated when a different concentration of APC

(4 nM) was used. The severely impaired APC cofactor function of protein S D95A was also confirmed in protein S deficient plasma supplied by Hyphen Biomed to ensure that these effects were not plasma specific (n=1, results not shown). While the cofactor activity of WT protein S is highly dependent upon the APC concentration used (see Figure 3.10, Panel B) that of protein S D95A is not, explaining the difference in fold activity between WT protein S and protein S D95A in Figure 3.12 and Figure 3.14.

Using 9 nM APC, (Figure 3.14, Panel A and B), 120 nM of WT protein S reduced peak height and ETP by ~80%, while 120 nM of protein S D95A only reduced peak height and ETP by ~10%.

Protein S is susceptible to proteolysis in the TSR domain and this may impair its APC cofactor function.<sup>175-177</sup> To ensure that any difference in activity between WT protein S and protein S D95A was not due to difference in TSR proteolytical cleavage, samples were analysed by Western blot under reducing conditions. No cleavage was observed either in WT protein S or in the D95A variant, showing that the different APC cofactor activity was not a result of different cleavage of the two preparations (results not shown).

Asp95 in protein S is  $\beta$ -hydroxylated and while it has not been shown to bind calcium, it is part of a partially conserved calcium binding motif. To ensure disruption of  $\beta$ -hydroxylation or loss of coordination of a calcium ion was not the reason behind the impaired APC cofactor activity of protein S D95A, Asp95 was also substituted with Asn, rather than Ala. Asn is structurally more similar to Asp than Ala and it can also be  $\beta$ -hydroxylated and potentially coordinate a calcium ion. When the protein S D95N variant was titrated, it was observed that it also had severely reduced APC cofactor activity (Figure 3.14, Panel C), with 120 nM of the D95N variant inhibiting peak height and ETP by only ~20%.

All results in Figure 3.14A-C were conducted with protein S and its variants in concentrated conditioned media. Concentrated conditioned media from cells not expressing protein S was used as a control and had no influence on thrombin generation (results not shown). The % change in peak is represented in Figure 3.14, Panel F. To confirm that the differences in APC cofactor activity I observed arose from protein S rather than the media, WT protein S and the protein S D95A variant were purified to homogeneity using anion exchange and immunoaffinity chromatography (see section 2.3.1). Figure 3.14D shows SDS-PAGE gel

visualised with silver staining which suggests that the protein S preparations were essentialy pure. Addition of 90 nM of purified WT protein S to protein S deficient plasma in the presence of 9 nM of APC, appreciably attenuated thrombin generation (Figure 3.14, Panel D), replicating the findings with WT protein S in conditioned media. When 90 nM purified protein S D95A was used, only a minimal effect on thrombin generation was seen (Figure 3.14, Panel D) corroborating the results from the experiments using concentrated conditioned media.

Inspection of a model of Gla-TSR-EGF1 domains of protein S<sup>193</sup> suggested that two residues with solvent exposed R groups, Asp78 and Gln79, are in close proximity to that of Asp95. These residues were not included in the first round of screening (shown in Figure 3.12). These were therefore mutated to alanine and expressed in HEK293T cells as described in sections 2.1.2 and 2.2.2. When protein S D78A and Q79A in concentrated conditioned media were analysed in the thrombin generation assay alongside WT protein S, a severely reduced APC cofactor activity was also observed for both (Figure 3.14, Panel E). The proximity of these residues to Asp95 suggests these three residues could form a possible cluster of residues involved in APC cofactor activity of protein S.



Figure 3.14 APC cofactor activity in plasma of WT protein S, protein S D95A, D95N, D78A and Q79A variants.

Thrombin generation was measured in protein S deficient plasma supplemented with 9 nM APC, 100 nM antibodies against TFPI and 0-120 nM WT protein S (A), protein S D95A (B), protein S D95N (C) or 90 nM purified WT (dashed line) or purified protein S D95A (dotted line) (D). Protein S concentrations are positioned adjacent to the peak heights they refer to. The cofactor activity of 60 nM WT protein S and protein S variants D95A ( $\mathbf{\nabla}$ ), D78A ( $\Diamond$ ) and Q79A ( $\circ$ ) was compared at 9 nM APC (E). Typical experiments are shown (n=3). Dose response data from titrations with WT protein S D95A, and protein S D95N in the presence of 9 nM APC are shown in Panel F (data is expressed as mean ± SD of two independent experiments performed in duplicate). Inset in Panel B shows recognition of WT protein S and protein S D95A in media by polyclonal antibodies (pAb) and a monoclonal antibody recognising only  $\gamma$ -carboxylated Gla domains ( $\alpha$ -Gla). Inset in Panel D shows the See Blue prestained marker, plasma purified protein S from Enzyme Research Laboratories (ERL, lane 1), purified recombinant WT protein S (lane 2) and purified protein S D95A (lane 3) visualised with silver staining.

### 3.3.3 Binding of protein S to phospholipid surfaces

As phospholipid binding is a prerequisite for protein S function, the binding of WT protein S and of protein S D95A to phospholipids was evaluated. Partially purified (using barium citrate precipitation) protein S, 0-120 nM, was incubated for 2 hours on a plate coated with 25  $\mu$ g/ml phospholipids and bound protein S was detected by an HRP-conjugated antibody against protein S. As shown in Figure 3.15, both WT protein S and protein S D95A were able to bind phospholipids, with K<sub>d(app)</sub> of 5.7 ± 1.2 and 9.5 ± 2.3 nM (n=3), respectively (Table 3.3).

 $K_{d(app)}$  values were analysed by the Mann Whitney test and were found not to be statistically different (p>0.05). Binding was observed in the presence of Ca<sup>2+</sup>, but not in the presence of EDTA or in the absence of phospholipids, as expected, confirming the specificity of the assay. These findings are broadly consistent with previously reported  $K_{d(app)}$  values for protein S binding to phospholipid surfaces.<sup>182,190,223-225</sup> My results suggest the loss of APC cofactor activity observed for protein S D95A is not due to reduction or loss of binding to phospholipid surfaces.





Protein S (0-120 nM) was incubated in a plate coated with 25  $\mu$ g/ml phospholipids. Bound protein S was detected with an HRP-conjugated polyclonal antibody against protein S. A representative experiment is shown. The K<sub>d(app)</sub> values, 5.69 ± 1.24 and 9.54 ± 2.26 nM for WT protein S and protein S D95A, respectively, were obtained by calculating the mean ± SD of three independent experiments performed in duplicate for each concentration of protein S.

## 3.3.4 Binding of protein S to domain specific monoclonal antibodies

Substituting residues in a domain can sometimes affect the overall folding of the protein, the structure of the domain in which the substitution has been introduced or the structure of adjacent domains. To assess the integrity of the domain structure and folding of protein S variants, binding of protein S to conformational domain specific antibodies was evaluated. Protein S in concentrated conditioned media was incubated in a plate coated with monoclonal antibodies recognising either the Gla domain (MK21), the EGF1 domain (MK54) or the C-terminal SHBG domain (MK61). Bound protein S was detected with polyclonal antibodies as described in the Methodology. Binding curves were fitted with a one site binding equation and the  $K_{d(app)}$  values were obtained.

Table 3.3 represents the mean  $\pm$  SD K<sub>d(app)</sub> values of three independent experiments performed in duplicate. K<sub>d(app)</sub> values were analysed by the Mann Whitney test and were found not to be statistically different (p>0.05). The results suggest that substitution of Asp95 does not result in a significant change in the domain structure of protein S.

	DOPS/DOPC/DOPE vesicles	MK21 (Gla)	MK54 (EGF1)	MK61 (SHBG)
WT protein S	5.69 ± 1.24	2.18 ± 0.97	0.81 ± 0.11	5.31 ± 1.11
Asp95 protein S variant	9.54 ± 2.26	2.53 ± 1.06	$0.83 \pm 0.03$	5.39 ± 0.54

## Table 3.3 Binding of protein S to phospholipids and domain specific monoclonal antibodies.

 $K_{d(app)}$  values (nM) of WT protein S and the Asp95 protein S variant for phospholipid vesicles and domain specific monoclonal antibodies are expressed as mean ± SD of three independent experiments performed in duplicate. Binding of WT protein S and the Asp95 variant to phospholipids and domain specific monoclonal antibodies was analysed by Mann Whitney test and the differences were found not to be statistically significant (p>0.05).

#### 3.3.5 Protein S enhancement of APC mediated cleavage of FVa at Arg306

Protein S cofactor function towards APC was initially assessed in the CAT assay (described above), which assesses thrombin generation in plasma. To more precisely measure and quantify the influence of protein S variants on APC cofactor function, a purified system was used that examines cleavage of FVa. The FVa variant, R506Q/R679Q, was used to evaluate protein S enhancement of APC mediated cleavage at Arg306 in FVa. FVa inactivation was performed in the presence or absence of 0.5 nM APC and a titration of purified protein S (0-100 nM) was performed as described in the Methodology (section 2.6.3). After 10 minutes the remaining FVa activity was measured using a prothrombinase assay. Using this approach, WT protein S efficiently enhanced APC mediated cleavage at Arg306 of FVa, in contrast to protein S D95A and protein S Face2 which showed almost no enhancement of APC activity (Figure 3.16, Panel A).

To quantify the rate of cleavage at FVa Arg306 in the presence and absence of protein S, time course experiments were performed. In the absence of protein S, or in the presence of protein S D95A or protein S Face2, 3 nM APC was used and aliquots were quenched at different time points (0-20 min). In the presence of WT protein S the APC concentration was lowered to 0.5 nM, as APC is efficiently enhanced by WT protein S. The remaining FVa activity at all time points was measured in the prothrombinase assay (Figure 3.16, Panel B). It is observed that approximately 6-fold more APC is needed in the presence of protein S D95A and protein S Face2 to obtain a similar amount of APC-mediated FVa R506Q/R679Q inactivation as with WT protein S. Using the FVa inactivation curves obtained, the apparent pseudo-first order rate constants were calculated (see section 2.6.3) and corrected for the APC concentrations used. Rate constants for APC, APC+WT protein S, APC+protein S D95A and APC+protein S Face2 were respectively 6.1±1.2•10<sup>5</sup>, 8.3±1.6•10<sup>6</sup>, 1.1±0.2•10<sup>6</sup> and 2.0±0.3•10<sup>6</sup> and were broadly consistent with those previously reported in the same laboratory.<sup>141</sup> Under these experimental conditions, APC mediated cleavage at FVa Arg306 was enhanced by WT protein S by 13.9 ± 3.6-fold, while protein S D95A was only able to enhance APC by 1.8  $\pm$  0.4-fold and protein S Face2 by 3.3  $\pm$  0.3-fold. Accordingly, protein S D95A had 12.9% and protein S Face2 30.3% of the activity

of WT protein S. Protein S had no effect on FVa activity in the absence of APC (not shown).



Figure 3.16 Protein S enhancement of APC mediated cleavage of FVa at Arg306.

Protein S (0-120 nM) in the presence of 0.5 nM APC was incubated with 0.8 nM FVa R306Q/R679Q in the presence of phospholipids for 10 minutes. The remaining FVa activity was measured with a prothrombinase assay. Results are plotted as mean  $\pm$  SD from three independent experiments performed in duplicate (A). A time course experiment was performed to calculate the apparent pseudo-first-order rate constants of WT protein S, protein S D95A and protein S Face2 (B).

### 3.3.6 Discussion

Initially, 28 protein S variants with substitutions in the Gla, TSR, EGF1 and EGF2 domains were constructed and expressed. I evaluated their ability to enhance APC mediated anticoagulant activity in plasma. The anticoagulant function of these 28 variants, and the subsequently expressed variants protein S D78A, D79A and D95N, was assessed by a thrombin generation assay, the specificity of which was demonstrated by utilising polyclonal antibodies both against protein S and protein C that completely inhibited the anticoagulant response observed when adding protein S in the presence of APC. The advantage of the thrombin generation assay conducted in plasma is that it shows very clearly how APC is heavily dependent upon protein S.<sup>218</sup> In the absence of protein S, 10 nM APC has no effect on thrombin generation. In contrast, in the presence of protein S, maximal anticoagulant activities with near ablation of thrombin generation are obtained with APC concentrations of around 10 nM.

My initial screening results identified three protein S variants with appreciable reduction in APC cofactor function. These were the already reported protein S Face1 and Face2 variants (a 4 and a 7 residue composite variant in the Gla domain, respectively) and the novel protein S D95A point variant in the EGF1 domain. The Face2 variant, in the report by Saller *et al.*, retained binding to phospholipids and yet had appreciably reduced APC cofactor activity, while Face1 was shown to have an approximate 9-fold reduced phospholipid binding.<sup>224</sup> It is suggested that the 7 residues substituted in the protein S Face2 variant collectively present a face of the Gla domain to APC and form a potential contact region. This variant was selected for investigation here, because it is a well-characterised protein S variant with substantially reduced APC cofactor activity and could therefore act as a representative control for a dysfunctional protein S. Saller *et al.*, however, did not assess protein S Face2 and Face1 APC cofactor activity in the CAT assay. I confirmed a reduction in APC cofactor activity in plasma, assessed by CAT.

The natural protein S variants T103N<sup>255</sup> and K155E<sup>256,257</sup> have been reported to have reduced APC cofactor activity. I could not observe any significant reduction in APC cofactor activity of protein S T103A in the CAT assay. It is, however, important to point out that defective APC cofactor activity of protein S T103N was

not observed in the APC mediated inactivation of WT FVa.<sup>255</sup> In addition, at higher concentrations of APC (8 nM) no difference was observed between WT protein S and protein S T103N in the FVIIIa degradation assay.<sup>255</sup> Protein S Tokushima (K155E) has been described in the literature to have a cofactor activity towards APC of 31% <sup>256</sup> and 58% <sup>257</sup> of that of WT protein S. I did observe a decrease in protein S K155A cofactor activity towards APC, although this was quite moderate under conditions used in my screening assay. However, as this reduction was appreciably less than that for protein S D95A, it was not further analysis during my PhD project.

Considering pooled results from multiple assays (n=3), protein S S99A and N154A seem to be the variants that have most enhanced APC cofactor activity. It would also be interesting to further analyse these variants. A full titration in the CAT assay would allow an assessment of the extent of increased activity. It would be interesting to elucidate their importance in APC cofactor activity by assessing their activity in a purified assay and their ability to bind phospholipids.

The substitution of Asn136 in protein S abolishes the calcium binding site in the EGF2 domain of protein S.<sup>258</sup> I did not observe a reduction in APC cofactor activity. However, it was secreted in significantly lower levels than WT protein S and all other protein S variants from stably transfected cells (see section 3.2.1). As APC cofactor activity appeared normal I did not further investigate the expression of protein S N136A, but it is likely that mutating Asn136 to Ala influences the structure of protein S leading to intracellular retention of incorrectly folded protein S.

Protein S variants R49Q and Q52R in the TSR domain of protein S and protein S variants K97Q and P106S in the EGF1 domain were assessed in Dahlbäck's laboratory (He *et al.*),<sup>244</sup> and found to be important for species specificity towards APC. Following substitution of the human protein S residues into the corresponding residues of bovine protein S, an increased cofactor activity towards human APC was observed. They were also shown to specifically increase the activity of human protein S towards bovine APC. During my project I did not mutate Gln52 and as the substitution R49Q was part of the composite variant TSR it is not possible to directly compare my results with those previously reported. My protein S P106A variant had slightly increased activity in comparison with WT protein S, in accordance with previous results.<sup>244</sup> Protein S K97A,

however, had decreased activity in comparison with WT protein S. This is in discrepancy with the results of the K97Q variant of He *et al.* but is likely to depend on the different substitution, to alanine or glutamine. It is interesting that residues previously found to be important for species specificity (residues Arg49, Gln52, Lys97, Pro106) are spatially located in close proximity to residues Asp95, Asp78 and Gln79 here shown to be important for APC cofactor activity (Figure 3.17).



Figure 3.17 Location of residues important for species specificity in relation to Asp95, Asp78 and GIn79.

Residues Arg49, GIn52 in the TSR domain of protein S and residues Lys97 and Pro106 in the EGF1 domain of protein S have been reported to be important for species specificity<sup>244</sup> and are here highlighted in cyan. Residues Asp95, Asp78 and GIn79 important for APC cofactor activity are highlighted in green. This N-terminal model of the GIa-TSR-EGF1 domains of protein S is adapted from the model of Villoutreix *et al.*.<sup>193</sup>

Analysis of protein S function can be complicated due to its propensity to multimerise upon purification.<sup>191,192</sup> Accordingly, I have performed APC cofactor assays with unpurified protein S in concentrated conditioned media as well as with fully purified preparations. My results were consistent between protein S preparations. Both forms of the protein S D95A variant, purified or in concentrated conditioned media, had an appreciable reduction in APC cofactor activity when compared to the respective WT protein S preparation. In all comparison experiments carried out, the protein S D95A variant had less APC cofactor than the protein S Face2 variant. This suggests that Asp95 in protein S may occupy a pivotal position with respect to its interaction with APC. The importance of Asp95 in APC cofactor activity was further confirmed by the severely reduced activity

observed when mutating to alanine the two residues in close proximity to Asp95, Asp78 and Gln79. Asp78, Gln79 and Asp95 are conserved across species with the exception of Asp78 that is replaced by the structurally similar Asn78 in birds (Figure 3.18). Using a working model of the Gla-TSR-EGF1 domains of protein S, I demonstrate the likely proximal spatial location of these three residues, see Figure 3.19, and their relationship to protein S Face2 residues.

Homo sapiens Chimpanzee (Pan troglodytes)	IPDQCSPLPCNEDGYMSCKDGKASFTCTCKPGWQGEKCE
Rhesus Macaque (Macaca mulatta)	**************************************
Cow (Bos Taurus)	*S***N*******F*T***Q*T***I**S*******
Wild boar (Sus scrofa)	****N******F*T***Q*M***I**S**E****
Rabbit(Oryctolagus cuniculus)	*****N****SE***LN****Q*T***I**********
Dog (Canis lupus familiaris)	**************************************
Rat (Rattus norvegicus)	****D*M*****L****QGA***I*****D**Q
Mouse(Mus musculus)	*S***D*I******LA*Q**Q*A***F*****DR*Q
Xenopus (Xenopus tropicali)	L****T****RK**KE*I***GN***I***G***PL*D
Chicken (Gallus gallus)	*SN*******HR***KE*I**Q*KY**I*******N**
Bird (Taeniopygia guttata)	*SN******HK***KD*I**Q*KY**V**A**R**N**
<pre>Platypus(Ornithorhynchus anatinus)</pre>	*****F****H****LF****QGRY**V*R***H**R**

#### Figure 3.18 Conservation of Asp78, GIn79 and Asp95 across species.

The sequence of human protein S between amino acid residue 76 and 114 is shown. Asp95 (in red), and Gln79 (in purple) are conserved throughout species. Asp78 (in blue) is conserved across species with the exception of birds where Asp78 is replaced by the structurally similar Asn78.



## Figure 3.19 Location of Asp78, GIn79, Asp95 and Face2 within the protein S GIa-TSR-EGF1 model.

Domains are labelled on the right hand side cartoon model. Residues mutated in the protein S Face2 variant, Asp78, Gln79 and Asp95 are in light gray in the left hand side surface model. Residues Asp78, Gln79 and Asp95 are highlighted by the box to show their proximal spatial location. The model is that of Villoutreix *et al.*.<sup>193</sup>

I have investigated the mechanism of reduction of activity of protein S Asp95 variants. Asn136, in the EGF2 domain, Asn178 in the EGF3 domain and Asn217 in the EGF4 domain are all β-hydroxylated. These and other residues that are part of the calcium binding sequence motif in the respective domains have all been shown to coordinate a calcium ion with high affinity (Figure 3.20).<sup>258</sup> The EGF1 domain of protein S has not been shown to contain a calcium binding site.<sup>258</sup> The calcium binding motif in EGF1 is only partially conserved and according to the protein S Gla-TSR-EGF1 model of Villotreux *et al.*<sup>193</sup> residues of the calcium binding sequence motif before the first Cys residue and between the third and fourth Cys residues are localised on opposite sides of the EGF1 domain. However, to be sure that substitution of Asp95 to Ala did not result in disruption of a potential calcium binding site, I substituted Asp95 to Asn rather than Ala. Asn is structurally similar to Asp, can be β-hydroxylated and is able to coordinate a calcium ion (as has been observed in EGF2, 3 and 4).



#### Figure 3.20 EGF domains of protein S and binding to calcium.

 $\beta$ -hydroxylated residues in the EGF domains of protein S are highlighted in yellow. The K<sub>d</sub> values of each domain for calcium is shown above each domain. K<sub>d</sub> values were taken and the EGF diagram modified from Stenflo *et al.*<sup>258</sup> The calcium binding sequence motif is shown in the box, in the red squares in the sequence alignment of the EGF domains of protein S and the specific residues are circled in red in the EGF diagram.

However, as the protein S D95N variant also had reduced activity, it is unlikely that disruption of a calcium binding site is responsible for this effect as this variant would be predicted to bind calcium normally. Furthermore, while Asp95 is known to be  $\beta$ -hydroxylated it has previously been shown that  $\beta$ -hydroxylation itself is not a requirement for anticoagulant activity of protein S.<sup>179</sup> My work favours a direct effect of Asp95 in APC cofactor activity. However, I can not discount the possibility that the dramatic effect on protein S activity of the Asp95 residue substitution could be through a conformational repositioning of other functional domains. To assess this I performed binding to phospholipids, as this property underpins all protein S function. Plate binding assays, however, indicated no major functional defect on phospholipid binding. Furthermore, domain specific monoclonal antibody binding to WT and variant protein S (against the Gla domain, the EGF1 domain and the C-terminal SHBG domain) was also normal, suggesting that substitution of Asp95 does not reduced APC cofactor function by disrupting adjacent domain structure.

A current view of protein S enhanced APC cofactor activity suggests a functional repositioning of the APC cleavage site away from FVa Arg506 towards FVa Arg306.<sup>228</sup> I therefore performed a FVa inactivation assay using a FVa variant, FVa R506Q/R679Q, that cannot be cleaved at position 506 and 679. This allowed me to specifically analyse cleavage at Arg306 by APC. This is the APC cleavage site in FVa that is mainly enhanced by protein S. Using both concentration dependent and time course assays, I was able to confirm reduced cleavage of this variant by APC in the presence of protein S D95A or protein S Face2. Importantly, the single point variant protein S D95A had more impaired cofactor activity towards APC mediated cleavage of FVa R506Q/R679Q than the previously described 7 amino acid prothrombin swap variant Face2.

The available results therefore suggest that Asp95 of protein S may play an important and a direct role in APC recognition, resulting in enhanced APC function. This is compatible with the study performed by Hackeng *et al.*,<sup>246</sup> who used the isolated EGF1 domain of protein S to functionally disrupt the protein S and APC interaction. They showed a direct binding of EGF1 to APC suggesting EGF1 as a potentially important APC contact site on protein S. My results suggest that Asp95 constitutes a critical residue within EGF1 mediating the APC

cofactor function. Together with Asp78 and Gln79, Asp95 could form an important functional interaction site for APC.

The importance of residues in protein S Face2 should be further investigated. Phospholipid binding and binding to monoclonal antibodies were not evaluated in my thesis but should also be reassessed. Saller *et al.* found no significant difference in binding to monoclonal antibodies, however, binding of protein S Face2 to the monoclonal antibody MK54 against the EGF1 domain had a slightly higher  $K_{d(app)}$  value (5.0 ± 1.7 nM) in comparison with that of WT protein S (1.6 ± 0.3 nM).<sup>224</sup> It is possible that protein S Face2 has reduced APC cofactor activity primarily because so many residues have been substituted: this may cause a conformational change in regions important for direct interaction with APC. In all my experiments the APC cofactor activity of the single point variant protein S Asp95 was more pronounced than that of the 7 residue prothrombin swap in protein S Face2 binding to monoclonal antibodies should be repeated and point variants of the 7 amino acids substituted in protein S Face2 should be made and evaluated in functional assays.

Face1 had previously been shown not to bind phospholipids.<sup>224</sup> Interestingly, when protein S Face1 was evaluated for its binding to phospholipids surfaces, I found that it was able to bind phospholipids with a similar affinity to WT protein S (n=1). Analysis of protein S Face1 to phospholipids (n=3) has been subsequently further evaluated by my colleague Dr Josefin Ahnstöm, who found no statistical difference between WT protein S and protein S Face1 in binding to phospholipids. To identify which residue(s) in the composite variant protein S Face1 is important for APC cofactor activity she made the individual point variants contained in protein S Face1; protein S N33S, protein S P35T, protein S E36A and protein S Y39V. Analysis by CAT showed that variants protein S P35T and E36A were those with most reduced APC cofactor activity (similar to the reduced activity of protein S Face1) (as these are results of Dr Josefin Ahnstöm, I have not shown them). As proline is known to be important for protein folding and stability, only protein S E36A was further investigated (Dr Josefin Ahnstöm). Protein S E36A bound phospholipids with a  $K_{d(app)}$  similar to that of WT protein S. The binding of protein S Face1 and protein S E36A to domain specific monoclonal antibodies against the Gla domain (MK21 and MK47) and the TSR domain (MK67) was also

104

evaluated by my colleague Dr Josefin Ahnström. No difference between protein S Face1, protein S E36A and WT protein S was found. When she analysed the cofactor activity of protein S E36A in the FVa inactivation assay she found that it had only 11.3% of the cofactor activity of WT protein S (n=4). This reduction in protein S cofactor activity towards APC mediated cleavage in Arg306 of FVa was similar to that of protein S D95A (12.9%) and both were more pronounced than that of protein S Face2 (30.3%).

Residues Asp95, Asp78, Gln79 (red), Gla36 (green) and residues in Face2 (blue) are highlighted in the working model of the Gla-TSR-EGF1 domains of protein S showing their likely spatial location (Figure 3.21).



#### Figure 3.21 Location of residues found to have reduced APC cofactor activity.

The novel residues found to have reduced APC cofactor activity in my project Asp95, Asp78 and Gln79 in EGF1 are highlighted in red. Residues in the previously reported protein S Face2 variant are shown in blue. The novel residue identified by my colleague Dr Josefin Ahnström, Gla36, is highlighted in green. The model is taken from Villoutreix *et al.*.<sup>193</sup>

These results suggests that Gla36 of protein S seems to be equally important as is Asp95 for APC cofactor function, suggesting that possibly both the Gla and EGF1 domains of protein S are important for APC cofactor activity.

Collectively, our results suggest that Asp95 and Glu36 constitute critical residues within the EGF1 and Gla domain of protein S mediating the APC cofactor function.

## 3.4 Evaluation of TFPI cofactor activity of protein S

## 3.4.1 Evaluation of TFPI cofactor activity of protein S by CAT

From preliminary results described in section 3.3.1, I decided to use 1 pM TF to trigger coagulation in all my plasma assays and to supplement TFPI in the protein S deficient plasma. Hence, TFPI cofactor activity of protein S was assayed by CAT in protein S deficient plasma supplemented with recombinant TFPI and protein S. When 0-5 nM TFPI and 0-120 nM protein S were supplemented in protein S deficient plasma from Affinity Biologicals only a very small prolongation of lag time and reduction in peak height were observed. I therefore evaluated the protein S enhancement of TFPI in protein S deficient plasma supplied by two different companies, Affinity Biologicals and Hyphen Biomed. I observed that the effect on thrombin generation of TFPI (3 nM) with or without protein S (120 nM) varied depending on the source of plasma (Figure 3.22), with a more pronounced protein S/TFPI anticoagulant activity being observed in plasma supplied from Hyphen Biomed (Panel B) in comparison with that from Affinity Biologicals (Panel A). As can be seen in Figure 3.22, while the total protein S/TFPI anticoagulant activity in plasma from Affinity Biologicals prolonged the lag time by ~210% and reduced the peak height by ~20%, in plasma from Hyphen Biomed the lag time was prolonged by ~340% and the peak height reduced by ~65% (Figure 3.22). To investigate the difference between the two different sources of plasma my colleague, Verity Hockey, determined the amount of total TFPI in an in-house ELISA (n=1). This suggested that protein S deficient plasma from Affinity Biologicals contained approximately 25%, and Hyphen Biomed approximately 50%, of the amount of total TFPI present in normal plasma. During the course of this thesis it was shown by Castoldi et al. that TFPI can be co-depleted when protein S is immunodepleted from normal plasma in the production of protein S deficient plasma.<sup>241</sup> They also analysed the amount of TFPI in protein S deficient plasma from Affinity Biological and found it contained 25.5% of normal TFPI antigens levels which is in accordance with our findings. In subsequent experiments, protein S deficient plasma supplied by Hyphen Biomed was used, as it allowed better visualisation of the protein S/TFPI anticoagulant activity. However, to ensure that the effect observed in protein S deficient plasma from

Hyphen Biomed was not unique to this plasma, protein S deficient plasma supplied from Enzyme Research Laboratories was also tested. Both plasma from Hyphen Biomed and Enzyme Research Laboratories showed a similar prolongation of the lag time and reduction of the peak height following addition of TFPI in the presence and absence of protein S (results not shown).

In my initial experiments, I used TFPI in concentrated conditioned media that I had expressed from a stable cell line in our lab. This was quantified by a preliminary ELISA by my colleague Verity Hockey. Later in my project, purified TFPI (kind gift of the Chemoserotheraputic Institute Kaketsuken) was used. Unless specifically stated, protein S in concentrated conditioned media was used in all experiments.



Figure 3.22 TFPI anticoagulant activity in the presence or absence of protein S in plasma. Purified TFPI (3 nM) was incubated in the presence or absence of protein S (120 nM) in protein S deficient plasma from Affinity Biologicals (A) and Hyphen Biomed (B), in the presence of 50  $\mu$ M phospholipids, 65  $\mu$ g CTI per ml plasma and 1 pM TF.

The specificity of the anticoagulant activity seen when supplementing TFPI in the plasma in the presence or absence of protein S was confirmed by performing CAT assays in presence or absence of inhibitory polyclonal antibodies. Polyclonal antibodies against TFPI (100 nM, Figure 3.23, Panel A) or polyclonal antibodies against protein S (1.5  $\mu$ M, Figure 3.23, Panel B) were used. These completely inhibited the effect seen when supplementing TFPI and protein S in concentrated conditioned media to the plasma. Concentrated conditioned mock media not containing protein S and TFPI had no effect on thrombin generation (results not shown).



Figure 3.23 Influence on thrombin generation of polyclonal antibodies against TFPI and against protein S.

TFPI (2.5 nM) in concentrated conditioned media was incubated in the presence or absence of protein S (60 nM) in protein S deficient plasma from Hyphen Biomed. The activity observed by supplementing the plasma with TFPI and protein S was reversed by use of 100 nM polyclonal antibodies against TFPI (A) or 1.5  $\mu$ M polyclonal antibodies against protein S (B).
While protein C does not circulate in its activated form, there remained a possibility that endogenous protein C might be activated in low levels by thrombin generated during the assay. To eliminate this possibility, TFPI and protein S were supplemented in the protein S deficient plasma in the presence or absence of 130 nM polyclonal antibodies against protein C (Figure 3.24). The polyclonal antibodies had previously been shown to be inhibitory against APC (see Figure 3.10). No difference in thrombin generation was observed following addition of inhibitory antibodies against protein C. This demonstrated that the thrombin generated during these assay conditions (i.e. in the absence of TM and EPCR) was unable to significantly activate protein C. Any anticoagulant activity of protein S was therefore independent of its APC cofactor function.



**Figure 3.24 TFPI and protein S anticoagulant activity in plasma in the presence and absence of polyclonal antibodies against protein C.** The effect of TFPI in concentrated conditioned media (2.5 nM) in the absence (Panel A) and presence (Panel B) of protein S (60 nM) was assessed in the presence and absence of 130 nM polyclonal antibodies against protein C (PC). A representative experiment is shown. The results of addition of antibodies against protein C in the presence of TFPI and protein S was confirmed in additional experiments (n=3).

Once the assay conditions had been set up to guarantee the visualisation and specificity of the protein S enhancement of TFPI, I wanted to ensure that my different WT protein S preparations all enhanced TFPI to a similar extent. Thus, the activity of WT protein S in concentrated conditioned media in the presence of TFPI was compared with that of WT protein S purified by either purification protocol (see Methodology, section 2.3). Results (n=3) were analysed by a paired Wilcoxon test and no statistical difference in terms of thrombin peak height or lag

time was observed between the different preparations. Consequently, and because of its ready availability, monomeric protein S in concentrated conditioned media was preferentially used.

The TFPI cofactor activity of WT protein S in concentrated conditioned media was evaluated by titrating protein S (0-120 nM) in the presence of 3 nM TFPI. As can be seen in Panel A of Figure 3.25, WT protein S enhanced TFPI in a dosedependent manner. To evaluate whether residues in protein S variants with reduced APC cofactor activity (see section 3.3.2) were also important for TFPI cofactor activity, protein S Face2 (Panel B), protein S D95A (Panel C), protein S D95N (Panel D), protein S D78A (Panel E) and protein S Q79A (Panel F) were titrated alongside WT protein S to evaluate their TFPI cofactor activity (Figure 3.25). To more easily visualise the difference in peak height and lag time between the variants and WT protein S, the results were plotted as % change in lag time and peak height in comparison with the control (Figure 3.26). Interestingly, Figure 3.25 and Figure 3.26 show that all variants were able to enhance TFPI anticoagulant function. However, protein S variants D95A, D78A and Q79A were somewhat less efficient than WT protein S. They all had a reduced lag time (by ~10-20%) and increased peak height (by ~5-20%) in comparison with WT protein S. Protein S D95N had a similar lag time to WT protein S, but an increased peak height (by ~10-25%). While protein S Face2 had a reduced effect on the lag time (by ~10-20%) compared to WT protein S, its effect on the peak height was enhanced (by up to 60%). The titration of WT protein S and protein S D95A was performed twice but the percentage effect on peak height and lag time could not be expressed as mean  $\pm$  SD due to the different assay conditions (3 nM vs 2.5 nM TFPI). The TFPI cofactor activities of protein S variants D95A, D95N, D78A and Q79A were in addition evaluated using single protein S concentrations (n=3), which confirmed the results obtained from the titration. When the cofactor activity of protein S Face2 was reassessed, however, it was generally similar to that of WT protein S in terms of lag time, while the effect on peak height was more pronounced, suggesting that this variant has enhanced TFPI cofactor activity. To further assess the TFPI cofactor activity of protein S Face2 and protein S D95A these variants were purified and assessed in the FXa inhibition assay (see section 3.4.2.2).



Figure 3.25 Thrombin generation in the presence of TFPI and WT protein S, protein S Face2, D95A, D95N, D78A or Q79A.

Thrombin generation was assessed at 3 nM TFPI in concentrated conditioned media in the presence of 0-120 nM WT protein S (A), protein S Face2 (B), protein S D95A (C), protein S D95N (D), protein S D78A (E) or protein S Q79A (F). Protein S D95N, protein S D78A and protein S Q79A could not be evaluated at higher concentrations than those described here due to their stock concentrations.



Figure 3.26 Change (%) in peak height and lag time of WT protein S, protein S Face2, D95A, D95N, D78A and Q79A in comparison with the control.

The effects of TFPI and protein S on peak height and lag time from Figure 3.26 were expressed as a % change in comparison with the control. The % change of WT protein S, protein S D95A, protein S D95N and protein S Face2 on lag time and peak height is shown in Panel A and B respectively. The % change of WT protein S, protein S D78A and protein S Q79N on lag time and peak height is shown in Panel C and D, respectively.

Additional experiments were performed with the EGF1 and EGF2 protein S point variants produced by my collaborator Dr S.M Rezende (Universidade Federal de Minas Gerais, Brazil). These were screened at 90 nM using 2.5 nM TFPI (Figure 3.27 and Figure 3.28). The TFPI cofactor activity of protein S T108A, protein S N136A, protein S S153A and protein S K155A could not be evaluated due to lack of material. As can be seen, observing the lag times and peak heights in Figure 3.27 and Figure 3.28, most protein S variants had a similar cofactor activity towards TFPI to that of WT protein S. Several protein S variants showed a slightly enhanced prolongation of the lag time and reduction in the peak height, however, none of the 19 variants tested, exhibited greatly reduced TFPI cofactor activity.



Figure 3.27 Thrombin generation in the presence of TFPI and WT protein S and protein S EGF1 and EGF2 point variants.

Thrombin generation was assessed in protein S deficient plasma supplemented with 2.5 nM TFPI in concentrated conditioned media. The cofactor activity of protein S variants, 90 nM, towards TFPI was assessed.



Figure 3.28 Thrombin generation in the presence of TFPI and WT protein S and protein S EGF1 and EGF2 point variants.

Thrombin generation was assessed in protein S deficient plasma supplemented with 2.5 nM TFPI in concentrated conditioned media. The cofactor activity of protein S variants, 90 nM, towards TFPI was assessed.

As stated in section 3.4.1, two different sources of TFPI were available, TFPI in concentrated conditioned media and purified TFPI. When 90 nM WT protein S was evaluated for its TFPI cofactor activity towards 2.5 nM purified TFPI (Figure 3.29, Panel A), a more pronounced anticoagulant activity was observed both in terms of prolongation of the lag time and reduction of the peak heights to that previously observed when using TFPI in concentrated conditioned media. When the anticoagulant activity of TFPI in concentrated conditioned media was compared with that of purified TFPI, a difference was observed both in the absence and presence of protein S (Figure 3.29, Panel B).



Figure 3.29 Thrombin generation in the presence of purified TFPI and TFPI in concentrated conditioned media in the presence and absence of protein S.

Thrombin generation in protein S deficient plasma was analysed in the presence of 2.5 nM purified TFPI in the presence or absence of 60 nM protein S (A). The anticoagulant activity of purified TFPI in the presence or absence of protein S was compared to that of TFPI in concentrated conditioned media (B).

To determine the reason for this discrepancy, I assessed how dependent the total anticoagulant activity of TFPI/protein S is upon the concentration of TFPI. Consequently, purified TFPI was titrated (2.5, 5 and 7.5 nM) in the presence or absence of 60 nM protein S (Figure 3.30). I observed that a small difference, in terms of anticoagulant activity, between 2.5, 5 and 7.5 nM TFPI in the absence of protein S translated into a large difference in the presence of protein S. Previous results (Figure 3.25), in which a titration of protein S was performed at a fixed concentration of TFPI, showed only small changes in cofactor activity between 30 nM increments in concentration of protein S. Taken together, these results (Figure 3.25 and Figure 3.30) suggest that the total TFPI anticoagulant activity in

the presence of protein S is more dependent upon the TFPI concentration than upon the protein S concentration.

The different anticoagulant activities of TFPI in concentrated conditioned media or in fully purified TFPI is likely attributable to inaccuracies in determination of the concentration of TFPI in concentrated conditioned media, as this and the initial experiments were performed prior to complete optimisation of our in-house TFPI ELISA. As TFPI in conditioned media was expressed in HEK293 cells and my purified TFPI was expressed in Chinese hamster ovary cells, the glycosylation will be different. However, this is unlikely to be the reason behind the discrepancy, as the purified TFPI from the Chemoserotheraputic Institute Kaketsuken was shown to have the same activity as plasma purified TFPI,<sup>259</sup> and the glycosylation from human HEK293 cells is likely to be broadly similar to that of plasma TFPI. In addition, Huang et al. evaluated the inactivation of FXa both by bacterial and mammalian expressed TFPI and found no significant difference, suggesting posttranslational modifications do not play a major role.<sup>72</sup> To determine the reason behind the discrepancies between the two different sources of TFPI, I believe that the ELISA should be repeated in future studies and the TFPI from HEK293 cells purified and assessed alongside TFPI from the Chemoserotheraputic Institute Kaketsuken in functional assays.



**Figure 3.30 Protein S enhancement of TFPI at different concentrations of TFPI.** Thrombin generation in the presence of different concentrations of TFPI (2.5, 5 and 7.5 nM) was assessed in the absence (A) and presence (B) of 60 nM protein S.

To ascertain whether the different assay conditions (i.e. source of TFPI and TFPI concentration) influenced the outcome of the initial screenings (Figure 3.25, Figure 3.27 and Figure 3.28), the activity of some of the protein S variants (60 nM) was reassessed using 5 nM purified TFPI. Protein S variants Face2, D95N, Q79A, T103A and K112A, previously assayed in the presence of 2.5 nM TFPI in conditioned media (Figure 3.25 and Figure 3.28) were reassessed using 5 nM purified TFPI (Figure 3.31B,C,D). The results obtained with these protein S variants were consistent with earlier results, showing that protein S D95N and Q79A had somewhat reduced TFPI cofactor activity, while protein S Face2, T103A, K112A had slightly enhanced TFPI cofactor activity. This suggests that while the total effect of protein S in the presence of TFPI varies between different assay conditions, variants with reduced, normal or enhanced TFPI cofactor activity in comparison with WT protein S are reproducibly visualised in both assay conditions. Protein S variants Face1, Gla1 and TSR, that had not previously been assessed for TFPI cofactor activity, were also analysed (Figure 3.31A,C). Of the newly screened variants, protein S Gla1 had a similar TFPI cofactor activity to that of WT protein S, protein S Face1 had reduced cofactor activity and protein S TSR an increased TFPI cofactor activity (Figure 3.31). To further assess protein S variants with altered TFPI cofactor activity, protein S TSR, in addition to protein S D95A and protein S Face2, were purified and evaluated in the FXa inhibition assay. Protein S Face1 has been investigated by my colleague Dr Josefin Ahnström.



Figure 3.31 Thrombin generation in the presence of TFPI and WT protein S and protein S variants.

Thrombin generation was assessed in protein S deficient plasma supplemented with 5 nM purified TFPI. The cofactor activity of protein S variants towards TFPI was assessed at 60 nM.

Of the variants screened up to this point for their TFPI cofactor activity in the CAT assay, none had completely impaired cofactor activity, in terms of both impaired prolongation of the lag time and an impaired reduction of the peak height. In a further attempt to identify residues of protein S important for TFPI cofactor activity, additional variants in the Gla-TSR-EGF1 domains were made. The Gla-TSR-EGF1 model<sup>193</sup> was used to identify potential new residues to substitute. However, not many residues were left to mutate in the Gla domain once those close to the  $\omega$ -loop,  $\gamma$ -carboxylated residues, alanine residues, cysteine residues, residues with a buried side chain and those already substituted, were excluded. All charged residues in the TSR had already been substituted and had been

incorporated into the protein S TSR variant. Polar residue Asn74 in the TSR was, however, chosen to create an additional point substitution, protein S N74A, as it is in close proximity of the partially conserved calcium binding motif in the EGF1. A small cluster of amino acid residues with an exposed R group, not conserved in the EGF1 domains of other coagulation factors, were identified on one side of EGF1. These were substituted to create the composite variant, protein S NEDM (N86A/E87Q/D88A/M91A). Asn74 was also incorporated in this composite variant creating composite variant protein S NNEDM. As single point variants might not be enough to disrupt protein S cofactor activity towards TFPI, a composite variant composed of the previously examined point substitutions K94A, T103A and K105A was also made, generating the variant protein S KTK. These residues were selected as T103N is a natural protein S variant in which reduction in APC cofactor activity could not be fully explained using purified FVa and FVIIIa inactivation assays.<sup>255</sup> Residues Lys94 and Lys105 were selected as they were charged and in close proximity to Thr103, and these three residues formed a cluster on a separate side to that of NEDM.

The new variants were screened by CAT at 60 nM in the presence of 2.5 nM purified TFPI (Figure 3.32, Panel A and B). The protein S variants were also simultaneously evaluated for their APC cofactor activity as this had not been done previously (Figure 3.32, Panel C and D). Protein S N74A had a normal APC cofactor activity (Panel C) and a slightly enhanced TFPI cofactor activity (Panel A). Composite variants protein S NEDM and protein S KTK had normal APC cofactor activity (Panel C), but reduced TFPI cofactor activity (Panel A). The addition of the N74A substitution to protein S NEDM, creating protein S NNEDM, resulted in a variant with both reduced APC cofactor activity (Panel D) and TFPI cofactor activity (Panel B). As can be seen in Figure 3.32 protein S NNEDM has reduced APC cofactor activity in comparison with protein S NEDM, however both variants have a similar TFPI cofactor activity. To further analyse these variants, protein S N74A, KTK, NEDM and NNEDM were purified to assess them in the FXa inhibition assay. Purification of protein S NEDM was not successful and it could not be re-expressed and purified, due to the time constraints of writing up this thesis. Protein S N74A, protein S KTK and protein S NNEDM were, however, successfully purified and further analysed in the FXa inhibition assay (see section 3.4.2.2).



Figure 3.32 WT protein S, protein S N74A, KTK, NEDM and NNEDM studied by thrombin generation in the presence of TFPI or APC.

Thrombin generation was assessed in protein S deficient plasma supplemented with protein S either in the presence of 2.5 nM TFPI (A, B) or 4 nM APC (Xigris) (C, D). Protein S was used at 60 nM to evaluate TFPI cofactor activity and 90 nM to evaluate APC cofactor activity. Protein S N74A, KTK and NEDM all had normal APC cofactor activity (C). Addition of substitution N74A to protein S NEDM creating protein S NNEDM resulted in a partial loss of APC cofactor activity (D). Protein S N74A had an increased TFPI cofactor activity, while the cofactor activity of protein S KTK and NEDM towards TFPI was reduced (A). Addition of substitution N74A to protein S NEDM towards TFPI was reduced (A). Addition of substitution N74A to protein S NEDM did not affect its cofactor activity towards TFPI (Panel B in comparison with Panel A). Note that APC from Xigris was used in this experiment and was more active than the commercial APC preparation mostly used in this thesis (Enzyme Research Laboratories).

# 3.4.2 Evaluation of TFPI cofactor activity of protein S in the FXa inhibition assays

#### 3.4.2.1 Evaluation of FXa inhibition by WT purified TFPI and protein S

The FXa inhibition assay was optimised with purified TFPI (Chemoserotheraputic Institute, Kaketsuken) and purified plasma protein S (Enzyme Research Laboratories). Importantly, a recent paper by Mori *et al.*, (from the

Chemoserotheraputic Institute Kaketsuken) showed that the protease inhibitior activity of the recombinant purified TFPI towards FXa and TF/FVIIa was identical to that of plasma purified TFPI.<sup>259</sup> Protein S in media could not be used as an unknown protease in both this preparation and the mock media (without protein S) was able to cleave the FXa substrate in a dose-dependent manner (results not shown).

To assess the specificity of FXa inhibition by TFPI, purified TFPI (0-8 nM) was titrated and was shown to inhibit FXa in a dose-dependent manner (Figure 3.33). To assess the cofactor activity of protein S, low concentrations of TFPI (0.5-2 nM) were used to maximise the detection of the enhancing effect in the presence of protein S. Purified plasma protein S (0-160 nM) was titrated in the presence of 1.5 nM TFPI (Figure 3.34). A dose-dependent enhancement of TFPI was observed suggesting that the activity of TFPI was specifically enhanced by protein S. Importantly, protein S did not affect FXa activity in the absence of TFPI (as shown in subsequent figures), confirming the high specificity of the assay.





TFPI (0-8 nM) was incubated with phospholipids (10  $\mu$ M), CaCl<sub>2</sub> (5 mM) and the FXa chromogenic substrate S-2765 (200  $\mu$ M) in a 20 mM Tris-HCl, 140 mM NaCl, 5 mg/ml BSA buffer. The assay was started by the addition of FXa (0.5 nM) and the absorbance at 405 nm was read over time.



Figure 3.34 Dose-dependent enhancement of TFPI activity by protein S. TFPI (1.5 nM) was incubated with 0-160 nM protein S, phospholipids,  $CaCI_2$  and the FXa chromogenic substrate S-2765. The assay was started by the addition of FXa and the absorbance at 405 nm was read over time.

As previously reported, the TFPI cofactor activity of protein S was phospholipid dependent (Figure 3.35).<sup>81</sup> In accordance with the literature,<sup>72</sup> TFPI mediated inhibition of FXa did not require phospholipids although its activity was moderately enhanced by their presence (see effect of TFPI on FXa inhibition in Panel A and B of Figure 3.35). This could be due to co-localisation of FXa and TFPI on the phospholipid surfaces. FXa binds phospholipids through its Gla domain and it has been suggested that the positively charged C-terminal tail of TFPI may interact with negatively charged phospholipid surfaces and cell surfaces<sup>74-76</sup> and FXa<sup>79,80</sup>. No difference in TFPI inhibition of FXa and protein S cofactor activity towards TFPI was observed between 10 and 50  $\mu$ M phospholipids suggesting that saturating amounts of phospholipids were present at 10  $\mu$ M phospholipids. This is potentially important, as protein S has previously been shown to have a protein S direct activity (independent of APC and TFPI) due to competition for phospholipid surfaces when these are present in limiting amounts.<sup>189,190,242</sup>



Figure 3.35 Inhibition of FXa by TFPI and protein S in the absence and presence of phospholipids.

TFPI (1.5 nM) and protein S (20 nM) were incubated with  $CaCl_2$ , S-2765 with no phospholipids (A), 10  $\mu$ M phospholipid vescicles (B) or 50  $\mu$ M phospholipid vescicles (C). The assay was started by the addition of FXa and the absorbance at 405 nm was measured over time. TFPI inhibition of FXa cleavage of S-2765 was observed in the absence of phospholipids (Panel A) but was enhanced by their presence (Panel B and C). Protein S enhanced the TFPI mediated inhibiton of FXa in the presence (Panel B and C) but not in the absence (Panel A) of phospholipids. Protein S did not affect FXa cleavage of S-2765 in the absence of TFPI in any of the experimental conditions.

#### 3.4.2.2 <u>Evaluation of domains and amino acids of protein S involved in TFPI</u> cofactor activity by use of domain specific monoclonal antibodies and protein S variants

In an attempt to identify the domains of protein S important for TFPI cofactor activity, monoclonal domain specific antibodies against protein S (kind gift of Professor Björn Dahlbäck and Sofia Carlsson, University of Lund, Sweden) were used to inhibit protein S enhancement of TFPI. Protein S (20 nM) was preincubated in the presence or absence of monoclonal antibody (200-400 nM) in the presence of CaCl<sub>2</sub> (as some of the antibodies are calcium dependent, the final concentration of CaCl<sub>2</sub> in the well was 5 mM) for up to 30 minutes. Phospholipids and the FXa chromogenic substrate were subsequently added and the reaction was started by addition of 0.5 nM FXa (Figure 3.36). None of the monoclonal antibodies tested completely inhibited protein S enhancement of TFPI. The antibodies against the Gla domain of protein S, MK21 and MK47, showed the most inhibition of protein S cofactor activity. However, these monoclonal antibodies have previously been reported to inhibit phospholipid binding, which is a prerequisite for protein S function.<sup>243</sup> It was therefore not possible to conclude whether the Gla domain of protein S is involved in an interaction with TFPI. Monoclonal antibodies against the TSR domain, the EGF1 domain, the EGF3-4 domain and the SHBG domain had no or little effect upon protein S enhancement of TFPI (Figure 3.36). While this would suggest that the epitopes recognised by these antibodies are not involved in an interaction with TFPI, it is possible that other amino acid residues within the respective domains may be involved and remain accessible. Interestingly, the epitope of the monoclonal antibody MK54 against the EGF1 domain of protein S includes residue Lys97.<sup>260</sup> It is therefore likely that this antibody would inhibit a potential interaction involving Asp95 in protein S with TFPI. No monoclonal antibody directed against the EGF2 domain was available.



## Figure 3.36 Inhibition of protein S cofactor activity towards TFPI by monoclonal domain specific antibodies.

Protein S (20 nM) was preincubated in the presence or absence of 200-400 nM of the domain specific monoclonal antibody against protein S in the presence of CaCl<sub>2</sub> for up to 30 minutes. TFPI (1.5 nM), S-2765 and phospholipids were subsequently added and the reaction was started by addition of FXa. Monoclonal antibodies against the Gla domain, TSR domain and the EGF1 domain are shown in Panel A. Monoclonal antibodies directed against the EGF3-4 domain and the SHBG domain are shown in Panel B. All data point were normalised against the control. Results represents the mean from two different experiments performed in duplicate. The labelling TFPI + protein S has been omitted in the presence of the antibody. The domain against which the antibody is directed precedes the name of each antibody.

Of the over 30 protein S variants screened for their TFPI cofactor activity in plasma (section 3.4.1), those of interest were purified and assessed in the FXa inhibition assay. Protein S Face2 and protein S TSR had been purified by anion exchange chromatography followed by immunoaffinity chromatography and were compared with WT protein S purified using the same method. Figure 3.37 shows

the protein S Face2 and protein S TSR cofactor enhancing activity in the FXa assay compared with that of WT protein S (20 nM) using 1.5 nM TFPI. Both protein S Face2 and protein S TSR showed a similar TFPI cofactor activity to that of WT protein S. The same results were observed over a range of TFPI (0.75-2 nM) and protein S (5-40 nM) concentrations (not shown) suggesting residues mutated in these variants are not critical for TFPI cofactor activity. This is in agreement with the results obtained in the plasma assay.



Figure 3.37 TFPI inhibition of FXa in the presence and absence of WT protein S, protein S Face2 and protein S TSR.

TFPI (1.5 nM) was incubated with 20 nM protein S, phospholipids,  $CaCI_2$  and the FXa chromogenic substrate S-2765. The assay was started by the addition of FXa and the absorbance at 405 nm was read over time.

When protein S D95A (purified by anion exchange chromatography followed by immunoaffinity chromatography) was tested in the FXa inhibition assay, it appeared to have a reduced cofactor activity for TFPI. Due to lack of material, protein S D95A in conditioned media was re-purified by barium citrate precipitation followed by anion exchange chromatography alongside WT protein S, protein S N74A, KTK and NNEDM. Protein S variants (20 nM) were screened for their TFPI cofactor activity at 1.5 nM TFPI and were compared with WT protein S purified in the same way. Protein S KTK and protein S N74A had TFPI cofactor activities similar to that of WT protein S, as can be seen in Figure 3.38. The cofactor activities of protein S D95A and protein S NNEDM were, however, reduced in comparison with that of WT protein S (Figure 3.38). These results are

in agreement with those obtained in plasma, with the exception of protein S KTK which exhibited reduced cofactor activity in plasma.

None of the protein S preparations affected FXa cleavage of S-2765 in the absence of TFPI (results not shown). To further investigate the protein S cofactor activity for TFPI, kinetic assays were performed to allow determination of the  $K_i$  values and, consequently, to quantify the difference between WT protein S and protein S variants.





TFPI (1.5 nM) was incubated with 20 nM protein S, phospholipids,  $CaCI_2$  and the FXa chromogenic substrate S-2765. The assay was started by the addition of FXa and the absorbance at 405 nm was read over time.

#### 3.4.2.3 <u>Kinetic analysis of TFPI mediated inhibition of FXa in the presence and</u> <u>absence of protein S</u>

To calculate the inhibition constant, K<sub>i</sub>, of TFPI for FXa, the K<sub>m</sub> for the hydrolysis of the chromogenic substrate S-2765 by FXa was first determined. The FXa inhibition assay was performed as described in section 2.7 but in the absence of TFPI and protein S and at a fixed concentration of FXa (0.5 nM). The substrate was titrated (0-1200  $\mu$ M) and the initial velocity V<sub>0</sub> was plotted against the substrate concentration (Figure 3.39). Curve fitting of the initial velocities allowed determination of the K<sub>m</sub>. A K<sub>m</sub> of 46.3 ± 11.7  $\mu$ M was obtained from four different experiments performed in duplicate. This is compatible with the K<sub>m</sub> of 42.6 ± 2.3  $\mu$ M reported by Broze and colleagues.<sup>72</sup>



**Figure 3.39 Determination of the K**<sub>m</sub> **of FXa for the chromogenic substrate S-2765.** The K<sub>m</sub> of the hydrolysis of S-2765 by FXa was determined by titrating the substrate over the range 0-1200  $\mu$ M in the presence of phospholipids and CaCl<sub>2</sub> and in the absence of TFPI and protein S. A representative experiment is shown (A). The mean ± SD of the V<sub>0</sub> from four independent experiments performed in duplicate was plotted against the concentration of the substrate (B). A K<sub>m</sub> of 46.3 ± 11.7  $\mu$ M was obtained.

When the FXa inactivation assay was started either by the addition of FXa (as described by Ndonwi *et al.*<sup>238</sup>) or substrate and TFPI (similar to what was described by Hackeng *et al.*<sup>81</sup>), a difference in the V<sub>0</sub> was observed in the presence of TFPI and protein S. While the respective absorbance values at 30 minutes were similar, the shapes of the curves were appreciably different. When the assay was started with FXa, the data points describing the protein S enhancement of TFPI formed a distinctive curve with a high V<sub>0</sub> (Figure 3.40, Panel A). However, when the assay was started with the addition of substrate and TFPI, the protein S enhancement of TFPI was more linear and resulted in a lower V<sub>0</sub> (Figure 3.40, Panel B). These differences can be more easily visualised by looking at the first 5 minutes of the assay (Figure 3.40 C and D). An explanation for these discrepancies is not yet available.



Figure 3.40 FXa inhibition assay started by the addition of FXa or by the addition of TFPI and S-2765.

Protein S (100 nM) enhancement of TFPI (0-4 nM) was evaluated. The assay was started either by addition of FXa (A) or addition of TFPI and S-2765 (B). Panels C and D shows the first 5 minutes of the results shown in Panel, A and B, respectively. Representative graphs are shown.

As the K<sub>i</sub> of TFPI for FXa is dependent upon calculation of the V<sub>0</sub>, the FXa inhibition assay was started with TFPI and S-2765 where a change in V<sub>0</sub> is more readily seen. The K<sub>i</sub> values of TFPI in the presence or absence of protein S determined in the literature were derived by starting the FXa inhibition assay with FXa and then, after approximately 5 minutes, dispensing TFPI.<sup>81</sup> To determine the K<sub>i</sub>, the timepoint at which TFPI was added was considered to be timepoint zero. As no dispenser was available for my experiments, the K<sub>i</sub> values of TFPI in the presence or absence of protein S were determined by starting the FXa inhibition assay with a mix of substrate and TFPI. Titration of 0-8 nM TFPI was performed in the absence of protein S. As protein S efficiently enhances TFPI, the concentration of TFPI was lowered to 0-4 nM when using 100 nM protein S. Figure 3.41 shows the activity of TFPI in the absence (Panel A) and presence of

WT protein S purified by anion exchange chromatography followed by immunoaffinity chromatography (Panel B) or of WT protein S purified by barium citrate precipitation followed by anion exchange chromatography (Panel C). This is expressed in Figure 3.41, Panel D as a change in V<sub>0</sub> at increasing concentrations of TFPI in the presence or absence of protein S. The V<sub>0</sub>, derived from the fitted equations in PanelA-C, has been expressed as a percentage of V<sub>max</sub> (which corresponds to the V<sub>0</sub> obtained in the absence of TFPI). The results in Panel D represent the mean  $\pm$  SD of three independent experiments. As can be seen in Panel D of Figure 3.41 both WT preparations enhanced TFPI to a similar extent.



Figure 3.41 FXa inhibition by TFPI in the presence and absence of WT protein S. TFPI (0-8 nM) was titrated in the absence (A) or presence of 100 nM WT protein S purified by anion exchange chromatography followed by immunoaffinity chromatography (B), or by barium citrate precipitation followed by anion exchange chromatography (C). The change in V<sub>0</sub> at increasing concentrations of TFPI in the presence or absence of WT protein S is shown in Panel D as mean  $\pm$  SD from three independent assays. Anion chr, purified by barium citrate precipitation followed by anion chromatography; Affinity chr, purified by anion exchange chromatography followed by immunoaffinity chromatography.

Protein S D95A and protein S Face2 previously found to be critical for APC cofactor activity and protein S NNEDM, which had reduced TFPI and APC cofactor activity in the plasma assay, were assessed at 100 nM, at increasing concentrations of TFPI (0-4 nM). Protein S KTK was also assessed. Protein S KTK had normal APC cofactor activity. While plasma assays suggested it had reduced TFPI cofactor activity, its activity in the FXa inhibition assay seemed normal. The change in V<sub>0</sub> in the presence or absence of the protein S variants in comparison with WT protein S purified in the same way is shown in Figure 3.42. Protein S Face2 (Panel A), protein S KTK (Panel B) and protein S D95A (Panel C) all showed similar dose-dependent changes in V<sub>0</sub> to that of WT protein S. Protein S NNEDM (Panel D), however, had a reduced cofactor activity towards TFPI, visualised as a smaller reduction in V<sub>0</sub>.



# Figure 3.42 Change in $V_0$ at increasing concentrations of TFPI in the presence or absence of protein S.

TFPI (0-8 nM) was titrated in the presence or absence of 100 nM protein S. The graphs show the change in  $V_0$  at increasing concentrations of TFPI in the presence or absence of protein S Face2 (A), protein S KTK (B), protein S D95A (C) and protein S NNEDM (D). Data points are expressed as the mean  $\pm$  SD from three independent experiments. Anion chr, purified by barium citrate precipitation followed by anion exchange chromatography; Affinity chr, purified by anion exchange chromatography.

To quantify the difference between WT protein S and the protein S variants, the  $K_i$  values were determined. This was done by plotting the  $V_{max}/V_0$  against the concentration of TFPI. The x intercept was then used to calculate the  $K_i$ , as described in section 2.7.3. Differences in  $K_i$  values obtained were analysed using an unpaired t-test.

There was no significant difference between the K<sub>i</sub> values obtained with WT protein S purified by anion exchange chromatography followed by immunoaffinity chromatography (K<sub>i</sub>=  $0.13 \pm 0.03$  nM) or WT protein S purified by barium citrate precipitation followed by anion exchange chromatography (K<sub>i</sub>=  $0.19 \pm 0.04$  nM) (Table 3.4).

When protein S variants were compared to WT protein S purified using the same method, there was no significant difference between protein S Face2 and WT protein S in terms of K<sub>i</sub> values (Table 3.4). This is in agreement with what I had seen in the single concentration screening assay (Figure 3.37). Confirming what was observed in the single concentration screening assay (Figure 3.38), protein S KTK enhancement of TFPI inhibition of FXa was not significantly different from WT protein S (Table 3.4). Interestingly, while protein S D95A seemed to have reduced cofactor activity towards TFPI at 20 nM in the screening assays (Figure 3.38), no difference was observed at 100 nM protein S which was used to determine the K<sub>i</sub> of TFPI activity (Table 3.4). Importantly, my experiments with protein S NNEDM confirmed results obtained in the screening assay (Table 3.4). Protein S NNEDM had reduced TFPI cofactor activity ( $K_i = 0.91 \pm 0.35$  nM) in comparison with WT protein S ( $K_i = 0.19 \pm 0.04$  nM). While some of the residues in protein S NNEDM might be important for cofactor activity towards TFPI mediated inhibition of FXa, this variant also had reduced APC cofactor activity. Evaluation of protein S NEDM, which had normal APC cofactor activity might therefore provide interesting results in future experiments.

	K <sub>i</sub> (nM)
TFPI	3.49 ± 0.96
WT protein S (anion chr)	0.19 ± 0.04
Protein S 95A	0.21 ± 0.06
Protein S KTK	$0.24 \pm 0.05$
Protein S NNEDM	0.91 ± 0.35
WT protein S (affinity chr)	0.13 ± 0.03
Protein S Face2	0.12 ± 0.02

# Table 3.4 Inhibition constants for TFPI inhibition of FXa in the presence or absence of protein S.

 $K_i$  values (nM) for WT protein S and protein S variants were determined from the x intercept obtained by plotting the  $V_{max}/V_0$  values against TFPI concentration. Results are expressed as the mean ± SD from three independent experiments. WT protein S (anion chr) is protein S purified by barium citrate precipitation followed by anion exchange chromatography. WT protein S (affinity chr) is protein S purified by anion exchange chromatography followed by immunoaffinity chromatography.

#### 3.4.3 Discussion

The CAT assay to assess protein S cofactor activity towards TFPI was initially set up. This assay allows evaluation of the cofactor activity of protein S towards TFPI in a plasma milieu, potentially representing a more physiological environment than purified systems. When I began my thesis, a method to evaluate protein S cofactor activity towards TFPI by CAT in protein S deficient plasma had not been published. However, recently Maurissen *et al.* were able to reproducibly study the TFPI/protein S anticoagulant activity by CAT in normal plasma.<sup>240</sup> This was achieved by using inhibitory polyclonal antibodies against TFPI and protein S.

During my thesis, I noticed differences in the anticoagulant activity of TFPI and protein S supplemented to protein S deficient plasma from different suppliers. Generally, as can be seen by the different peak heights (Figure 3.22), the protein S deficient plasma from Hyphen Biomed was less procoagulant that that of Affinity Biologicals. This is likely to influence the total TFPI/protein S anticoagulant activity as the latter has been shown to be dependent upon TF concentrations, and therefore the procoagulant environment.<sup>218</sup> Alternatively, the reported

difference in the concentration of endogenous TFPI between the different plasma preparations might be important. Co-depletion of full-length TFPI following removal of protein S during the production of protein S deficient plasma has recently been shown by Castoldi *et al.*<sup>241</sup> However, differences in endogenous TFPI concentrations alone, are unlikely to explain the differences I observed between the different plasma preparations. Firstly, because the exogenous full-length TFPI (~2.5 nM) supplemented to the protein S deficient plasma exceeds by far the endogenous full-length TFPI (~0.25 nM) the total amount of TFPI in the two different plasma preparations were fairly similar. Secondly, the anticoagulant activity of 5 nM TFPI and protein S in plasma from Affinity Biomed was less than that observed using 2.5 nM TFPI and protein S in plasma from Hyphen Biomed (results not shown). Therefore, it is possible that co-depletion of other coagulation factors might account for the differences observed between the two plasma preparations, Affinity Biological and Hyphen Biomed.

It is interesting to note that the protein S/TFPI anticoagulant activity influences both the peak height, ETP and the lag time. This is in contrast with what is observed in the presence of APC, where protein S cofactor enhancement results mainly in an effect on peak height and ETP. Compatibly with my results, Hackeng and Rosing observed both a reduction in peak height and prolongation of the lag time following addition of TFPI to TFPI depleted plasma.<sup>261</sup> Other studies evaluating protein S enhancement of TFPI anticoagulant activity have been carried out in normal plasma. When Brodin et al. evaluated the effect of addition of recombinant TFPI to normal plasma, they noticed a prolongation of the lag time but saw no effect on the ETP.<sup>253</sup> Similarly, when they inhibited endogenous TFPI with antibodies, a shortening in the lag time but no effect on the ETP was observed. Previous reports have also shown that protein S, in the absence of APC, affects the lag time but not the ETP.<sup>262</sup> This study, however, combined protein S deficient plasma with different percentages of normal plasma, thus potentially confounding any effect of differences in other coagulation factors between the two plasma preparations. In contrast, when I supplemented TFPI to normal plasma (results not shown) I observed both an effect on the peak height and a prolongation of the lag time. Maurissen et al. assessed TFPI anticoagulant activity and its enhancement by protein S in normal plasma using inhibitory antibodies and showed a change in peak height, but not in lag time.<sup>240</sup> The

discrepancies between results from different protein S deficient plasma preparations and between protein S deficient plasma and normal plasma do not have a coherent explanation. It is possible that co-depletion of other coagulation factors contribute to the different TFPI/protein S anticoagulant activity observed between different plasma preparations. Considering that plasma levels of FV have been shown to correlate with full-length TFPI plasma levels, that TFPI and FV potentially form a complex *in vivo*,<sup>263</sup> that depletion of full-length TFPI results in co-depletion of FVa,<sup>241</sup> together with the observation that FV levels have been shown to influence the lag time in the CAT assay,<sup>262</sup> it should be evaluated if differences in FV level could contribute to the differences observed between normal and protein S depleted plasma.

To decrease intra-assay variability, variability between different sources of plasma and variabilities between different laboratories, parameters such as supplier of the plasma, potential differences between lyophilised and frozen plasma, batch variability of the plasma, source of TF, source of phospholipids, preincubation of additional components to the plasma, plasma dilution and temperature should all be carefully considered.<sup>218,262,264,265</sup>

In this thesis, the TFPI anticoagulant activity in the presence or absence of protein S was evaluated in protein S deficient plasma from Hyphen Biomed and the specificity of the assay was demonstrated with inhibitory antibodies against TFPI, protein S or protein C. As observed for APC (section 3.3.2), no significant difference in TFPI cofactor activity was observed between protein S in concentrated conditioned media and purified protein S. The assay developed during this thesis allowed visualisation of a clear TFPI effect and its enhancement by protein S. I also showed that the protein S activity in the presence of TFPI is highly dependent on the TFPI concentration and its anticoagulant activity. This is compatible with results of Hackeng and Rosing in their recent review paper in which TFPI depleted plasma was supplemented with TFPI in the presence or absence of inhibitory antibodies against protein S.<sup>261</sup> The dependence of the protein S/TFPI anticoagulant activity upon the TFPI concentration might be partially explained by the fact that the normal concentration of protein S present in plasma (free ~145 nM, total ~362 nM) is in vast excess over that of free full-length TFPI (~0.25 nM).

The inter-assay variability was quite large making it difficult to pool data from multiple assays. The CAT assay used in this thesis was, however, an efficient screening tool to identify protein S variants with reduced, normal or enhanced TFPI cofactor activity in comparison to WT protein S. Protein S variants in concentrated conditioned media were evaluated both for their ability to reduce the peak height of thrombin generated and to prolong the lag time. Of the 31 protein S variants screened for their TFPI cofactor activity by CAT none had completely ablated activity. Twenty-three protein S variants had normal to enhanced activity in comparison with WT protein S (Table 3.5). The protein S variants previously found to have reduced APC cofactor activity protein S D95A, D95N, D78A and Q79A had approximately 80% of WT protein S cofactor activity towards TFPI (Figure 3.25). Interestingly, protein S Face2, which had decreased APC cofactor activity, reduced the peak height of thrombin to approximately half of that of WT protein S (at 90-120 nM) in the presence of TFPI (Figure 3.25 and Figure 3.31). Protein S KTK, NEDM, NNEDM and Face1 (Table 3.5) showed reduced cofactor activity towards TFPI in comparison with WT protein S when screened by CAT (Figure 3.31 and Figure 3.32). This was visualised mainly as a change in lag time rather than in a change of peak height in comparison to WT protein S.

TFPI cofactor activity compared to WT protein S	Protein S variants
Normal to enhanced	Gla1, Face2, TSR, N74A, K94A, K97A, S99A, T101A, T103A, K105A, P106A, Q109A, E111A, K112A, D135A, T137A, Y141A, H142A, S144A, K146A, N154A, K156A, D157A
Reduced	Face1, KTK, NEDM, NNEDM, D78A, Q79A, D95A, D95N

# Table 3.5 TFPI cofactor activity of protein S variants in comparison with WT protein S in plasma studied by CAT assay.

Any variant which in the CAT assay had shorter lag time or increased peak in comparison with WT protein S was classified in this table as having reduced activity compared to WT protein S.

Protein S variants Face2 and D95A, found to be important for APC cofactor activity, were purified and evaluated for their TFPI cofactor activity in the FXa

inhibition assay. Protein S KTK and NNEDM which had reduced TFPI cofactor activity in plasma were also purified and assessed in the FXa inhibition assay. Protein S N74A and TSR which had normal APC cofactor activity and normal to enhanced TFPI cofactor activity in plasma were also analysed in the FXa inhibition assay.

The FXa inhibition assay has the advantage of being a well established assay and allows specific evaluation of TFPI mediated inhibition of FXa without the complexity associated with normal plasma assays.<sup>68,72,78,81,237</sup> Importantly, purified protein S alone does not affect FXa cleavage of the chromogenic substrate S-2765 in the absence of TFPI ensuring the assay specifically assesses TFPI cofactor activity of protein S. A screening assay using 1.5 nM purified TFPI and 20 nM purified protein S was performed. Results from the screening assay showed that protein S variants Face2, TSR, KTK and N74A all had similar TFPI cofactor activity to that of WT protein S. Protein S D95A and protein S NNEDM, however, had a reduced TFPI cofactor. My colleague Dr Josefin Anhnström evaluated the cofactor activity of protein S E36A (a substitution present in the Face1 variant that had reduced APC cofactor activity, see section 3.3.6) and found it enhanced TFPI mediated inhibition of FXa to the same extent as WT protein S.

To more accurately quantify the differences between WT protein S and protein S variants, I derived the  $K_i$  values for TFPI inhibition of FXa in the presence of protein S. WT protein S reduced the  $K_i$  of TFPI in the absence of protein S from 3.49 nM to 0.13-0.19 nM, enhancing TFPI inhibition of FXa by 18-27-fold. The  $K_i$  values of protein S Face2, D95A, KTK and NNEDM were determined. No statistical difference between the  $K_i$  obtained in the presence of protein S Face2, D95A, KTK with that of WT protein S was found. Protein S NNEDM, however, enhanced TFPI inhibition of FXa only by ~3.8-fold in contrast to the ~18-fold of WT protein S. These kinetic results were in agreement with results from the screening in the FXa inhibition assay, with the exception of protein S D95A. While protein S D95A had reduced TFPI cofactor activity in the screening experiment in the FXa inhibition assay, the  $K_i$  value obtained with 100 nM protein S was not statistically different from that of WT protein S. It is possible that while at lower protein S concentrations protein S D95A is not as efficient as WT protein S, this is overcome at saturating concentrations of protein S. Overall, results from the FXa

inhibition assay correlate relatively well with those from the plasma assay, considering the intrinsic differences of the two assays. The main exception is protein S KTK, which had reduced TFPI cofactor activity in plasma and normal TFPI cofactor activity for the inhibition of FXa. The reason for this discrepancy still needs to be evaluated.

Importantly, results from the plasma based assay and the purified FXa inhibition assay suggest that protein S D95A, protein S Face2 and protein S E36A (result of Dr Josefin Anhnström, not shown) were able to enhance TFPI mediated inhibition of FXa to a similar extent as WT protein S. This is of considerable importance as it suggests distinct residues in protein S interact with APC and TFPI.

Further work has to be done to identify residues in protein S involved in TFPI cofactor activity. Experiments with protein S KTK should be repeated to evaluate if there are any significant differences between the plasma based assay and the purified assay, and if so to determine the reason for any discrepancy. Protein S NNEDM had reduced TFPI cofactor but also reduced APC cofactor activity. Importantly, protein S NEDM had reduced TFPI cofactor activity and normal APC cofactor activity. Protein S NEDM should be purified and analysed in the FXa inhibition assay.

Further work also needs to be done in order to fully optimise the plasma based assay and potentially also the FXa inhibition assay. In the first instance the interassay variability of the plasma based assay has to be improved. Secondly, the different effect on V<sub>0</sub> in the FXa inhibition assay according to how the assay is started (Figure 3.40) has to be rationalised. While the reason for this is not clear, it might explain why Ndonwi *et al.*<sup>238</sup> quantifies the protein S enhancement of TFPI as a change in V<sub>s</sub> determining the EC<sub>50</sub> and Hackeng *et al.*<sup>81</sup> as a change in V<sub>0</sub> determining the K<sub>i</sub>. To get accurate V<sub>0</sub> results a dispenser should also be used. Protein S enhancement of TFPI mediated inhibition of TF/FVIIa should also

Results obtained during my thesis would suggest that substitution of a single residue of protein S is not enough to disrupt its cofactor activity towards TFPI. Even if protein S and TFPI interact through multiple interaction sites it is unlikely that residues substituted in my thesis and found to have normal to enhanced TFPI cofactor activity are involved in a critical way in the interaction with TFPI. Figure 3.43 shows the residues in protein S substituted in the Gla-TSR-EGF1

domain and analysed in this thesis (blue). Red residues in Figure 3.43 represent amino acids potentially involved in phospholipid binding, alanine residues or cysteine residues and are therefore not good candidates to substitute. Similarly, yellow residues represent residues which are predicted to have their R group buried inside the protein S structure. This leaves only few clusters of residues in these domains, of which only two charged residues, that have not been mutated during this thesis (gray) and could be evaluated alongside protein S KTK and NEDM. Alternatively, the interaction site between protein S and TFPI is not located at the sites and domains evaluated in this thesis.



**Figure 3.43 Residue substitutions in the Gla-TSR-EGF1 domains made during this project.** Residues substituted during this project, either as single or composite variants, are coloured in blue. Residues in red and yellow were not mutated because their likely involvement in phospholipid binding, because they are alanine or cysteine residues or because their R group is buried within the protein S structure. Protein S model from Villoutreix *et al.*.<sup>193</sup>

### **4 CONCLUDING REMARKS**

It is well established that protein S acts as a cofactor for APC. Recently, different groups have shown that protein S is also able to act as a cofactor for TFPI.<sup>81,237</sup> The aim of my project was to elucidate the molecular mechanism behind these interactions by identifying the residues in protein S involved in APC and TFPI cofactor activity. To address this aim, more than 30 protein S variants were created by site-directed mutagenesis, expressed and characterised for their cofactor activities towards both APC and TFPI. In this concluding chapter, I will attempt to put into context the results achieved during my PhD with what is already known in the literature. I will first address the issue of protein S cleavage, multimerisation and binding to phospholipids. I will postulate an interaction mechanism between protein S and APC, I will discuss the potential interaction and mechanisms regulating TFPI cofactor activity of protein S, and I will speculate on the regulation of protein S cofactor activity *in vivo* based on the novel results in this thesis and on recent results from the literature.

### 4.1 <u>Protein S cleavage, multimerisation and its binding to</u> <u>phospholipids</u>

It has been shown that 10-15% of plasma protein S circulates in a cleaved form.<sup>178</sup> The protease responsible for this cleavage *in vivo* is still not known and the effect of cleavage upon APC-dependent and APC-independent cofactor activity of protein S is still under debate.<sup>175-177</sup> It has been suggested that cleavage at both Arg49 and Arg70 affects protein S binding to phospholipids.<sup>176</sup> During this thesis cleavage of protein S was sometimes observed in processed concentrated conditioned media. Because of this, cleavage of protein S was minimised by keeping the time of processing of protein S as short as possible. Because of the issue of cleavage, I preferred using protein S in concentrated conditioned media rather than purified samples. I also aliquoted protein S to avoid extensive freeze thawing. Cleavage of protein S variants was formally assessed by performing SDS-PAGE under reducing conditioned media were all analysed

by Western blot and this confirmed that no cleavage products were present. Purified WT protein S, protein S D95A and protein S Face2 were analysed by silver staining to make sure they were not more cleaved than WT protein S. These precautions suggest that reduced APC cofactor activity of these variants was not a consequence of increased cleavage.

The formation of protein S multimers is well documented in the literature.<sup>189-192</sup> It is not yet clear what causes multimer formation or how to prevent it. While some reports suggest that protein S multimers already exists in the blood,<sup>189,190</sup> others claim it is a consequence of *in vitro* manipulation.<sup>191,192</sup> It has been hypothesised that it is a result of 1) purification, 2) removal of  $Ca^{2+}$  ions, 3) concentration, or 4) spontaneously occurs over time. The effect of multimers on the ability of protein S to bind to phospholipids and to act as a cofactor for APC or TFPI is still being debated.<sup>189-192</sup> It has been shown that multimeric protein S is able to bind to phospholipids with a higher affinity than monomeric protein S.<sup>191</sup> APC- and TFPIindependent activities of purified multimeric protein S due to competition for phospholipids have also been documented at limitina amounts of phospholipids.191,192,216,217

Multimer formation was analysed during my work using Western blot of protein S from native gels. Protein S in concentrated conditioned media and plasma protein S both appeared monomeric. Partial purification on the anion exchange QFF column by FPLC resulted in protein S preparations that were largely monomeric with trace amounts of dimers being present. Purification by immunoaffinity chromatography or by barium citrate precipitation of protein S, however, definitely lead to multimer formation. Commercial plasma purified protein S from Enzyme Research Laboratories also appeared multimeric using this technique.

Accordingly, functional assays to assess protein S cofactor activity towards APC and TFPI were carried out both with monomeric protein S in concentrated conditioned media (where assay conditions permitted this) as well as with multimeric purified protein S. Saturating amounts of phospholipids are also used in all assays.

Protein S is a plasma protein whose functionality is dependent upon its ability to bind to phospholipid surfaces through its Gla domain. Binding to phospholipids requires post transcriptional  $\gamma$ -carboxylation of the glutamic acid residues in the Gla domain of protein S by the vitamin K-dependent  $\gamma$ -carboxylase.<sup>171</sup> Variations

in the efficiency of y-carboxylation of protein S variants secreted from HEK293 cells can therefore potentially influence the cofactor activity of protein S measured in the functional assays. Variants of interest in concentrated conditioned media were analysed by Western blot and protein S was detected either by a polyclonal antibody or an antibody recognising only y-carboxylated protein S. No difference in band intensity with the latter antibodies were observed between WT protein S and protein S D95A, suggesting both preparations contained a similar amount of y-carboxylated protein S. y-carboxylation was also controlled for by purifying WT protein S and the variants of interest by methods that selectively removes non ycarboxylated protein S. Lastly, protein S was assessed for its ability to bind to phospholipids in a plate binding assay. My recombinant barium citrate precipitated WT protein S bound to phospholipids with a  $K_{d(app)}$  similar to that reported in the literature.<sup>182,223-225</sup> Importantly, my colleague Dr Josefin Ahnström also confirmed that there was no difference in binding to phospholipid between partially and fully purified protein S (results not shown). Protein S D95A was assessed for its binding to phospholipid surfaces and there was no significant difference in K<sub>d(app)</sub> compared to WT protein S. While not formally assessed in this thesis, when protein S in concentrated conditioned media was evaluated for its ability to bind phospholipid surfaces, a much lower affinity was observed.

In summary, my results are compatible with those published studies showing that monomeric and multimeric protein S have different affinities for phospholipids, that protein S circulates in plasma as a monomer and that multimerisation is a result of *in vitro* manipulation.<sup>191,192</sup>

### 4.2 <u>APC cofactor activity of protein S; interaction sites and a</u> <u>suggested mechanism</u>

Protein S has been suggested to act as a cofactor for APC in three distinct ways; by increasing the affinity of APC to phospholipid surfaces, by relocating the active site of APC and by overcoming the protective effect of FVa following binding to FXa.<sup>127,137,221,228,229</sup>

Protein S circulates in plasma as a monomer and consequently its affinity for phospholipids is reduced compared to that reported in the literature for multimeric

protein S. The reported affinity of monomeric protein S for phospholipids  $(K_{d(app)} = 250 \text{ nM})^{191}$  is, however, higher than that of APC  $(2-7 \mu \text{M})^{226,227}$  suggesting protein S enhancement of APC binding to phospholipids might have a physiological relevance.

It is well accepted that protein S enhances mainly the APC mediated cleavage at Arg306 of FVa. It has been shown that binding of protein S to APC results in relocalisation of the active site of APC closer to the membrane surface.<sup>228,229,266</sup> This relocalisation was proposed to be the mechanism behind protein S induced preferential cleavage at Arg306. Binding of protein S might cause a major conformational change in APC, resulting in the relocalisation of its active site. A recent paper using a FVa variant R306Q/R506Q/R679Q, identified additional APC cleavage sites in FVa (in close proximity to Arg306 or Arg506). Protein S equally enhanced the cleavage of all these residues (Lys309, Arg313, Arg316, Arg317 and Arg505).<sup>130</sup> While this does not contradict protein S mediated relocalisation of the active site of APC, it argues against the suggestion that this relocalisation is how protein S changes the preferential cleavage site of APC from Arg506 to Arg306. Tran et al. suggest their results argue against the hypothesis that Arg306 and Arg506 are located at different heights above the phospholipid surface. Relocalisation of the active site of APC closer to the membrane surface has also been observed following construction of APC chimeras containing the Gla domain of either FX or prothrombin.<sup>228,267</sup> In contrast to WT APC, protein S was not able to enhance the APC anticoagulant activity of these chimeras.

Because cleavage at Arg506 of FVa by APC does not require phospholipids, the affinity between the two proteins may be sufficiently high for them to bind to each other in the absence of membrane surfaces. It is possible that following cleavage at Arg506 the affinity of APC for FVa decreases. It has been shown that protein S binds with high affinity ( $K_d \sim 40$  nM) to the 493-506 peptide of FV.<sup>116</sup> Consequently, protein S could also act as an APC cofactor by stabilising the APC/FVa complex on the phospholipid membrane (through binding to both APC and FVa), thereby allowing cleavage at Arg306.

The third mechanism by which protein S has been reported to enhance APC anticoagulant activity is by overcoming the protective effect of FXa on FVa cleavage.<sup>219</sup> One proposed hypothesis is that this is due to competitive binding of protein S and FXa for FVa.<sup>116</sup> If a competitive binding between protein S and FXa

for FVa exsists, I believe it is unlikely to play a physiological role in vivo, considering that binding of protein S to FVa is likely reduced once FVa is incorporated in the prothrombinase complex with its physiological substrate prothrombin. Prothrombin (1.39 µM) is also available in much higher concentrations than protein S (free protein S ~145 nM, total protein S ~362 nM). The alternative mechanism proposed was that protein S enhanced APC mediated cleavage of free FVa, thereby potentially affecting the equilibrium between free FXa and FVa, and the FXa/FVa complex.<sup>137</sup> This would cause FVa to dissociate from the complex, consequently being more susceptible to cleavage by APC. This proposed mechanism seems to me more likely to occur *in vivo*, as protein S is known to enhance APC mediated cleavage of FVa. In addition, considering that the plasma concentration of FV (~30 nM) is less than that of FX (~136 nM), the available FVa is likely to influence the equilibrium between free FVa and FVa bound to FXa. It has also been proposed that FXa only protects Arg506 cleavage and that protein S enhances APC mediated cleavage at Arg306 even when FVa is bound to FXa.<sup>138</sup> This proposed mechanism could occur *in vivo*, however, it is likely that it is reduced once the prothrombinase complex binds to its substrate prothrombin.

Three residues in the Gla domain of APC, Asp36/Leu38/Ala39, have been reported to be essential for cofactor activity of protein S.<sup>90</sup> When point substitutions of these residues in APC were analysed, Leu38 was shown to be the most critical residue for protein S cofactor activity.<sup>268</sup> One study used Fab fragments against APC and suggested the EGF1 domain of APC was important for protein S cofactor activity.<sup>91</sup> During this thesis, I identified Asp95, Asp78 and Gln79 in the EGF1 domain of protein S and my colleague Dr Josefin Ahnström identified Gla36 in the Gla domain of protein S as residues essential for APC cofactor activity. When residues Asn33, Pro35 and Tyr39 in protein S were substituted and evaluated by my colleague Dr Josefin Ahnström in the CAT assay, a reduced APC cofactor activity was also observed. This suggests amino acids in close proximity to Gla36 in protein S are also likely to contribute either directly or indirectly to APC cofactor function of protein S. I aligned the Gla and EGF1 domains of protein S and protein C (Figure 4.1A). Interestingly, residues of APC shown to be important for enhancement of its anticoagulant activity by protein S align with the residues of protein S identified as crucial for APC cofactor
activity. Similarly, when I highlighted these residues in the respective models of the proteins, they are predicted to be located at a similar distance from the Nterminal residues comprising the  $\omega$ -loop involved in binding to the phospholipid membrane (Figure 4.1, B and C). It is likely that the pronounced APC cofactor activity of protein S depends on the formation of a functional complex between the two proteins. Protein interactions are usually mediated through multiple residue reciprocal contacts and can involve distinct interaction sites. However, it is unlikely, because of charge considerations, that the identified residues of negative charge in the respective domains (Asp95 and Gla36 in protein S and Asp71 and Asp36 in APC) interact directly with each other. The possibility that Leu38 of APC in conjunction with surrounding residues (such as Asp36 and Ala39), interact with a cluster of residues in protein S composed in part by residues such as Asn33, Pro35, Gla36 and Tyr39 (see arrow Figure 4.1) needs to be further investigated. While a monoclonal antibody directed against the EGF1 of APC, with Asp71 being part of its epitope, inhibited protein S enhancement of the anticoagulant activity of APC,<sup>91</sup> Asp71 is not necessarily involved in an interaction with protein S. However, it is interesting that Asp71 in APC coordinates a calcium ion<sup>91</sup> and that Asp95 in protein S is part of a partially conserved calcium binding site. It is possible that the calcium ion in EGF1 of protein C masks part of the negative charges in this region. It is difficult for me to predict how these residues (Asp95 in the EGF1 domain of protein S and Asp71 in the EGF1 domain of APC) might interact, and I suggest that further studies on these EGF domains of APC and protein S are required before a plausible model can be proposed. Nevertheless, it is worth noting that at least two distinct domains, Gla and EGF1, have been implicated in both APC and protein S mediated APC cofactor activity. Co-operative multidomain interactions could form the basis of further investigation.

To further assess protein S interaction with APC and relocation of the active site of APC by protein S, my colleague Dr Josefin Ahnström is planning to perform FRET analysis with WT protein S, protein S D95A, protein S E36A, WT APC and APC D36A/L38D/A39V.

# A

PROTEIN	S	GLA	DOMAN
PROTEIN	С	GLA	DOMAN

PROTEIN S EGF1 DOMAN PROTEIN C EGF1 DOMAN IPDQCSPLPCNEDGYMSCKDGKASFTCTCKPGWQGEKCEF LEHPCASLCCGHG---TCIDGIGSFSCDCRSGWEGRFCQR : . \*:.\* \*... :\* \*\* .\*\*:\* \*:.\*\*:\*. \*:



Figure 4.1 Location of residues in protein S and protein C shown to be important for APC cofactor activity of protein S.

The alignment of the Gla domains and EGF1 domains of protein S and protein C is shown in A. The model of protein  $S^{193}$  (B) and APC<sup>88</sup> (C) show the Gla domains in yellow, the TSR domain of protein S in light green and the EGF1 domains in cyan. Residues found to be important for APC cofactor activity of protein S are shown as red sticks. Arrows indicate potential interaction sites between protein S and APC.

# 4.3 <u>TFPI cofactor activity of protein S; possible interaction sites</u> and mechanism

Hackeng *et al.* showed that the anticoagulant activity of truncated TFPI, lacking its K3 domain and its C-terminal tail, was not enhanced by protein S.<sup>81</sup> In our lab, we have shown that protein S is able to enhance TFPI lacking its C-terminal tail to the same extent as full-length TFPI (unpublished results of my colleague Verity Hockey) suggesting the C-terminal tail is not directly involved in the interaction with protein S. Recently, it was demonstrated that the K3 domain of TFPI is essential for protein S cofactor activity.<sup>238</sup> Within the K3 domain the P1 residue,

Arg199, was reported to play an important role.<sup>238</sup> Direct binding of protein S to the K3 domain was also shown.<sup>238</sup> Verity Hockey has constructed composite variants of charged residues in the K3 domain of full-length TFPI. These were shown inhibit FXa to the same extent as WT TFPI, however, the cofactor activity of protein S for E226Q/E234Q/R237Q and D194Q/R195Q/R199Q was reduced approximately 7 and 3-fold in comparison with WT TFPI (unpublished results). The TFPI variant K213Q/R215Q/K232Q was enhanced normally by protein S (unpublished results).

Protein S variants constructed during my thesis have failed to clearly identify residues in protein S critical for TFPI cofactor activity. However, I did identify a potentially interesting variant in the EGF1 domain, protein S NEDM, that needs to be further investigated. Considering the pronounced TFPI cofactor activity of protein S and the low concentrations of plasma full-length TFPI (~0.25 nM), their binding affinity in vivo must be much higher than those derived from surface plasmon resonance (SPR) experiments (K<sub>d</sub>> 150 nM).<sup>238,241</sup> Binding experiments by blotting and SPR were performed in the absence of phospholipids. It is possible that the presence of phospholipids would increase the affinity of TFPI for protein S. Interestingly, co-immunoprecipitation studies have suggested fulllength TFPI and free protein S do exist in plasma as a complex.<sup>241</sup> If protein S and TFPI interact *in vivo*, they are likely to do so through multiple sites, potentially explaining why substitution of single residues within protein S failed to completely disrupt protein S enhancement of TFPI. Alternatively, the interaction site on protein S for TFPI might be located in domains that were not evaluated during this thesis.

The mechanism by which protein S enhances TFPI is not yet completely understood. It is possible that, as for APC, protein S enhances the affinity of TFPI for negatively charged phospholipid surfaces. Protein S could also, by binding both TFPI and FXa, co-localise the proteins on the phospholipid surface and increase their affinity for each other by forming a trivalent complex. Alternatively, protein S could induce a conformational change and thereby facilitate TFPI binding and inhibition of FXa. I believe it is unlikely that protein S, by binding to the K3 domain of TFPI, induces a conformational change in the K2 domain of TFPI as the Kunitz domains of TFPI are connected through quite long (30 amino acid) linker regions. It is more plausible that protein S induces a conformational change in FXa, as it has been suggested to do for APC. Interestingly, FXa and APC have a very similar domain structure and protein S is reported to bind FXa with a high affinity (~15 nM).<sup>214</sup> Conceivably, protein S binding to FXa induces a conformational change in its active site causing it to be more susceptible to inhibition by the K2 domain of TFPI.

While recent publications suggest a direct interaction of protein S with TFPI and possibly the K3 domain, the reported affinities are surprisingly low (K<sub>d</sub> >150 nM) in comparison to those reported for protein S binding to FXa (K<sub>d</sub> ~15 nM).<sup>238,241</sup> The interaction of the K3 domain of TFPI with protein S would represent an unusual interaction since Kunitz domains usually bind in active site clefts. It has been shown that FXa is able to cleave the K3 domain of TFPI after its P1 residue Arg199 in vitro.<sup>71</sup> While the physiological importance of this is questionable, it suggests that under certain conditions the K3 domain of TFPI is able to bind the active site of FXa. While not supported by current binding studies between protein S and TFPI, the possibility that protein S induces a conformational change in FXa that favours a stronger and quicker inhibition of FXa by the K3 domain of TFPI rather than a weaker and slower inhibition by the K2 domain of TFPI, needs to be examined further. If this is the case, the deletion or substitutions in the K3 domain of TFPI would result in loss of protein S cofactor activity, not because of lack of binding to protein S but because of lack of binding to the active site of FXa. This hypothesis could easily be tested by evaluating FXa inhibition by synthetic K2 and K3 domains in the presence and absence of protein S.

## 4.4 Protein S direct activity

It was initially thought that protein S could directly inhibit the prothrombinase complex by direct binding to FVa and FXa. This APC-independent anticoagulant activity of protein S was termed "protein S direct". Later, it was shown that this activity of protein S in purified systems was due to competition of protein S multimers for limiting amounts of phospholipids.<sup>191,192,216,217</sup> However, an APC-independent activity of protein S was observed in plasma also at saturating amounts of phospholipids.<sup>218</sup> This was identified as protein S cofactor activity towards TFPI in 2006.<sup>81</sup> However, Heeb and colleagues have observed an APC-and TFPI-independent activity of protein S also at saturating amounts of

phospholipids.<sup>189,190,242</sup> They claim that protein S can directly inhibit FXa and TF/FVIIa through direct binding. That protein S direct has not been observed in many other laboratories has been potentially explained by Heeb and colleagues by showing that purification of protein S often lead to loss of binding to a zinc ion, which correlates with loss of protein S direct activity.<sup>242</sup> The low IC<sub>50</sub> reported (~120 nM)<sup>190</sup> for protein S inhibition of the prothrombinase complex contrasts with the huge cofactor activity observed for APC and TFPI suggesting that while protein S direct might in some instances be evident *in vitro*, it has little *in vivo* relevance.

## 4.5 In vivo regulation of coagulation by protein S

Protein S plays a central role in regulating haemostasis and its importance is underpinned by the severe outcome of patients with protein S deficiencies, with homozygous deficiencies generally being incompatible with life.<sup>201-203</sup>

The recent finding that protein S acts as a cofactor not only for APC but also for TFPI raises the guestion whether APC and TFPI compete for binding to protein S in vivo. While it is yet too early to give a definitive answer, a first indication is given by the results obtained in this thesis. Both protein S Asp95 and protein S Gla36 had almost no APC cofactor activity but were still able to enhance the anticoagulant activity of TFPI. While this does not rule out a competition between APC and TFPI for protein S, it suggests that APC and TFPI interact with distinct residues of protein S. It has been shown that addition of APC to plasma increases the anticoagulant activity of TFPI/ protein S.<sup>241</sup> The reason behind this is probably twofold. Firstly, APC reduces the procoagulant response and delays the start of massive thrombin generation. This allows time for the slow binding and inhibition of FXa by TFPI to occur. Secondly, inactivation of FVIIIa and FVa by APC results in a reduced rate of activation of FX and could result in a shift in the equilibrium from FXa/FVa towards free FXa. This overcomes the protective effect of prothrombin on TFPI inhibition of FXa when FXa is incorporated in the prothrombinase complex.<sup>77</sup> It would therefore, seem that the APC/protein S and the TFPI/protein S anticoagulant activities act in synergy rather than the two pathways competing for protein S. Considering also the excess plasma concentration of protein S (free protein S ~145 nM, total protein S ~362 nM) over that of protein C (~65 nM) and TFPI (full-length ~0.25 nM, total ~2.5 nM) and the different phases of regulation by APC and TFPI (propagation phase and initiation phase, respectively), I believe that it is unlikely that these two proteins compete for binding to protein S *in vivo*.

While in vitro assays represent valuable and powerful tools to analyse the biochemical interactions and processes that occur in vivo they inevitably exclude many of the variables present in vivo. In my plasma based assay, performed by CAT, phospholipids were used to provide a surface for coagulation factors on which to interact. While this simplifies the assay and decreases inter-assay variability, it represents a simplified model for platelets, platelet activation and its components such as platelet protein S, TFPI and FV. The procoagulant and anticoagulant activity of coagulation factors also seem to be differently regulated by the membrane composition and distribution.<sup>125,269</sup> In addition, anticoagulant coagulation factors seem more susceptible to plasma dilution than procoagulant factors.<sup>264</sup> Cell-free plasma based assays differs from *in vivo* assays as they lead to the formation of coagulation in suspension rather than on a surface. In my assays I have used 1 pM TF. However, it is unclear how this relates to the local concentrations of TF in vivo. The concentration of TF is also likely to vary during the course of the haemostatic response *in vivo*, following platelet plug formation above the subendothelial cells. A limiting factor of the *in vitro* plasma assays is that it inevitably leads to depletion of coagulation factors during its course. This is likely to be of particular importance for those factors that are present in small amounts such as TFPI. Based upon the vast amount of results derived from in vitro assays and those from in vivo assays it is, however, reasonable to speculate how coagulation is likely to occur in vivo.

Following vessel wall injury, the ECM and TF are exposed to the blood.<sup>10</sup> VWF binds to exposed collagen in the ECM allowing platelet adherance, aggregation and activation.<sup>1,5</sup> This leads to exposure of phosphatidylserine on the outer leflet of the platelet membrane which provides a surface for the binding of coagulation factors.<sup>9</sup> Simultaneusly, as the primary platelet plug is forming, TF triggers coagulation through binding to and activation of FVIIa. TF/FVIIa activates FIX and FX. TFPI is the main regulator of the initiation of coagulation as it inhibits FXa and, in a FXa dependent manner, inhibits TF/FVIIa on the subendothelial cells.<sup>66-</sup>

In the presence of protein S, the K<sub>i</sub> for TFPI inhibition of FXa falls within the physiological plasma concentration of TFPI, allowing the TFPI pathway to play a potentially major anticoagulant role. While protein S only enhances free TFPI containing its K3 domain,<sup>81</sup> truncated forms of TFPI have been shown to be able to inhibit FXa and TF/FVIIa.78-80 These free truncated TFPI variants in plasma (~5% of plasma TFPI) are unlikely to play a major role in regulating initiation of coagulation as they represent half the amount of free full-length plasma TFPI (~10% of plasma TFPI) and are not enhanced by protein S. They could, however, have a role in regulating the haemostatic balance under condition in which no tissue damage occurs. The majority (~80%) of plasma TFPI is, however, bound to lipoproteins and their anticoagulant role is still not clear.<sup>64,65</sup> Cell suface bound TFPI (~80% of total TFPI) has been shown to have some anticoagulant activity.<sup>43</sup> Considering vascular smooth muscle cells express TFPI in addition to TF, this pool of TFPI is likely to regulate the local initiation of coagulation on subendothelial cells exposed to the blood. TFPI on intact endothelial cells surrounding the tissue damage might also play a significant role in regulating coagulation by inhibiting FXa that escapes the site of vascular injury, therefore by confining the haemostatic plug. It is still not clear if protein S is able to act as a cofactor for endothelial bound TFPI. Free full-length TFPI circulates in plasma in low concentrations (~0.25 nM). This does exceed the predicted amount of TF (pM range) that triggers coagulation. However, it is far lower than the amount of FXa predicted to be generated from FX (total plasma concentration of FX ~136 nM) potentially explaining why TFPI is only able to inhibit initiation of coagulation. FXa that escapes inhibition by TFPI/protein S generates trace amount of thrombin. Thrombin produces a feedback mechanism by activating FVIII and FV.<sup>12</sup> This leads to massive generation of FXa which can no longer be significantly inhibited by TFPI. The feedback mechanism of thrombin finally results in a massive amplification of itself and is usually referred to as propagation of the coagulation cascade. Thrombin that escapes the immediate site of injury binds to TM on the endothelial cells surrounding the vessel damage. This leads to activation of protein C which acts as a potent anticoagulant together with its cofactor protein S.<sup>114</sup> Results in my thesis and previous reports show that the APC anticoagulant activity is highly dependent upon its cofactor protein S.<sup>218</sup> As protein C is activated at the edge of the vessel injury it is likely to play a role in spatially

confining the haemostatic plug. APC/protein S are also able to bind to the activated platelets through their Gla domains and inactivate FVa and FVIIIa on the growing haemostatic plug. Following the initial formation of the loose platelet plug over the subendothelial cells, it has been suggested that soluble TF or TF in microparticles continue to trigger coagulation on the surface of the growing platelet plug.<sup>16-18</sup> While this requires further investigation, it is seems plausable that TFPI secreted by platelets is an important regulator of this source of TF. As TFPI in platelets exists in its full-length form it is likely that even the activity of platelet TFPI is dependent upon its cofactor protein S.<sup>63</sup>

To conclude, protein S is crucial in regulating haemostasis. It does so by substancially enhancing the enzymatic activity of both TFPI and APC, thereby regulating both initiation and propagation of the coagulation cascade. Further work is necessary to unravel the complex molecular mechanism behind protein S function including, elaboration of the potential molecular interactions between residues found to be crucial in APC and protein S, the interacting residues between protein S and TFPI, and the molecular mechanism governing protein S enhancement of the anticoagulant activity of TFPI.

## APPENDIX

## **Sequencing primers**

	Nucleotide sequence	Area of protein S covered
T7 primer	5' taatacgactcactataggg 3'	ATG-middle EGF4
Nt456 primer	5' tattcactgctgcacgtcag 3'	TSR- beginning SHBG
Nt769 primer	5' gatgtggatgaatgctctttg 3'	EGF3- middle SHBG
BGH reverse	5' tagaaggcacagtcgagg 3'	SHBG-stop

## **Mutagenesis primers**

All mutants were made by using the pcDNA6/protein S vector containing the WT protein S cDNA sequence, with the exception of NNEDM which was made by introducing the N74A substitution in the vector containing the NEDM substitutions and KTK which was made by introducing the T103A/K105A substitution in the vector already containing the K94A mutation (made by Dr S.M. Rezende). Please note that sequencing primers designed to introduce subsequent mutations in a region already mutated are designed to be complementary to the mutated template cDNA (with the exception of the new mutation to introduce) and not to the WT protein S template. All sequencing primers are from 5' to 3'. The name of each sequencing primer corresponds to the amino acid mutated, its numerical location in the mature protein S, followed by the residue into which it was mutated. f; foreward, r; reverse.

Gla1 composite mutants				
fD38N	cccggaaacgaattatttttatcc			
rD38N	ggataaaaataattcgtttccggg			
fV46G	ccaaaatacttaggttgtcttcgctc			
rV46G	gagcgaagacaacctaagtattttgg			
fK97Q	gcaaagatggacaagcttcttttacttgcac			
rK97Q	gtgcaagtaaaagaagcttgtccatctttgc			

## TSR composite mutant

fR49Q	ccaaaatacttagtttgtcttcagtcttttcaaactgggttattcac
rR49Q	gtgaataacccagtttgaaaagactgaagacaaactaagtattttgg
fR60Q	tgggttattcactgctgcacagcagtcaactaatgcttatcc
rR60Q	ggataagcattagttgactgctgtgcagcagtgaataaccca
fD68N R70Q	cagtcaactaatgcttatcctaacctacaaagctgtgtcaatgcc
rD68N R70Q	ggcattgacacagctttgtaggttaggataagcattagttgactg
fD78N	gctgtgtcaatgccattccaaaccagtgtagtcct
rD78N	aggactacactggtttggaatggcattgacacagc

## **NEDM** composite mutant

fN86A E87Q D88N M91AcctctgccatgcgctcaaaatggatatgcgagctgcaaagrN86A E87Q D88N M91Actttgcagctcgcatatccattttgagcgcatggcagagg

## KTK composite mutant

fT103A K105A	gcttcttttacttgcgcttgtgcaccaggttggcaaggag
rT103A K105A	ctccttgccaacctggtgcacaagcgcaagtaaaagaagc

Point mutants	S
fN74A	cctgacctaagaagctgtgtcgctgccattccagaccagtg
rN74A	cactggtctggaatggcagcgacacagcttcttaggtcagg
fD78A	gtcaatgccattccagcccagtgtagtcctctgccatgc
rD78A	gcatggcagaggactacactgggctggaatggcattgac
fQ79A	gtgtcaatgccattccagacgcgtgtagtcctctgccatgc
rQ79A	gcatggcagaggactacacgcgtctggaatggcattgacac
fD95N	ggatatatgagctgcaaaaatggaaaagcttcttttacttg
rD95N	cctatatactcgacgtttttaccttttcgaagaaaatgaac

## REFERENCES

1. Varga-Szabo D, Pleines I, Nieswandt B. Cell adhesion mechanisms in platelets. *Arterioscler Thromb Vasc Biol.* 2008;28(3):403-412.

2. Jeanneau C, Avner P, Sultan Y. Use of monoclonal antibody and colloidal gold in E.M. localization of von Willebrand factor in megakaryocytes and platelets. *Cell Biol Int Rep.* 1984;8(10):841-848.

3. Wagner DD, Marder VJ. Biosynthesis of von Willebrand protein by human endothelial cells: processing steps and their intracellular localization. *J Cell Biol.* 1984;99(6):2123-2130.

4. Zucker MB, Broekman MJ, Kaplan KL. Factor VIII-related antigen in human blood platelets: localization and release by thrombin and collagen. *J Lab Clin Med.* 1979;94(5):675-682.

5. Schneider SW, Nuschele S, Wixforth A, et al. Shear-induced unfolding triggers adhesion of von Willebrand factor fibers. *Proc Natl Acad Sci U S A*. 2007;104(19):7899-7903.

6. Yago T, Lou J, Wu T, et al. Platelet glycoprotein Ibalpha forms catch bonds with human WT vWF but not with type 2B von Willebrand disease vWF. *J Clin Invest.* 2008;118(9):3195-3207.

7. Angiolillo DJ, Ueno M, Goto S. Basic principles of platelet biology and clinical implications. *Circ J*. 2010;74(4):597-607.

8. Reininger AJ. Function of von Willebrand factor in haemostasis and thrombosis. *Haemophilia*. 2008;14 Suppl 511-26.

9. Lentz BR. Exposure of platelet membrane phosphatidylserine regulates blood coagulation. *Prog Lipid Res*. 2003;42(5):423-438.

10. Norris LA. Blood coagulation. *Best Pract Res Clin Obstet Gynaecol.* 2003;17(3):369-383.

11. Morrissey JH, Macik BG, Neuenschwander PF, Comp PC. Quantitation of activated factor VII levels in plasma using a tissue factor mutant selectively deficient in promoting factor VII activation. *Blood.* 1993;81(3):734-744.

12. Esmon CT, Lollar P. Involvement of thrombin anion-binding exosites 1 and 2 in the activation of factor V and factor VIII. *J Biol Chem.* 1996;271(23):13882-13887.

13. Nesheim ME, Taswell JB, Mann KG. The contribution of bovine Factor V and Factor Va to the activity of prothrombinase. *J Biol Chem.* 1979;254(21):10952-10962.

14. Rosing J, Tans G, Govers-Riemslag JW, Zwaal RF, Hemker HC. The role of phospholipids and factor Va in the prothrombinase complex. *J Biol Chem.* 1980;255(1):274-283.

15. van Rijn JL, Govers-Riemslag JW, Zwaal RF, Rosing J. Kinetic studies of prothrombin activation: effect of factor Va and phospholipids on the formation of the enzyme-substrate complex. *Biochemistry*. 1984;23(20):4557-4564.

16. Owens AP, 3rd, Mackman N. Tissue factor and thrombosis: The clot starts here. *Thromb Haemost.* 2010;104(3):432-439.

17. Chou J, Mackman N, Merrill-Skoloff G, et al. Hematopoietic cell-derived microparticle tissue factor contributes to fibrin formation during thrombus propagation. *Blood.* 2004;104(10):3190-3197.

18. Bogdanov VY, Balasubramanian V, Hathcock J, et al. Alternatively spliced human tissue factor: a circulating, soluble, thrombogenic protein. *Nat Med.* 2003;9(4):458-462.

19. Huang M, Rigby AC, Morelli X, et al. Structural basis of membrane binding by Gla domains of vitamin K-dependent proteins. *Nat Struct Biol.* 2003;10(9):751-756.

20. Adams TE, Hockin MF, Mann KG, Everse SJ. The crystal structure of activated protein C-inactivated bovine factor Va: Implications for cofactor function. *Proc Natl Acad Sci U S A*. 2004;101(24):8918-8923.

21. Majumder R, Quinn-Allen MA, Kane WH, Lentz BR. A phosphatidylserine binding site in factor Va C1 domain regulates both assembly and activity of the prothrombinase complex. *Blood*. 2008;112(7):2795-2802.

22. Takeshima K, Smith C, Tait J, Fujikawa K. The preparation and phospholipid binding property of the C2 domain of human factor VIII. *Thromb Haemost.* 2003;89(5):788-794.

23. Muszbek L, Bagoly Z, Bereczky Z, Katona E. The involvement of blood coagulation factor XIII in fibrinolysis and thrombosis. *Cardiovasc Hematol Agents Med Chem.* 2008;6(3):190-205.

24. Crawley JT, Zanardelli S, Chion CK, Lane DA. The central role of thrombin in hemostasis. *J Thromb Haemost.* 2007;5 Suppl 195-101.

25. Dahlback B. Blood coagulation and its regulation by anticoagulant pathways: genetic pathogenesis of bleeding and thrombotic diseases. *J Intern Med.* 2005;257(3):209-223.

26. Crawley JT, Lane DA. The haemostatic role of tissue factor pathway inhibitor. *Arterioscler Thromb Vasc Biol.* 2008;28(2):233-242.

27. Esmon CT. The protein C pathway. *Chest.* 2003;124(3 Suppl):26S-32S.

28. Conard J, Brosstad F, Lie Larsen M, Samama M, Abildgaard U. Molar antithrombin concentration in normal human plasma. *Haemostasis*. 1983;13(6):363-368.

29. Rau JC, Beaulieu LM, Huntington JA, Church FC. Serpins in thrombosis, hemostasis and fibrinolysis. *J Thromb Haemost*. 2007;5 Suppl 1102-115.

30. Whisstock JC, Pike RN, Jin L, et al. Conformational changes in serpins: II. The mechanism of activation of antithrombin by heparindagger. *J Mol Biol*. 2000;301(5):1287-1305.

31. Huntington JA. Mechanisms of glycosaminoglycan activation of the serpins in hemostasis. *J Thromb Haemost*. 2003;1(7):1535-1549.

32. Quinsey NS, Greedy AL, Bottomley SP, Whisstock JC, Pike RN. Antithrombin: in control of coagulation. *Int J Biochem Cell Biol*. 2004;36(3):386-389.

33. Rijken DC, Lijnen HR. New insights into the molecular mechanisms of the fibrinolytic system. *J Thromb Haemost.* 2009;7(1):4-13.

34. Nicholl SM, Roztocil E, Davies MG. Plasminogen activator system and vascular disease. *Curr Vasc Pharmacol*. 2006;4(2):101-116.

35. Wang JW, Eikenboom J. Von Willebrand disease and Weibel-Palade bodies. *Hamostaseologie*. 2010;30(3):150-155.

36. Bolton-Maggs PH. Optimal haemophilia care versus the reality. *Br J Haematol.* 2006;132(6):671-682.

37. Dahlback B. Advances in understanding pathogenic mechanisms of thrombophilic disorders. *Blood.* 2008;112(1):19-27.

38. Bajaj MS, Kuppuswamy MN, Saito H, Spitzer SG, Bajaj SP. Cultured normal human hepatocytes do not synthesize lipoprotein-associated coagulation inhibitor: evidence that endothelium is the principal site of its synthesis. *Proc Natl Acad Sci U S A*. 1990;87(22):8869-8873.

39. Crawley J, Lupu F, Westmuckett AD, et al. Expression, localization, and activity of tissue factor pathway inhibitor in normal and atherosclerotic human vessels. *Arterioscler Thromb Vasc Biol.* 2000;20(5):1362-1373.

40. Sandset PM, Abildgaard U, Larsen ML. Heparin induces release of extrinsic coagulation pathway inhibitor (EPI). *Thromb Res.* 1988;50(6):803-813.

41. Girard TJ, Eddy R, Wesselschmidt RL, et al. Structure of the human lipoproteinassociated coagulation inhibitor gene. Intro/exon gene organization and localization of the gene to chromosome 2. *J Biol Chem.* 1991;266(8):5036-5041.

42. Zhang J, Piro O, Lu L, Broze GJ, Jr. Glycosyl phosphatidylinositol anchorage of tissue factor pathway inhibitor. *Circulation*. 2003;108(5):623-627.

43. Piro O, Broze GJ, Jr. Comparison of cell-surface TFPIalpha and beta. *J Thromb Haemost*. 2005;3(12):2677-2683.

44. Wun TC, Kretzmer KK, Girard TJ, Miletich JP, Broze GJ, Jr. Cloning and characterization of a cDNA coding for the lipoprotein-associated coagulation inhibitor shows that it consists of three tandem Kunitz-type inhibitory domains. *J Biol Chem.* 1988;263(13):6001-6004.

45. Novotny WF, Girard TJ, Miletich JP, Broze GJ, Jr. Purification and characterization of the lipoprotein-associated coagulation inhibitor from human plasma. *J Biol Chem.* 1989;264(31):18832-18837.

46. Belaaouaj AA, Li A, Wun TC, Welgus HG, Shapiro SD. Matrix metalloproteinases cleave tissue factor pathway inhibitor. Effects on coagulation. *J Biol Chem.* 2000;275(35):27123-27128.

47. Li A, Wun TC. Proteolysis of tissue factor pathway inhibitor (TFPI) by plasmin: effect on TFPI activity. *Thromb Haemost*. 1998;80(3):423-427.

48. Ohkura N, Enjyoji K, Kamikubo Y, Kato H. A novel degradation pathway of tissue factor pathway inhibitor: incorporation into fibrin clot and degradation by thrombin. *Blood.* 1997;90(5):1883-1892.

49. Steppich BA, Seitz I, Busch G, Stein A, Ott I. Modulation of tissue factor and tissue factor pathway inhibitor-1 by neutrophil proteases. *Thromb Haemost.* 2008;100(6):1068-1075.

50. Lupu C, Goodwin CA, Westmuckett AD, et al. Tissue factor pathway inhibitor in endothelial cells colocalizes with glycolipid microdomains/caveolae. Regulatory mechanism(s) of the anticoagulant properties of the endothelium. *Arterioscler Thromb Vasc Biol.* 1997;17(11):2964-2974.

51. Ott I, Miyagi Y, Miyazaki K, et al. Reversible regulation of tissue factor-induced coagulation by glycosyl phosphatidylinositol-anchored tissue factor pathway inhibitor. *Arterioscler Thromb Vasc Biol.* 2000;20(3):874-882.

52. Sevinsky JR, Rao LV, Ruf W. Ligand-induced protease receptor translocation into caveolae: a mechanism for regulating cell surface proteolysis of the tissue factor-dependent coagulation pathway. *J Cell Biol.* 1996;133(2):293-304.

53. Mast AE, Acharya N, Malecha MJ, Hall CL, Dietzen DJ. Characterization of the association of tissue factor pathway inhibitor with human placenta. *Arterioscler Thromb Vasc Biol.* 2002;22(12):2099-2104.

54. Maroney SA, Cunningham AC, Ferrel J, et al. A GPI-anchored co-receptor for tissue factor pathway inhibitor controls its intracellular trafficking and cell surface expression. *J Thromb Haemost*. 2006;4(5):1114-1124.

55. Novotny WF, Brown SG, Miletich JP, Rader DJ, Broze GJ, Jr. Plasma antigen levels of the lipoprotein-associated coagulation inhibitor in patient samples. *Blood*. 1991;78(2):387-393.

56. Lupu C, Poulsen E, Roquefeuil S, et al. Cellular effects of heparin on the production and release of tissue factor pathway inhibitor in human endothelial cells in culture. *Arterioscler Thromb Vasc Biol.* 1999;19(9):2251-2262.

57. Gori AM, Pepe G, Attanasio M, et al. Tissue factor reduction and tissue factor pathway inhibitor release after heparin administration. *Thromb Haemost.* 1999;81(4):589-593.

58. Lindahl AK, Jacobsen PB, Sandset PM, Abildgaard U. Tissue factor pathway inhibitor with high anticoagulant activity is increased in post-heparin plasma and in plasma from cancer patients. *Blood Coagul Fibrinolysis*. 1991;2(6):713-721.

59. Novotny WF, Palmier M, Wun TC, Broze GJ, Jr., Miletich JP. Purification and properties of heparin-releasable lipoprotein-associated coagulation inhibitor. *Blood.* 1991;78(2):394-400.

60. Novotny WF, Girard TJ, Miletich JP, Broze GJ, Jr. Platelets secrete a coagulation inhibitor functionally and antigenically similar to the lipoprotein associated coagulation inhibitor. *Blood.* 1988;72(6):2020-2025.

61. Lupu C, Lupu F, Dennehy U, Kakkar VV, Scully MF. Thrombin induces the redistribution and acute release of tissue factor pathway inhibitor from specific granules within human endothelial cells in culture. *Arterioscler Thromb Vasc Biol.* 1995;15(11):2055-2062.

62. Maroney SA, Haberichter SL, Friese P, et al. Active tissue factor pathway inhibitor is expressed on the surface of coated platelets. Vol. 109; 2007:1931-1937.

63. Maroney SA, Mast AE. Expression of tissue factor pathway inhibitor by endothelial cells and platelets. *Transfus Apher Sci.* 2008;38(1):9-14.

64. Hansen JB, Huseby NE, Sandset PM, et al. Tissue-factor pathway inhibitor and lipoproteins. Evidence for association with and regulation by LDL in human plasma. *Arterioscler Thromb.* 1994;14(2):223-229.

65. Broze GJ, Jr., Lange GW, Duffin KL, MacPhail L. Heterogeneity of plasma tissue factor pathway inhibitor. *Blood Coagul Fibrinolysis*. 1994;5(4):551-559.

66. Girard TJ, Warren LA, Novotny WF, et al. Functional significance of the Kunitztype inhibitory domains of lipoprotein-associated coagulation inhibitor. *Nature*. 1989;338(6215):518-520.

67. Sanders NL, Bajaj SP, Zivelin A, Rapaport SI. Inhibition of tissue factor/factor VIIa activity in plasma requires factor X and an additional plasma component. *Blood.* 1985;66(1):204-212.

68. Baugh RJ, Broze GJ, Jr., Krishnaswamy S. Regulation of extrinsic pathway factor Xa formation by tissue factor pathway inhibitor. *J Biol Chem.* 1998;273(8):4378-4386.

69. Callander NS, Rao LV, Nordfang O, et al. Mechanisms of binding of recombinant extrinsic pathway inhibitor (rEPI) to cultured cell surfaces. Evidence that rEPI can bind to and inhibit factor VIIa-tissue factor complexes in the absence of factor Xa. *J Biol Chem.* 1992;267(2):876-882.

70. Jackson CM, Nemerson Y. Blood coagulation. *Annu Rev Biochem*. 1980;49765-811.

71. Salemink I, Franssen J, Willems GM, et al. Factor Xa cleavage of tissue factor pathway inhibitor is associated with loss of anticoagulant activity. *Thromb Haemost.* 1998;80(2):273-280.

72. Huang ZF, Wun TC, Broze GJ, Jr. Kinetics of factor Xa inhibition by tissue factor pathway inhibitor. *J Biol Chem*. 1993;268(36):26950-26955.

73. Mast AE, Broze GJ, Jr. Physiological concentrations of tissue factor pathway inhibitor do not inhibit prothrombinase. *Blood*. 1996;87(5):1845-1850.

74. Piro O, Broze GJ, Jr. Role for the Kunitz-3 domain of tissue factor pathway inhibitor-alpha in cell surface binding. *Circulation*. 2004;110(23):3567-3572.

75. Warshawsky I, Bu G, Mast A, et al. The carboxy terminus of tissue factor pathway inhibitor is required for interacting with hepatoma cells in vitro and in vivo. *J Clin Invest*. 1995;95(4):1773-1781.

76. Valentin S, Schousboe I. Factor Xa enhances the binding of tissue factor pathway inhibitor to acidic phospholipids. *Thromb Haemost.* 1996;75(5):796-800.

77. Franssen J, Salemink I, Willems GM, et al. Prothrombinase is protected from inactivation by tissue factor pathway inhibitor: competition between prothrombin and inhibitor. *Biochem J*. 1997;323 (Pt 1)33-37.

78. Lindhout T, Willems G, Blezer R, Hemker HC. Kinetics of the inhibition of human factor Xa by full-length and truncated recombinant tissue factor pathway inhibitor. *Biochem J.* 1994;297 (Pt 1)131-136.

79. Wesselschmidt R, Likert K, Girard T, Wun TC, Broze GJ, Jr. Tissue factor pathway inhibitor: the carboxy-terminus is required for optimal inhibition of factor Xa. *Blood*. 1992;79(8):2004-2010.

80. Lockett JM, Mast AE. Contribution of regions distal to glycine-160 to the anticoagulant activity of tissue factor pathway inhibitor. *Biochemistry*. 2002;41(15):4989-4997.

81. Hackeng TM, Sere KM, Tans G, Rosing J. Protein S stimulates inhibition of the tissue factor pathway by tissue factor pathway inhibitor. *Proc Natl Acad Sci U S A*. 2006;103(9):3106-3111.

82. Broze GJ, Jr., Warren LA, Novotny WF, et al. The lipoprotein-associated coagulation inhibitor that inhibits the factor VII-tissue factor complex also inhibits factor Xa: insight into its possible mechanism of action. *Blood*. 1988;71(2):335-343.

83. Girard TJ, MacPhail LA, Likert KM, et al. Inhibition of factor VIIa-tissue factor coagulation activity by a hybrid protein. *Science*. 1990;248(4961):1421-1424.

84. Carlsson K, Freskgard PO, Persson E, Carlsson U, Svensson M. Probing the interface between factor Xa and tissue factor in the quaternary complex tissue factor-factor VIIa-factor Xa-tissue factor pathway inhibitor. *Eur J Biochem.* 2003;270(12):2576-2582.

85. Thiec F, Cherel G, Christophe OD. Role of the Gla and first epidermal growth factor-like domains of factor X in the prothrombinase and tissue factor-factor VIIa complexes. *J Biol Chem.* 2003;278(12):10393-10399.

86. Plutzky J, Hoskins JA, Long GL, Crabtree GR. Evolution and organization of the human protein C gene. *Proc Natl Acad Sci U S A*. 1986;83(3):546-550.

87. Beckmann RJ, Schmidt RJ, Santerre RF, et al. The structure and evolution of a 461 amino acid human protein C precursor and its messenger RNA, based upon the DNA sequence of cloned human liver cDNAs. *Nucleic Acids Res.* 1985;13(14):5233-5247.

88. Perera L, Foley C, Darden TA, et al. Modeling zymogen protein C. *Biophys J*. 2000;79(6):2925-2943.

89. Zhang L, Jhingan A, Castellino FJ. Role of individual gamma-carboxyglutamic acid residues of activated human protein C in defining its in vitro anticoagulant activity. *Blood.* 1992;80(4):942-952.

90. Preston RJ, Ajzner E, Razzari C, et al. Multifunctional specificity of the protein C/activated protein C Gla domain. *J Biol Chem.* 2006;281(39):28850-28857.

91. Ohlin AK, Landes G, Bourdon P, et al. Beta-hydroxyaspartic acid in the first epidermal growth factor-like domain of protein C. Its role in Ca2+ binding and biological activity. *J Biol Chem.* 1988;263(35):19240-19248.

92. Grinnell BW, Walls JD, Gerlitz B. Glycosylation of human protein C affects its secretion, processing, functional activities, and activation by thrombin. *J Biol Chem.* 1991;266(15):9778-9785.

93. He X, Rezaie AR. Identification and characterization of the sodium-binding site of activated protein C. *J Biol Chem.* 1999;274(8):4970-4976.

94. Mather T, Oganessyan V, Hof P, et al. The 2.8 A crystal structure of Gladomainless activated protein C. *EMBO J.* 1996;15(24):6822-6831.

95. Mesters RM, Heeb MJ, Griffin JH. A novel exosite in the light chain of human activated protein C essential for interaction with blood coagulation factor Va. *Biochemistry*. 1993;32(47):12656-12663.

96. Mesters RM, Heeb MJ, Griffin JH. Interactions and inhibition of blood coagulation factor Va involving residues 311-325 of activated protein C. *Protein Sci.* 1993;2(9):1482-1489.

97. Mesters RM, Houghten RA, Griffin JH. Identification of a sequence of human activated protein C (residues 390-404) essential for its anticoagulant activity. *J Biol Chem.* 1991;266(36):24514-24519.

98. Mosnier LO, Gale AJ, Yegneswaran S, Griffin JH. Activated protein C variants with normal cytoprotective but reduced anticoagulant activity. *Blood*. 2004;104(6):1740-1744.

99. Sadler JE. Thrombomodulin structure and function. *Thromb Haemost*. 1997;78(1):392-395.

100. Fuentes-Prior P, Iwanaga Y, Huber R, et al. Structural basis for the anticoagulant activity of the thrombin-thrombomodulin complex. *Nature*. 2000;404(6777):518-525.

101. Yang L, Rezaie AR. The fourth epidermal growth factor-like domain of thrombomodulin interacts with the basic exosite of protein C. *J Biol Chem.* 2003;278(12):10484-10490.

102. Liu LW, Rezaie AR, Carson CW, Esmon NL, Esmon CT. Occupancy of anion binding exosite 2 on thrombin determines Ca2+ dependence of protein C activation. *J Biol Chem.* 1994;269(16):11807-11812.

103. Bukys MA, Orban T, Kim PY, et al. The structural integrity of anion binding exosite I of thrombin is required and sufficient for timely cleavage and activation of factor V and factor VIII. *J Biol Chem.* 2006;281(27):18569-18580.

104. Dharmawardana KR, Olson ST, Bock PE. Role of regulatory exosite I in binding of thrombin to human factor V, factor Va, factor Va subunits, and activation fragments. *J Biol Chem.* 1999;274(26):18635-18643.

105. Myles T, Yun TH, Hall SW, Leung LL. An extensive interaction interface between thrombin and factor V is required for factor V activation. *J Biol Chem.* 2001;276(27):25143-25149.

106. Myles T, Yun TH, Leung LL. Structural requirements for the activation of human factor VIII by thrombin. *Blood*. 2002;100(8):2820-2826.

107. Gerlitz B, Grinnell BW. Mutation of protease domain residues Lys37-39 in human protein C inhibits activation by the thrombomodulin-thrombin complex without affecting activation by free thrombin. *J Biol Chem.* 1996;271(37):22285-22288.

108. Knobe KE, Berntsdotter A, Shen L, et al. Probing the activation of protein C by the thrombin-thrombomodulin complex using structural analysis, site-directed mutagenesis, and computer modeling. *Proteins*. 1999;35(2):218-234.

109. Gale AJ, Griffin JH. Characterization of a thrombomodulin binding site on protein C and its comparison to an activated protein C binding site for factor Va. *Proteins*. 2004;54(3):433-441.

110. Taylor FB, Jr., Peer GT, Lockhart MS, Ferrell G, Esmon CT. Endothelial cell protein C receptor plays an important role in protein C activation in vivo. *Blood*. 2001;97(6):1685-1688.

111. Stearns-Kurosawa DJ, Kurosawa S, Mollica JS, Ferrell GL, Esmon CT. The endothelial cell protein C receptor augments protein C activation by the thrombin-thrombomodulin complex. *Proc Natl Acad Sci U S A*. 1996;93(19):10212-10216.

112. Fukudome K, Ye X, Tsuneyoshi N, et al. Activation mechanism of anticoagulant protein C in large blood vessels involving the endothelial cell protein C receptor. *J Exp Med.* 1998;187(7):1029-1035.

113. Xu J, Esmon NL, Esmon CT. Reconstitution of the human endothelial cell protein C receptor with thrombomodulin in phosphatidylcholine vesicles enhances protein C activation. *J Biol Chem.* 1999;274(10):6704-6710.

114. Dahlback B, Villoutreix BO. Regulation of blood coagulation by the protein C anticoagulant pathway: novel insights into structure-function relationships and molecular recognition. *Arterioscler Thromb Vasc Biol.* 2005;25(7):1311-1320.

115. Fay PJ, Mastri M, Koszelak ME, Wakabayashi H. Cleavage of factor VIII heavy chain is required for the functional interaction of a2 subunit with factor IXA. *J Biol Chem.* 2001;276(15):12434-12439.

116. Heeb MJ, Kojima Y, Hackeng TM, Griffin JH. Binding sites for blood coagulation factor Xa and protein S involving residues 493-506 in factor Va. *Protein Sci.* 1996;5(9):1883-1889.

117. Kalafatis M, Beck DO. Identification of a binding site for blood coagulation factor Xa on the heavy chain of factor Va. Amino acid residues 323-331 of factor V represent an interactive site for activated factor X. *Biochemistry*. 2002;41(42):12715-12728.

118. Singh LS, Bukys MA, Beck DO, Kalafatis M. Amino acids Glu323, Tyr324, Glu330, and Val331 of factor Va heavy chain are essential for expression of cofactor activity. *J Biol Chem.* 2003;278(30):28335-28345.

119. Dahlback B, Villoutreix BO. The anticoagulant protein C pathway. *FEBS Lett.* 2005;579(15):3310-3316.

120. Laffan MA. Activated protein C resistance and myocardial infarction. *Heart*. 1998;80(4):319-321.

121. Jenny RJ, Pittman DD, Toole JJ, et al. Complete cDNA and derived amino acid sequence of human factor V. *Proc Natl Acad Sci U S A*. 1987;84(14):4846-4850.

122. Kane WH, Davie EW. Cloning of a cDNA coding for human factor V, a blood coagulation factor homologous to factor VIII and ceruloplasmin. *Proc Natl Acad Sci U S A*. 1986;83(18):6800-6804.

123. Kane WH, Ichinose A, Hagen FS, Davie EW. Cloning of cDNAs coding for the heavy chain region and connecting region of human factor V, a blood coagulation factor with four types of internal repeats. *Biochemistry*. 1987;26(20):6508-6514.

124. Bakker HM, Tans G, Janssen-Claessen T, et al. The effect of phospholipids, calcium ions and protein S on rate constants of human factor Va inactivation by activated human protein C. *Eur J Biochem*. 1992;208(1):171-178.

125. Norstrom EA, Steen M, Tran S, Dahlback B. Importance of protein S and phospholipid for activated protein C-mediated cleavages in factor Va. *J Biol Chem.* 2003;278(27):24904-24911.

126. Smirnov MD, Esmon CT. Phosphatidylethanolamine incorporation into vesicles selectively enhances factor Va inactivation by activated protein C. *J Biol Chem.* 1994;269(2):816-819.

127. Walker FJ. Regulation of activated protein C by protein S. The role of phospholipid in factor Va inactivation. *J Biol Chem.* 1981;256(21):11128-11131.

128. Nicolaes GA, Tans G, Thomassen MC, et al. Peptide bond cleavages and loss of functional activity during inactivation of factor Va and factor VaR506Q by activated protein C. *J Biol Chem.* 1995;270(36):21158-21166.

129. Dirven RJ, Vos HL, Bertina RM. The R306G and R506Q mutations in coagulation Factor V reveals additional cleavage sites for Activated Protein C in the R313-R321 region and at R505. *Thromb Res.* 2010;125(5):444-450.

130. Tran S, Dahlback B. Novel APC-cleavage sites in FVa providing insights into mechanisms of action of APC and its cofactor protein S. *J Thromb Haemost*. 2009.

131. Kalafatis M, Bertina RM, Rand MD, Mann KG. Characterization of the molecular defect in factor VR506Q. *J Biol Chem.* 1995;270(8):4053-4057.

132. Kalafatis M, Rand MD, Mann KG. The mechanism of inactivation of human factor V and human factor Va by activated protein C. *J Biol Chem.* 1994;269(50):31869-31880.

133. Gale AJ, Xu X, Pellequer JL, Getzoff ED, Griffin JH. Interdomain engineered disulfide bond permitting elucidation of mechanisms of inactivation of coagulation factor Va by activated protein C. *Protein Sci.* 2002;11(9):2091-2101.

134. Guinto ER, Esmon CT. Loss of prothrombin and of factor Xa-factor Va interactions upon inactivation of factor Va by activated protein C. *J Biol Chem.* 1984;259(22):13986-13992.

135. Hockin MF, Cawthern KM, Kalafatis M, Mann KG. A model describing the inactivation of factor Va by APC: bond cleavage, fragment dissociation, and product inhibition. *Biochemistry*. 1999;38(21):6918-6934.

136. Nesheim ME, Canfield WM, Kisiel W, Mann KG. Studies of the capacity of factor Xa to protect factor Va from inactivation by activated protein C. *J Biol Chem.* 1982;257(3):1443-1447.

137. Norstrom EA, Tran S, Steen M, Dahlback B. Effects of factor Xa and protein S on the individual activated protein C-mediated cleavages of coagulation factor Va. *J Biol Chem.* 2006;281(42):31486-31494.

138. Rosing J, Hoekema L, Nicolaes GA, et al. Effects of protein S and factor Xa on peptide bond cleavages during inactivation of factor Va and factor VaR506Q by activated protein C. *J Biol Chem.* 1995;270(46):27852-27858.

139. Suzuki K, Stenflo J, Dahlback B, Teodorsson B. Inactivation of human coagulation factor V by activated protein C. *J Biol Chem.* 1983;258(3):1914-1920.

140. Smirnov MD, Safa O, Esmon NL, Esmon CT. Inhibition of activated protein C anticoagulant activity by prothrombin. *Blood*. 1999;94(11):3839-3846.

141. Tran S, Norstrom E, Dahlback B. Effects of prothrombin on the individual activated protein C-mediated cleavages of coagulation factor Va. *J Biol Chem.* 2008;283(11):6648-6655.

142. Yegneswaran S, Nguyen PM, Gale AJ, Griffin JH. Prothrombin amino terminal region helps protect coagulation factor Va from proteolytic inactivation by activated protein C. *Thromb Haemost.* 2009;101(1):55-61.

143. Fay PJ, Coumans JV, Walker FJ. von Willebrand factor mediates protection of factor VIII from activated protein C-catalyzed inactivation. *J Biol Chem.* 1991;266(4):2172-2177.

144. Weiss HJ, Sussman, II, Hoyer LW. Stabilization of factor VIII in plasma by the von Willebrand factor. Studies on posttransfusion and dissociated factor VIII and in patients with von Willebrand's disease. *J Clin Invest*. 1977;60(2):390-404.

145. Koedam JA, Meijers JC, Sixma JJ, Bouma BN. Inactivation of human factor VIII by activated protein C. Cofactor activity of protein S and protective effect of von Willebrand factor. *J Clin Invest*. 1988;82(4):1236-1243.

146. Hill-Eubanks DC, Parker CG, Lollar P. Differential proteolytic activation of factor VIII-von Willebrand factor complex by thrombin. *Proc Natl Acad Sci U S A*. 1989;86(17):6508-6512.

147. Vehar GA, Keyt B, Eaton D, et al. Structure of human factor VIII. *Nature*. 1984;312(5992):337-342.

148. Bihoreau N, Pin S, de Kersabiec AM, Vidot F, Fontaine-Aupart MP. Copper-atom identification in the active and inactive forms of plasma-derived FVIII and recombinant FVIII-delta II. *Eur J Biochem*. 1994;222(1):41-48.

149. Tagliavacca L, Moon N, Dunham WR, Kaufman RJ. Identification and functional requirement of Cu(I) and its ligands within coagulation factor VIII. *J Biol Chem.* 1997;272(43):27428-27434.

150. Wakabayashi H, Koszelak ME, Mastri M, Fay PJ. Metal ion-independent association of factor VIII subunits and the roles of calcium and copper ions for cofactor activity and inter-subunit affinity. *Biochemistry*. 2001;40(34):10293-10300.

151. Wakabayashi H, Schmidt KM, Fay PJ. Ca(2+) binding to both the heavy and light chains of factor VIII is required for cofactor activity. *Biochemistry*. 2002;41(26):8485-8492.

152. Wakabayashi H, Zhen Z, Schmidt KM, Fay PJ. Mn2+ binding to factor VIII subunits and its effect on cofactor activity. *Biochemistry*. 2003;42(1):145-153.

153. Lollar P, Parker CG. Subunit structure of thrombin-activated porcine factor VIII. *Biochemistry*. 1989;28(2):666-674.

154. Fay PJ, Haidaris PJ, Smudzin TM. Human factor VIIIa subunit structure. Reconstruction of factor VIIIa from the isolated A1/A3-C1-C2 dimer and A2 subunit. *J Biol Chem.* 1991;266(14):8957-8962.

155. Nogami K, Shima M, Nishiya K, et al. A novel mechanism of factor VIII protection by von Willebrand factor from activated protein C-catalyzed inactivation. *Blood*. 2002;99(11):3993-3998.

156. Vehar GA, Davie EW. Preparation and properties of bovine factor VIII (antihemophilic factor). *Biochemistry*. 1980;19(3):401-410.

157. Gale AJ, Cramer TJ, Rozenshteyn D, Cruz JR. Detailed mechanisms of the inactivation of factor VIIIa by activated protein C in the presence of its cofactors, protein S and factor V. *J Biol Chem.* 2008;283(24):16355-16362.

158. Varfaj F, Neuberg J, Jenkins PV, Wakabayashi H, Fay PJ. Role of P1 residues Arg336 and Arg562 in the activated-Protein-C-catalysed inactivation of Factor VIIIa. *Biochem J*. 2006;396(2):355-362.

159. Ploos van Amstel JK, van der Zanden AL, Bakker E, Reitsma PH, Bertina RM. Two genes homologous with human protein S cDNA are located on chromosome 3. *Thromb Haemost.* 1987;58(4):982-987.

160. Schmidel DK, Tatro AV, Phelps LG, Tomczak JA, Long GL. Organization of the human protein S genes. *Biochemistry*. 1990;29(34):7845-7852.

161. DiScipio RG, Davie EW. Characterization of protein S, a gamma-carboxyglutamic acid containing protein from bovine and human plasma. *Biochemistry*. 1979;18(5):899-904.

162. Koenen RR, Gomes L, Tans G, Rosing J, Hackeng TM. The Ser460Pro mutation in recombinant protein S Heerlen does not affect its APC-cofactor and APC-independent anticoagulant activities. *Thromb Haemost.* 2004;91(6):1105-1114.

163. Alhenc-Gelas M, Aubry ML, Aiach M. Contribution of platelet protein S (PS) to PS levels measured in plasma. *Nouv Rev Fr Hematol.* 1993;35(5):469-471.

164. Fair DS, Marlar RA, Levin EG. Human endothelial cells synthesize protein S. *Blood*. 1986;67(4):1168-1171.

165. Hoskins J, Norman DK, Beckmann RJ, Long GL. Cloning and characterization of human liver cDNA encoding a protein S precursor. *Proc Natl Acad Sci U S A*. 1987;84(2):349-353.

166. Stern D, Brett J, Harris K, Nawroth P. Participation of endothelial cells in the protein C-protein S anticoagulant pathway: the synthesis and release of protein S. *J Cell Biol.* 1986;102(5):1971-1978.

167. Bovill EG, Landesman MM, Busch SA, et al. Studies on the measurement of protein S in plasma. *Clin Chem.* 1991;37(10 Pt 1):1708-1714.

168. Rezende SM, Simmonds RE, Lane DA. Coagulation, inflammation, and apoptosis: different roles for protein S and the protein S-C4b binding protein complex. *Blood*. 2004;103(4):1192-1201.

169. Gershagen S, Fernlund P, Lundwall A. A cDNA coding for human sex hormone binding globulin. Homology to vitamin K-dependent protein S. *FEBS Lett.* 1987;220(1):129-135.

170. Lundwall A, Dackowski W, Cohen E, et al. Isolation and sequence of the cDNA for human protein S, a regulator of blood coagulation. *Proc Natl Acad Sci U S A*. 1986;83(18):6716-6720.

171. Hansson K, Stenflo J. Post-translational modifications in proteins involved in blood coagulation. *J Thromb Haemost.* 2005;3(12):2633-2648.

172. Sunnerhagen M, Forsen S, Hoffren AM, et al. Structure of the Ca(2+)-free Gla domain sheds light on membrane binding of blood coagulation proteins. *Nat Struct Biol.* 1995;2(6):504-509.

173. Brinkman HJ, Mertens K, van Mourik JA. Proteolytic cleavage of protein S during the hemostatic response. *J Thromb Haemost*. 2005;3(12):2712-2720.

174. Yegneswaran S, Hackeng TM, Dawson PE, Griffin JH. The thrombin-sensitive region of protein S mediates phospholipid-dependent interaction with factor Xa. *J Biol Chem.* 2008;283(48):33046-33052.

175. Chang GT, Aaldering L, Hackeng TM, et al. Construction and characterization of thrombin-resistant variants of recombinant human protein S. *Thromb Haemost.* 1994;72(5):693-697.

176. Heeb MJ, Griffin JH. Activated protein C-dependent and -independent anticoagulant activities of protein S have different structural requirements. *Blood Cells Mol Dis.* 2002;29(2):190-199.

177. Long GL, Lu D, Xie RL, Kalafatis M. Human protein S cleavage and inactivation by coagulation factor Xa. *J Biol Chem.* 1998;273(19):11521-11526.

178. Morboeuf O, Borgel D, Gaussem P, et al. Characterization of cleaved plasma protein S with a monoclonal antibody-based assay. *Thromb Haemost.* 2000;84(4):604-610.

179. Nelson RM, VanDusen WJ, Friedman PA, Long GL. beta-Hydroxyaspartic acid and beta-hydroxyasparagine residues in recombinant human protein S are not required for anticoagulant cofactor activity or for binding to C4b-binding protein. *J Biol Chem.* 1991;266(31):20586-20589.

180. Stenberg Y, Julenius K, Dahlqvist I, Drakenberg T, Stenflo J. Calcium-binding properties of the third and fourth epidermal-growth-factor-like modules in vitamin-K-dependent protein S. *Eur J Biochem.* 1997;248(1):163-170.

181. Persson KE, Stenflo J, Linse S, et al. Binding of calcium to anticoagulant protein S: role of the fourth EGF module. *Biochemistry*. 2006;45(35):10682-10689.

182. Baroni M, Mazzola G, Kaabache T, et al. Molecular bases of type II protein S deficiency: the I203-D204 deletion in the EGF4 domain alters GLA domain function. *J Thromb Haemost*. 2006;4(1):186-191.

183. Mille-Baker B, Rezende SM, Simmonds RE, et al. Deletion or replacement of the second EGF-like domain of protein S results in loss of APC cofactor activity. *Blood*. 2003;101(4):1416-1418.

184. He X, Shen L, Malmborg AC, et al. Binding site for C4b-binding protein in vitamin K-dependent protein S fully contained in carboxy-terminal laminin-G-type repeats. A study using recombinant factor IX-protein S chimeras and surface plasmon resonance. *Biochemistry*. 1997;36(12):3745-3754.

185. Saposnik B, Borgel D, Aiach M, Gandrille S. Functional properties of the sexhormone-binding globulin (SHBG)-like domain of the anticoagulant protein S. *Eur J Biochem.* 2003;270(3):545-555.

186. Chang GT, Maas BH, Ploos van Amstel HK, et al. Studies of the interaction between human protein S and human C4b-binding protein using deletion variants of recombinant human protein S. *Thromb Haemost*. 1994;71(4):461-467.

187. Dahlback B. Protein S and C4b-binding protein: components involved in the regulation of the protein C anticoagulant system. *Thromb Haemost*. 1991;66(1):49-61.

188. Griffin JH, Gruber A, Fernandez JA. Reevaluation of total, free, and bound protein S and C4b-binding protein levels in plasma anticoagulated with citrate or hirudin. *Blood.* 1992;79(12):3203-3211.

189. Heeb MJ, Radtke KP, Fernandez JA, Tonnu L. Plasma contains protein S monomers and multimers with similar direct anticoagulant activity. *J Thromb Haemost.* 2006;4(10):2215-2222.

190. Heeb MJ, Schuck P, Xu X. Protein S multimers and monomers each have direct anticoagulant activity. *J Thromb Haemost*. 2006;4(2):385-391.

191. Sere KM, Janssen MP, Willems GM, et al. Purified protein S contains multimeric forms with increased APC-independent anticoagulant activity. *Biochemistry*. 2001;40(30):8852-8860.

192. Sere KM, Willems GM, Rosing J, Hackeng TM. Protein S multimers are generated in vitro and affect protein S structure-function analyses. *Semin Hematol.* 2006;43(1 Suppl 1):S111-120.

193. Villoutreix BO, Teleman O, Dahlback B. A theoretical model for the Gla-TSR-EGF-1 region of the anticoagulant cofactor protein S: from biostructural pathology to species-specific cofactor activity. *J Comput Aided Mol Des.* 1997;11(3):293-304.

194. Rigby AC, Grant MA. Protein S: a conduit between anticoagulation and inflammation. *Crit Care Med*. 2004;32(5 Suppl):S336-341.

195. Webb JH, Blom AM, Dahlback B. The binding of protein S and the protein S-C4BP complex to neutrophils is apoptosis dependent. *Blood Coagul Fibrinolysis*. 2003;14(4):355-359.

196. Kask L, Trouw LA, Dahlback B, Blom AM. The C4b-binding Protein-Protein S Complex Inhibits the Phagocytosis of Apoptotic Cells. Vol. 279; 2004:23869-23873.

197. Liu D, Guo H, Griffin JH, Fernandez JA, Zlokovic BV. Protein S confers neuronal protection during ischemic/hypoxic injury in mice. *Circulation*. 2003;107(13):1791-1796.

198. Benzakour O, Formstone C, Rahman S, et al. Evidence for a protein S receptor(s) on human vascular smooth muscle cells. Analysis of the binding characteristics and mitogenic properties of protein S on human vascular smooth muscle cells. *Biochem J.* 1995;308 (Pt 2)481-485.

199. Kanthou C, Benzakour O. Cellular effects and signalling pathways activated by the anti-coagulant factor, protein S, in vascular cells protein S cellular effects. *Adv Exp Med Biol.* 2000;476155-166.

200. Trouw LA, Nilsson SC, Goncalves I, Landberg G, Blom AM. C4b-binding protein binds to necrotic cells and DNA, limiting DNA release and inhibiting complement activation. *J Exp Med*. 2005;201(12):1937-1948.

201. Lijfering WM, Mulder R, ten Kate MK, et al. Clinical relevance of decreased free protein S levels: results from a retrospective family cohort study involving 1143 relatives. *Blood.* 2009;113(6):1225-1230.

202. Gandrille S, Borgel D, Sala N, et al. Protein S deficiency: a database of mutations--summary of the first update. *Thromb Haemost.* 2000;84(5):918.

203. Makris M, Leach M, Beauchamp NJ, et al. Genetic analysis, phenotypic diagnosis, and risk of venous thrombosis in families with inherited deficiencies of protein S. *Blood*. 2000;95(6):1935-1941.

204. Koenen RR, Christella M, Thomassen LG, et al. Effect of oral contraceptives on the anticoagulant activity of protein S in plasma. *Thromb Haemost*. 2005;93(5):853-859.

205. Simmonds RE, Ireland H, Lane DA, et al. Clarification of the risk for venous thrombosis associated with hereditary protein S deficiency by investigation of a large kindred with a characterized gene defect. *Ann Intern Med.* 1998;128(1):8-14.

206. Burstyn-Cohen T, Heeb MJ, Lemke G. Lack of Protein S in mice causes embryonic lethal coagulopathy and vascular dysgenesis. *J Clin Invest*. 2009.

207. Saller F, Brisset AC, Tchaikovski SN, et al. Generation and phenotypic analysis of protein S-deficient mice. *Blood.* 2009;114(11):2307-2314.

208. Walker FJ. Regulation of activated protein C by a new protein. A possible function for bovine protein S. *J Biol Chem.* 1980;255(12):5521-5524.

209. Mitchell CA, Kelemen SM, Salem HH. The anticoagulant properties of a modified form of protein S. *Thromb Haemost*. 1988;60(2):298-304.

210. Hackeng TM, van 't Veer C, Meijers JC, Bouma BN. Human protein S inhibits prothrombinase complex activity on endothelial cells and platelets via direct interactions with factors Va and Xa. *J Biol Chem.* 1994;269(33):21051-21058.

211. Heeb MJ, Kojima Y, Rosing J, Tans G, Griffin JH. C-terminal Residues 621-635 of Protein S Are Essential for Binding to Factor Va. Vol. 274; 1999:36187-36192.

212. Heeb MJ, Mesters RM, Tans G, Rosing J, Griffin JH. Binding of protein S to factor Va associated with inhibition of prothrombinase that is independent of activated protein C. Vol. 268; 1993:2872-2877.

213. Stenberg Y, Muranyi A, Steen C, et al. EGF-like module pair 3-4 in vitamin K-dependent protein S: modulation of calcium affinity of module 4 by module 3, and interaction with factor X. *J Mol Biol*. 1999;293(3):653-665.

214. Heeb MJ, Rosing J, Bakker HM, et al. Protein S binds to and inhibits factor Xa. *Proc Natl Acad Sci U S A*. 1994;91(7):2728-2732.

215. Koppelman SJ, Hackeng TM, Sixma JJ, Bouma BN. Inhibition of the intrinsic factor X activating complex by protein S: evidence for a specific binding of protein S to factor VIII. *Blood.* 1995;86(3):1062-1071.

216. van 't Veer C, Butenas S, Golden NJ, Mann KG. Regulation of prothrombinase activity by protein S. *Thromb Haemost.* 1999;82(1):80-87.

217. van Wijnen M, Stam JG, van't Veer C, et al. The interaction of protein S with the phospholipid surface is essential for the activated protein C-independent activity of protein S. *Thromb Haemost.* 1996;76(3):397-403.

218. Sere KM, Rosing J, Hackeng TM. Inhibition of thrombin generation by protein S at low procoagulant stimuli: implications for maintenance of the hemostatic balance. *Blood*. 2004;104(12):3624-3630.

219. Solymoss S, Tucker MM, Tracy PB. Kinetics of inactivation of membrane-bound factor Va by activated protein C. Protein S modulates factor Xa protection. *J Biol Chem.* 1988;263(29):14884-14890.

220. Norstrom E, Thorelli E, Dahlback B. Functional characterization of recombinant FV Hong Kong and FV Cambridge. *Blood*. 2002;100(2):524-530.

221. Harris KW, Esmon CT. Protein S is required for bovine platelets to support activated protein C binding and activity. *J Biol Chem.* 1985;260(4):2007-2010.

222. Stern DM, Nawroth PP, Harris K, Esmon CT. Cultured bovine aortic endothelial cells promote activated protein C-protein S-mediated inactivation of factor Va. *J Biol Chem.* 1986;261(2):713-718.

223. Rezende SM, Lane DA, Mille-Baker B, et al. Protein S Gla-domain mutations causing impaired Ca(2+)-induced phospholipid binding and severe functional protein S deficiency. *Blood*. 2002;100(8):2812-2819.

224. Saller F, Villoutreix BO, Amelot A, et al. The gamma-carboxyglutamic acid domain of anticoagulant protein S is involved in activated protein C cofactor activity, independently of phospholipid binding. *Blood*. 2005;105(1):122-130.

225. Saller F, Kaabache T, Aiach M, Gandrille S, Borgel D. The protein S thrombinsensitive region modulates phospholipid binding and the gamma-carboxyglutamic acidrich (Gla) domain conformation in a non-specific manner. *J Thromb Haemost.* 2006;4(3):704-706.

226. Shen L, Shah AM, Dahlback B, Nelsestuen GL. Enhancement of human protein C function by site-directed mutagenesis of the gamma-carboxyglutamic acid domain. *J Biol Chem.* 1998;273(47):31086-31091.

227. Sun YH, Shen L, Dahlback B. Gla domain-mutated human protein C exhibiting enhanced anticoagulant activity and increased phospholipid binding. *Blood.* 2003;101(6):2277-2284.

228. Yegneswaran S, Smirnov MD, Safa O, et al. Relocating the active site of activated protein C eliminates the need for its protein S cofactor. A fluorescence resonance energy transfer study. *J Biol Chem.* 1999;274(9):5462-5468.

229. Yegneswaran S, Wood GM, Esmon CT, Johnson AE. Protein S alters the active site location of activated protein C above the membrane surface. A fluorescence resonance energy transfer study of topography. *J Biol Chem.* 1997;272(40):25013-25021.

230. Cramer TJ, Griffin JH, Gale AJ. Factor V Is an Anticoagulant Cofactor for Activated Protein C during Inactivation of Factor Va. *Pathophysiol Haemost Thromb.* 2010;37(1):17-23.

231. Maurissen LF, Thomassen MC, Nicolaes GA, et al. Re-evaluation of the role of the protein S-C4b binding protein complex in activated protein C-catalyzed factor Vainactivation. *Blood.* 2008;111(6):3034-3041.

232. O'Brien LM, Mastri M, Fay PJ. Regulation of factor VIIIa by human activated protein C and protein S: inactivation of cofactor in the intrinsic factor Xase. *Blood.* 2000;95(5):1714-1720.

233. Varadi K, Rosing J, Tans G, et al. Factor V enhances the cofactor function of protein S in the APC-mediated inactivation of factor VIII: influence of the factor VR506Q mutation. *Thromb Haemost*. 1996;76(2):208-214.

234. Shen L, Dahlback B. Factor V and protein S as synergistic cofactors to activated protein C in degradation of factor VIIIa. *J Biol Chem.* 1994;269(29):18735-18738.

235. van de Poel RH, Meijers JC, Bouma BN. C4b-binding protein inhibits the factor Vdependent but not the factor V-independent cofactor activity of protein S in the activated protein C-mediated inactivation of factor VIIIa. *Thromb Haemost*. 2001;85(5):761-765.

236. Takeyama M, Nogami K, Saenko EL, et al. Protein S down-regulates factor Xase activity independent of activated protein C: specific binding of factor VIII(a) to protein S inhibits interactions with factor IXa. *Br J Haematol.* 2008;143(3):409-420.

237. Ndonwi M, Broze G, Jr. Protein S enhances the tissue factor pathway inhibitor inhibition of factor Xa but not its inhibition of factor VIIa-tissue factor. *J Thromb Haemost.* 2008;6(6):1044-1046.

238. Ndonwi M, Tuley EA, Broze GJ, Jr. The Kunitz-3 domain of TFPI-alpha is required for protein S-dependent enhancement of factor Xa inhibition. *Blood.* 2010;116(8):1344-1351.

239. Dahm AE, Sandset PM, Rosendaal FR. The association between protein S levels and anticoagulant activity of tissue factor pathway inhibitor type 1. *J Thromb Haemost*. 2008;6(2):393-395.

240. Maurissen LF, Castoldi E, Simioni P, Rosing J, Hackeng TM. Thrombin generation-based assays to measure the activity of the TFPI-protein S pathway in plasma from normal and protein S-deficient individuals. *J Thromb Haemost*. 2010;8(4):750-758.

241. Castoldi E, Simioni P, Tormene D, Rosing J, Hackeng TM. Hereditary and acquired protein S deficiencies are associated with low TFPI levels in plasma. *J Thromb Haemost*. 2010;8(2):294-300.

242. Heeb MJ, Prashun D, Griffin JH, Bouma BN. Plasma protein S contains zinc essential for efficient activated protein C-independent anticoagulant activity and binding to factor Xa, but not for efficient binding to tissue factor pathway inhibitor. *FASEB J*. 2009;23(7):2244-2253.

243. Dahlback B, Hildebrand B, Malm J. Characterization of functionally important domains in human vitamin K-dependent protein S using monoclonal antibodies. *J Biol Chem.* 1990;265(14):8127-8135.

244. He X, Shen L, Villoutreix BO, Dahlback B. Amino acid residues in thrombinsensitive region and first epidermal growth factor domain of vitamin K-dependent protein S determining specificity of the activated protein C cofactor function. *J Biol Chem.* 1998;273(42):27449-27458.

245. Stenberg Y, Drakenberg T, Dahlback B, Stenflo J. Characterization of recombinant epidermal growth factor (EGF)-like modules from vitamin-K-dependent protein S expressed in Spodoptera cells--the cofactor activity depends on the N-terminal EGF module in human protein S. *Eur J Biochem.* 1998;251(3):558-564.

246. Hackeng TM, Yegneswaran S, Johnson AE, Griffin JH. Conformational changes in activated protein C caused by binding of the first epidermal growth factor-like module of protein S. *Biochem J*. 2000;349 Pt 3757-764.

247. Rezende SM, Razzari C, Simmonds RE. In vitro high level protein S expression after modification of protein S cDNA. *Thromb Haemost.* 2003;90(6):1214-1215.

248. Lorand L, Mann KG. Proteolytic enzymes in coagulation, fibrinolysis, and complement activation. Part A, Mammalian blood coagulation factors and inhibitors. San Diego: Academic Press; 1993.

249. Morrison JF, Walsh CT. The behavior and significance of slow-binding enzyme inhibitors. *Adv Enzymol Relat Areas Mol Biol*. 1988;61201-301.

250. Komoriya A, Hortsch M, Meyers C, et al. Biologically active synthetic fragments of epidermal growth factor: localization of a major receptor-binding region. *Proc Natl Acad Sci U S A*. 1984;81(5):1351-1355.

251. Pauls JE, Hockin MF, Long GL, Mann KG. Self-association of human protein S. *Biochemistry*. 2000;39(18):5468-5473.

252. Dahlback B. Purification of human vitamin K-dependent protein S and its limited proteolysis by thrombin. *Biochem J.* 1983;209(3):837-846.

253. Brodin E, Appelbom H, Osterud B, et al. Regulation of thrombin generation by TFPI in plasma without and with heparin. *Transl Res.* 2009;153(3):124-131.

254. Machlus KR, Colby EA, Wu JR, et al. Effects of tissue factor, thrombomodulin and elevated clotting factor levels on thrombin generation in the calibrated automated thrombogram. *Thromb Haemost*. 2009;102(5):936-944.

255. Giri TK, Garcia de Frutos P, Dahlback B. Protein S Thr103Asn mutation associated with type II deficiency reproduced in vitro and functionally characterised. *Thromb Haemost.* 2000;84(3):413-419.

256. Hayashi T, Nishioka J, Shigekiyo T, Saito S, Suzuki K. Protein S Tokushima: abnormal molecule with a substitution of Glu for Lys-155 in the second epidermal growth factor-like domain of protein S. *Blood*. 1994;83(3):683-690.

257. Tsuda H, Urata M, Tsuda T, et al. Four missense mutations identified in the protein S gene of thrombosis patients with protein S deficiency: effects on secretion and anticoagulant activity of protein S. *Thromb Res.* 2002;105(3):233-239.

258. Stenflo J, Stenberg Y, Muranyi A. Calcium-binding EGF-like modules in coagulation proteinases: function of the calcium ion in module interactions. *Biochim Biophys Acta*. 2000;1477(1-2):51-63.

259. Mori Y, Hamuro T, Nakashima T, et al. Biochemical characterization of plasmaderived tissue factor pathway inhibitor: post-translational modification of free, full-length form with particular reference to the sugar chain. *J Thromb Haemost*. 2009;7(1):111-120.

260. Giri TK, Villoutreix BO, Wallqvist A, Dahlback B, de Frutos PG. Topological studies of the amino terminal modules of vitamin K-dependent protein S using monoclonal antibody epitope mapping and molecular modeling. *Thromb Haemost.* 1998;80(5):798-804.

261. Hackeng TM, Rosing J. Protein S as cofactor for TFPI. *Arterioscler Thromb Vasc Biol.* 2009;29(12):2015-2020.

262. Duchemin J, Pan-Petesch B, Arnaud B, Blouch MT, Abgrall JF. Influence of coagulation factors and tissue factor concentration on the thrombin generation test in plasma. *Thromb Haemost*. 2008;99(4):767-773.

263. Duckers C, Simioni P, Spiezia L, et al. Low plasma levels of tissue factor pathway inhibitor in patients with congenital factor V deficiency. Vol. 112; 2008:3615-3623.

264. De Smedt E, Wagenvoord R, Coen Hemker H. The technique of measuring thrombin generation with fluorogenic substrates: 3. The effects of sample dilution. *Thromb Haemost*. 2009;101(1):165-170.

265. Dielis AW, Castoldi E, Spronk HM, et al. Coagulation factors and the protein C system as determinants of thrombin generation in a normal population. *J Thromb Haemost*. 2008;6(1):125-131.

266. Hackeng TM, Fernandez JA, Dawson PE, Kent SB, Griffin JH. Chemical synthesis and spontaneous folding of a multidomain protein: anticoagulant microprotein S. *Proc Natl Acad Sci U S A*. 2000;97(26):14074-14078.

267. Qureshi SH, Yang L, Manithody C, Bae JS, Rezaie AR. Functional properties and active-site topographies of factor X Gla- and prothrombin Gla-domain chimeras of activated protein C. *Biochim Biophys Acta*. 2008;1780(9):1080-1086.

268. Harmon S, Preston RJS, Ainle FN, et al. Dissociation of Activated Protein C Functions by Elimination of Protein S Cofactor Enhancement. Vol. 283; 2008:30531-30539.

269. Morrissey JH, Pureza V, Davis-Harrison RL, et al. Protein-membrane interactions: blood clotting on nanoscale bilayers. *J Thromb Haemost*. 2009;7 Suppl 1169-172.

## **PUBLICATIONS ARISING FROM THIS WORK**

**Helena M Andersson**, Márcia J Arantes, James TB Crawley, Brenda M Luken, Sinh Tran, Professor Björn Dahlbäck, David A Lane, Suely M Rezende. Activated protein C cofactor function of protein S: a critical role for Asp95 in the EGF1-like domain. Blood. 2010 Jun 10;115(23):4878-85.

Josefin Ahnström, **Helena M Andersson**, Kevin Canis, Eva Norstrøm, Professor Björn Dahlbäck, Maria Panico, Howard R Morris, James TB Crawley and David A Lane.

Activated protein C cofactor function of protein S: a novel function for  $\gamma$ carboxyglutamic acid residue 36. Manuscript in preparation.

# **ABSTRACTS LEADING TO ORAL PRESENTATION**

**Helena Andersson**, Verity Hockey, Yao Yu, S Mapara, Brenda M Luken, James TB Crawley, David A Lane.

The molecular interaction of TFPI with protein S.

BSHT and UKHCDO annual scientific meeting 1-3<sup>rd</sup> October 2008. Nottingham.

**Helena Andersson**, Márcia J Arantes, James TB Crawley, Brenda M Luken, David A Lane, Suely M Rezende.

Activated protein C cofactor function of protein S: a critical role for the EGF1-like domain

Awarded the young investigator award, XXII congress of the ISTH meeting 11-16<sup>th</sup> July 2009. Boston.

**Helena M Andersson**, Márcia J Arantes, James TB Crawley, Brenda M Luken, Suely M Rezende and David A Lane.

EGF1 residue Asp95 in protein S is essential for APC cofactor activity.

Awarded young scientist in training award, BSHT and UKHCDO annual scientific meeting 7-9<sup>rd</sup> October 2009. Newcastle.

Verity Hockey, **Helena Andersson**, Thomas AJ McKinnon, Tsutomu Hamuro, David A Lane and James TB Crawley.

TFPI inhibition of Factor Xa and its enhancement by protein S.

NVTH/BSHT joint symposium 23-25<sup>th</sup> June 2010. Noordwijkerhout.

Josefin Ahnström, **Helena M Andersson**, Kevin Canis, Yao Yu, Maria Panico, Howard R Morris, James TB Crawley and David A Lane.

Glutamic acid residue 36 of protein S and activated protein C cofactor function NVTH/BSHT joint symposium 23-25<sup>th</sup> June 2010. Noordwijkerhout.

# Activated protein C cofactor function of protein S: a critical role for Asp95 in the EGF1-like domain

Helena M. Andersson,<sup>1</sup> Márcia J. Arantes,<sup>2</sup> James T. B. Crawley,<sup>1</sup> Brenda M. Luken,<sup>1</sup> Sinh Tran,<sup>3</sup> Björn Dahlbäck,<sup>3</sup> David A. Lane,<sup>1</sup> and Suely M. Rezende<sup>4</sup>

<sup>1</sup>Department of Haematology, Faculty of Medicine, Imperial College London, London, United Kingdom; <sup>2</sup>Núcleo de Açcões e Pesquisa em Apoio Diagnóstico (NUPAD), Faculdade de Medicina, Universidade Federal de Minas Gerais, Minas Gerais, Brazil; <sup>3</sup>Department of Laboratory Medicine, Division of Clinical Chemistry, The Wallenberg Laboratory, Lund University, Malmö, Sweden; and <sup>4</sup>Departamento de Clínica Médica, Faculdade de Medicina, Universidade Federal de Minas Gerais, Minas Gerais, Brazil

Protein S has an established role in the protein C anticoagulant pathway, where it enhances the factor Va (FVa) and factor VIIIa (FVIIIa) inactivating property of activated protein C (APC). Despite its physiological role and clinical importance, the molecular basis of its action is not fully understood. To clarify the mechanism of the protein S interaction with APC, we have constructed and expressed a library of composite or point variants of human protein S, with residue substitutions introduced into the Gla, thrombin-sensitive region (TSR), epidermal growth factor 1 (EGF1), and EGF2 domains. Cofactor activity for APC was evaluated by calibrated automated thrombography (CAT) using protein S-deficient plasma. Of 27 variants tested initially, only one, protein S D95A (within the EGF1 domain), was largely devoid of functional APC cofactor activity. Protein S D95A was, however,  $\gamma$ -carboxylated and bound phospholipids with an apparent dissociation

constant (Kd<sub>app</sub>) similar to that of wildtype (WT) protein S. In a purified assay using FVa R506Q/R679Q, purified protein S D95A was shown to have greatly reduced ability to enhance APC-induced cleavage of FVa Arg306. It is concluded that residue Asp95 within EGF1 is critical for APC cofactor function of protein S and could define a principal functional interaction site for APC. (*Blood.* 2010;115(23): 4878-4885)

#### Introduction

Protein S, a vitamin K–dependent plasma anticoagulant protein, functions as an enhancing cofactor to activated protein C (APC) in the inactivation of activated factors V (FVa) and VIII (FVIIIa).<sup>1</sup> Protein S also has an APC-independent activity that has recently been attributed to its ability to enhance tissue factor pathway inhibitor (TFPI).<sup>2,3</sup> Protein S has an important role in vivo, as is shown clinically by infants with complete deficiency, who suffer purpura fulminans, and by heterozygous carriers of *PROS1* gene deletions and point mutations, who are at enhanced risk of venous thromboembolism.<sup>4,5</sup> The importance of protein S has also been demonstrated in murine knockouts, which fail to survive development.<sup>6,7</sup>

Protein S binds phospholipids and helps to localize APC to the membrane surface, in proximity to FVa and FVIIIa. In doing so, rather than the preferential cleavage at Arg506 of FVa by APC, Arg306 is then favored, with cleavage enhanced approximately 20-fold.<sup>8</sup> Although this effect has generally been accepted to be due to protein S repositioning of the active site of APC,<sup>9</sup> this has recently been questioned.<sup>10</sup> Cleavage at Arg506 and Arg306 fully inactivates FVa, thereby efficiently down-regulating coagulation.<sup>11</sup> APC also inactivates FVIIIa by cleaving first at Arg336 and thereafter at Arg562. In the presence of protein S, it has been shown that the efficiency of cleavage after Arg336 by APC is enhanced approximately 3-fold.<sup>12</sup>

Protein S is a 635–amino acid glycoprotein, circulating at a plasma concentration of approximately 350nM. It is composed of an N-terminal Gla domain (amino acids 1-45), a thrombinsensitive region (TSR; amino acids 46-75), 4 epidermal growth factor (EGF)–like domains (EGF1, amino acids 76-116; EGF2, amino acids 203-242), and a C-terminal sex hormone–binding globulin (SHBG)–like domain (amino acids 243-635).<sup>13,14</sup> Approximately 60% of protein S circulates as a noncovalent, high-affinity complex with  $\beta$ -chain containing C4b binding protein.<sup>15</sup> This interaction is mediated through the SHBG domain of protein S. Although it has been generally understood that the free form (~ 40%) of protein S alone is important in anticoagulant function, this has recently been questioned.<sup>16</sup>

Whereas the functions of the domains of protein S have been extensively studied, the APC cofactor function of protein S remains poorly understood at the molecular level. Several basic and clinical studies have confirmed that alteration of protein S can disrupt phospholipid binding and thereby impair function of protein S.<sup>17,18</sup> Saller et al prepared protein S/prothrombin chimeras and, on the basis of functional studies supported by molecular modeling, identified a cluster of protein S Gla domain residues (termed Face2), which may provide an APC interaction site.<sup>19</sup>

The publication costs of this article were defrayed in part by page charge payment. Therefore, and solely to indicate this fact, this article is hereby marked "advertisement" in accordance with 18 USC section 1734.

© 2010 by The American Society of Hematology

Submitted November 26, 2009; accepted February 20, 2010. Prepublished online as *Blood* First Edition paper, March 22, 2010; DOI 10.1182/blood-2009-11-256610.

Presented in part in abstract form at the International Society on Thrombosis and Haemostasis, Boston, MA, July 14, 2009; at the British Society for Haemostasis and Thrombosis, Newcastle, United Kingdom, October 7, 2009;

and at the Congresso Brasileiro de Hematologia e Hemoterapia, Florianópolis, Brazil, November 11, 2009.

Table 1. Protein S variants generated by site-directed mutagenesis

GLA1	D38N/V46G/K97Q
GLA2/Face2	L21T/N23S/K24Y/R28F/D34S/Y41W/L45T
TSR1	R49Q/R60Q/D68N/R70Q/D78N
EGF1 single-residue variants	K94A, D95A, K97A, S99A, T101A, T103A, K105A, P106A, W108A, Q109A, E111A, K112A
EGF2 single-residue variants	D135A, N136A, T137A, Y141A, H142A, S144A, K146A, S153A, N154A, K155A, K156A, D157A

Additional studies have focused on the possible role of TSR and EGF domains. Several studies have reported that cleavage in the TSR of protein S affects its APC cofactor activity.<sup>20-22</sup> Dahlbäck et al<sup>23</sup> and Giri et al<sup>24</sup> used domain-specific monoclonal antibodies directed against protein S and their Fab fragments and suggested that the TSR and EGF1 domains may be involved in APC cofactor activity. The TSR and EGF1 domains of protein S have also been reported to be responsible for species specificity.<sup>25</sup> Using EGF-like, modules Stenberg et al concluded the EGF1 module to be crucial for interaction with APC.26 Hackeng et al demonstrated that the isolated EGF1 domain of protein S could inhibit full-length protein S anticoagulant enhancing function against APC: direct binding of the domain to APC was also demonstrated.<sup>27</sup> Mille-Baker et al prepared EGF2 domain deletion and substitution mutants of protein S and demonstrated reduced anticoagulant activity, in the presence of normal phospholipid binding.<sup>28</sup> Collectively, these investigators have suggested Gla, TSR, EGF1, and EGF2 domains may each contain recognition elements on protein S needed for APC cofactor function.

Because these diverse approaches have not yet fully defined the functional interaction site(s) on protein S for APC, we have investigated the role of individual and groups of residue substitutions in Gla, TSR, EGF1, and EGF2 domains of protein S. Using a library of 27 protein S variants, we performed thrombin generation assays in protein S–deficient plasma to which APC and these protein S variants were added. We report here that EGF1 has a critical role in anticoagulant function of protein S, provided specifically by Asp95, the mutation of which almost completely abolishes APC cofactor activity.

#### Methods

#### Generation and expression of protein S and its variants

A previous reported pcDNA6/protein S vector containing the cDNA for wild-type (WT) protein S<sup>29</sup> was used to generate recombinant full-length WT protein S and as a template to produce protein S mutants by polymerase chain reaction site-directed mutagenesis (QuikChange XL Site-Directed Mutagenesis Kit; Stratagene) using mutagenic oligonucleotide primers (Thermo Scientific). Single and composite residue variants produced are shown in Table 1.

Residues to mutate were selected based on nonconserved residues in domains of other human coagulation factors (Gla mutants) and charge (TSR mutant). Residues in the B and C regions (between the third/fourth and fifth/sixth Cys residues, respectively) of both EGFs were mutated, as these regions on the EGF-like domains are likely to be involved in protein interaction as suggested by (1) the highly divergent sequence of the B region,<sup>30</sup> (2) the association of dysfunctional proteins with natural *PROS1* mutations in these regions,<sup>4</sup> and (3) the location in these regions of many mutations associated with dysfunctional proteins in the other coagulation genes coding for EGF domains, such as FVII, FIX, and protein C.

WT and mutant protein S expression vectors were either transiently or stably transfected into HEK293T or HEK293 cells (ATCC), respectively.

Stably transfected cells were selected with 5 µg/mL Blasticidin-HCl (Invitrogen) for 4 weeks. Protein S was expressed in OptiMem I (Invitrogen) with 10 µg/mL vitamin K (Roche Products Ltd) for 3 days. Media containing protein S was harvested, centrifuged, filtered, dialyzed in 20 mM Tris (tris(hydroxymethyl)aminomethane)–HCl (pH 7.5), 140 mM NaCl (TBS), and concentrated, as required. All variants were successfully expressed into the media and could be readily detected by sodium dodecyl sulfate–polyacrylamide gel electrophoresis (SDS-PAGE) followed by Western blotting and by enzyme-linked immunosorbent assay (ELISA). We have previously reported that the protein S expressed in this way is functionally active in APC cofactor assays and can be fully  $\gamma$ -carboxylated.<sup>17,31</sup>

#### **Purification of protein S**

WT protein S and selected variants were either partially purified by barium citrate precipitation as previously described<sup>32</sup> for use in phospholipid binding analysis or extensively purified by chromatography for use in APC-dependent functional assays. Chromatography was performed using an ÄKTA purifier UPC-10 and the Unicorn 5.11 (Build 407) software, strategy Version 1.00 (both from GE Healthcare). Concentrated and dialyzed conditioned media containing protein S supplemented with 4 mM EDTA (ethylenediaminetetraacetic acid) was applied to an anionic Q Sepharose Fast Flow (5 mL of HiTrap QFF) column (GE Healthcare) equilibrated with 20mM Tris-HCl (pH 7.4), 100mM NaCl, 4mM EDTA, and 5mM benzamidine-HCl (Sigma-Aldrich). The column was washed with 20mM Tris-HCl (pH 7.4), 100mM NaCl, and 5mM benzamidine-HCl. Protein S was eluted from the column with 20mM Tris-HCl (pH 7.4), 0.5M NaCl, and 5mM benzamidine-HCl.

Two volumes of 7.5mM CaCl<sub>2</sub> were added to the elution fraction from the anionic QFF column to lower the ionic strength, before injecting the fraction onto an affinity column prepared by coupling 2.5 mg of monoclonal antibody against protein S, MK21, to a 1-mL HiTrap NHS HP column (GE Healthcare), according to the manufacturer's instructions. All monoclonal antibodies against protein S used in this study are described in the paper of Dahlbäck et al.23 The protein S-containing fraction was applied to the equilibrated column (equilibration buffer: 50mM Tris-HCl [pH 7.5], 150mM NaCl, 2mM CaCl<sub>2</sub>). After washing with 1M NaCl, the ionic strength was lowered by passing additional 3 column volumes of equilibration buffer over the column prior to elution of protein S with 50mM Tris-HCl (pH 7.5), 150mM NaCl, and 10mM EDTA. An excess of CaCl<sub>2</sub> was present in the tubes collecting the eluted protein S to neutralize the EDTA. The column was regenerated by stripping with 0.1M glycine-HCl (pH 2.7). Purified protein S was dialyzed in TBS and concentrated, as required.

#### Determination of protein S concentration by ELISA

Protein S concentrations in media were determined by an in-house ELISA. Polyclonal rabbit anti-protein S antibody (1  $\mu$ g/mL; DAKO) was immobilized in a 96-well Nunc Maxisorp microplate in 50mM sodium carbonate buffer (pH 9.6) at 4°C overnight. Washing steps were performed in triplicate with 250  $\mu$ L of phosphate-buffered saline, 0.1% Tween between each step. Wells were blocked with phosphate-buffered saline containing 3% bovine serum albumin (BSA; Sigma-Aldrich) for 2 hours. Bound protein S was detected by a mouse monoclonal antibody directed against the EGF3-4 domain of protein S, MK55, followed by a horseradish peroxidase (HRP)–conjugated goat anti–mouse antibody (DAKO). The plate was developed with 100  $\mu$ L/well chromogenic substrate O-phenylene diamide dihydrochloride; OPD (Sigma-Aldrich), the enzymatic reaction was stopped with 50  $\mu$ L/well 2M H<sub>2</sub>SO<sub>4</sub>, and the absorbance was read at 492 nm.

#### SDS-PAGE and Western blotting

SDS-PAGE and Western blotting were performed using standard techniques. Briefly, 4% to 12% or 10% precast NuPAGE Novex Bis-Tris gels (Invitrogen) were used. For silver staining and Western blotting, standard techniques were used. The SeeBlue-prestained marker from Invitrogen was used. Protein S was detected either by a polyclonal rabbit anti–protein S antibody followed by an HRP-conjugated goat anti–rabbit antibody or by a monoclonal mouse antibody recognizing only  $\gamma$ -carboxylated Glu residues (American Diagnostica Inc), followed by a goat anti–mouse HRP-conjugated antibody (DAKO).

#### Phospholipid vesicle preparation

Phospholipids mixtures (Avanti Polar Lipids Inc) in chloroform were prepared and the chloroform was evaporated under a nitrogen stream. The phospholipids were resuspended in TBS or 25mM *N*-2-hydroxyethylpiperazine-*N'*-2-ethanesulfonic acid (pH 7.7), 150mM NaCl (HN) for the plasma and purified assays, respectively. Unilamellar phospholipids vesicles were obtained by sonicating the phospholipids in an ice container at amplitude 22% for 7 minutes. Extruded phospholipids were prepared as previously described<sup>33</sup> and used in the FVa inactivation assay. Synthetic phospholipids 1,2-Dioleoyl-*sn*-glycero-3-phosphocholine (DOPC), 1,2-Dioleoyl-*sn*glycero-3-phosphoserine (DOPS), and 1,2-Dioleoyl-*sn*-glycero-3-phosphoethanolamine (DOPE) were used in the plasma assay and the plate-binding assay. Natural phospholipids L- $\alpha$ -phosphatidylserine (PS; brain extract), L- $\alpha$ -phosphatidylethanolamine (PE; egg extract), and L- $\alpha$ -phosphatidylcholine (PC; egg extract) were used in the FVa inactivation and prothrombinase assays.

#### Binding of protein S to phospholipids

Phospholipids vesicles (DOPS/DOPC/DOPE, 20:60:20), 25  $\mu$ g/mL, were coated as described for the ELISA. Washing steps were performed in triplicate with 250  $\mu$ L/well TBS, 5mM CaCl<sub>2</sub>, 0.3% BSA between each step and incubations were carried out in a plate shaker at 37°C. Wells were blocked with TBS, 5mM CaCl<sub>2</sub>, 3% BSA for 2 hours. Protein S partially purified by barium citrate precipitation, 0 to 120nM, was incubated in the wells in duplicate for 2 hours and was detected with an HRP-conjugated anti–protein S antibody for 45 minutes (Affinity Biologicals). Data were analyzed with GraphPad Prism 4.03 and curves were fitted to a one-site binding hyperbola.

#### Binding of protein S to domain-specific monoclonal antibodies

Domain-specific monoclonal antibodies MK21 (Gla domain), MK54 (EGF1 domain), and MK61 (SHBG domain), 1  $\mu$ g/mL, were coated as described for the ELISA. Washing steps were performed in triplicate with 250  $\mu$ L of TBS, 3mM CaCl<sub>2</sub>, 0.1% Tween between each step. Incubations were carried out in a plate shaker at 37°C for 1 hour unless stated otherwise. Wells were blocked with 200  $\mu$ L of TBS, 3mM CaCl<sub>2</sub>, 3% BSA for 2 hours. Protein S, 0 to 160nM was incubated in the plate for 1 hour and was detected by a rabbit polyclonal antibody against protein S followed by a goat anti–rabbit HRP-conjugated antibody.

## Thrombin generation assay by calibrated automated thrombography

Thrombin generation was assessed using a Fluoroscan Ascent FL Plate Reader (Thermo Lab System) in combination with Thrombinoscope software (SYNAPSE BV). Protein S-deficient plasma (Affinity Biologicals), 80 µL, was incubated with 65 µg of corn trypsin inhibitor (Haematologic Technologies Inc) per milliliter of plasma to inhibit contact activation, 50µM phospholipid vesicles (DOPS/DOPC/DOPE, 20:60:20), 1pM tissue factor (Dade Innovin; Dade Behring), 4 to 16nM APC (Enzyme Research Laboratories Ltd) with 0 to 120nM protein S, in a final volume of 100 µL (all concentrations are final). A polyclonal antibody against TFPI (Haematologic Technologies Inc), 100nM, was used to inhibit any protein S cofactor activity toward TFPI. Although protein S cofactor activity toward TFPI was not observed at low concentrations of APC, it was observed to a small extent when lower amounts of thrombin were generated (eg, at higher concentrations of APC in the presence of protein S). Polyclonal antibodies against protein S and protein C were used to show the specificity of the system and were from, respectively, DAKO and Sigma-Aldrich. Thrombin generation was initiated by automatic dispensation of 20 µL of 2.5mM Z-GlyArg-AMC-HCl (Bachem), 2.5% Me<sub>2</sub>SO, 20mM Tris-HCl (pH 7.5),

60 mg/mL BSA,  $100 \text{mM CaCl}_2$  into each well. The reaction was performed at  $37^{\circ}$ C and measurements were performed with an excitation and emission wavelength of 390 nm and 460 nm, respectively.

# Protein S-dependent APC-mediated factor FVa inactivation assay

A previously prepared and reported<sup>34</sup> double mutant of FV, FV R506Q/ R679Q, was activated with 0.5 U/mL human thrombin at 37°C for 10 minutes. The reaction was stopped by the addition of hirudin (5 U/mL, final concentration). The activated FVa variant was used as a substrate for APC to determine the ability of protein S to enhance cleavage of Arg306 of FVa by APC. The highly purified protein S preparations, 0 to 100nM, were incubated with 0.5nM APC, 25µM phospholipids vesicles (PS/PC/PE, 10:70:20), and 0.8nM FVa R506Q/R679Q in HN, 5mM CaCl<sub>2</sub>, 5 mg/mL BSA (HNBSACa<sup>2+</sup>) in a total volume of 50 µL (all concentrations are final). The solution was incubated at 37°C for 10 minutes, and the reaction was stopped by performing a 1:25-fold dilution in ice-cold HNBSACa<sup>2+</sup>. The remaining FVa activity was measured in a prothrombinase assay.

Time course experiments were also performed to derive the rate constant of FVa cleavage. APC, 0.5 and 3nM, and 50nM protein S were then used and aliquots were quenched at intervals between 0 and 20 minutes. The remaining FVa activity was again measured in a prothrombinase assay. To calculate the pseudo–first-order rate constant for APC-mediated cleavage at Arg306, the previously reported equation<sup>33</sup> was used:  $Va_t = Va_0 \times e^{-k306 \times t} + C \times Va_0 \times (1 - e^{-k306 \times t})$ .

 $Va_t$  represents the cofactor activity determined at time t,  $Va_0$  is the cofactor activity determined at time point 0, C is the remaining procoagulant cofactor activity of FVa after cleavage at position 306, and k306 is the rate constant of cleavage at position 306.

#### Prothrombinase assay

An aliquot,  $25 \ \mu$ L, of the FVa inactivation reaction was incubated with phospholipid vesicles (PS/PC 10:90) and FXa in the presence of CaCl<sub>2</sub>. The reaction was initiated by the addition of prothrombin, in a final volume of 125  $\mu$ L. The buffer contained HN and 0.5 mg/mL ovalbumin. Final concentrations were 50 $\mu$ M PS/PC, 5nM FXa, and 0.5 $\mu$ M prothrombin. The solution was incubated at 37°C for 2 minutes and the reaction was terminated by an 8-fold dilution in 50mM Tris-HCl (pH 7.9), 100mM NaCl, 20mM EDTA, 1% PEG 6000. The amount of thrombin generated was measured by cleavage of its chromogenic substrate S-2238 (Chromogenix) at 405 nm for 15 minutes at 30-second intervals.

#### Results

# Screening of protein S mutants for APC cofactor activity in plasma

Concentrated and dialyzed conditioned media samples containing WT protein S and the 27 protein S variants were normalized with respect to protein S concentration, using the protein S–specific ELISA. Western blotting revealed single-band protein S in all samples (nonreducing conditions).

To assess the APC cofactor activity of protein S, thrombin generation was assessed by calibrated automated thrombography (CAT) using protein S–deficient plasma supplemented with recombinant WT protein S and its variants in conditioned media. Under the conditions of the assay, APC (0-10nM) did not affect peak, endogenous thrombin potential (ETP), lag time, or time to peak, when coagulation was initiated in protein S–deficient plasma (Figure 1A). Concentrated conditioned media from cells not expressing protein S had no influence on thrombin generation (data not shown). When 120nM WT protein S was progressively introduced into the assay, dose-dependent effects were observed, Figure 1. Effect of APC and protein S on thrombin generation. Thrombin generation was performed in protein S-deficient plasma with 100nM inhibitory antibodies against TFPI. Up to 10nM APC had no effect on thrombin generation in the absence of protein S. All concentrations generate lines that are superimposable (A). After addition of 120nM protein S (at 0-10nM APC), an APC dose-dependent effect was observed (B). The top single line represents 0 to 10nM APC in the absence of protein S. Protein S in the presence of no or 2.5nM APC generated lines that were superimposable. Conditions used are noted adjacent to the peaks to which they refer. The anticoagulant effect of 10nM APC and 120nM protein S was inhibited by polyclonal antibodies against protein S (C) or against protein C (D). PS indicates protein S; PC, protein C.



primarily on reduction of the peak and of ETP values (Figure 1B). The extent of inhibition of peak and ETP was dependent both on the concentration of APC and protein S. The higher the APC concentration, the less protein S was necessary to completely inhibit thrombin generation. In these experimental conditions, 10nM APC and 120nM protein S reduced both the peak and ETP by approximately 84%. To verify that those effects were a direct consequence of APC cofactor activity of protein S, polyclonal antibodies against either protein S (Figure 1C) or protein C (Figure 1D) were used. These inhibited the anticoagulant APC–protein S effect.

The screening of the protein S variants for their APC cofactor activity was performed at high concentration of APC (16nM) to achieve almost complete inhibition of thrombin generation with



Figure 2. Screening of protein S variants for APC cofactor activity. The APC cofactor activity of protein S was evaluated at 16nM APC and 100nM protein S by CAT. The peak height in the absence of protein S was set to 100%. A high concentration of APC, leading to almost complete inhibition of thrombin generation with 100nM WT protein S, was chosen specifically for screening purposes as this allows widening of the assay window at which mutants with reduced APC cofactor activity are visualized. Results were confirmed by evaluating protein S (at 60 and 90nM) cofactor activity toward 4 or 9nM APC.

100nM WT protein S (Figure 2). These screening experiments identified a single point mutant in EGF1 of protein S, D95A, which had a severely impaired APC cofactor activity. Under all conditions used, the reduction in cofactor activity was greater than that observed for the previously reported variant with 7 amino acids exchanged (prothrombin residue swap), GLA2/Face2.

# Evaluation of the importance of Asp95 for APC cofactor activity of protein S in plasma

To further evaluate the APC cofactor activity of protein S D95A, a titration (0-120nM) of the mutant (Figure 3B) was performed alongside WT protein S (Figure 3A), using 9nM APC. The protein S D95A had a severely impaired APC cofactor activity, a finding replicated when different concentrations of APC were used (results not shown). In these experimental conditions, whereas 120nM WT protein S reduced peak and ETP by approximately 80%, 120nM protein S D95A reduced peak and ETP by only approximately 10%. To confirm the importance of Asp95 in protein S APC cofactor function, Asp95 was also substituted with Asn, rather than Ala. Asn is structurally more similar to Asp than Ala and it can also be β-hydroxylated. When the protein S D95N mutant was titrated, it was observed that it also had severely reduced APC cofactor activity (Figure 3C) with 120nM D95N inhibiting peak and ETP by approximately 20%. All results in Figure 3A through C were conducted with protein S and its variants contained in concentrated conditioned media. To confirm that they were not influenced by the media, WT protein S and the protein S D95A variant were purified to homogeneity using anion exchange and affinity chromatography (see Figure 3D inset for SDS-PAGE gel visualized with silver staining). Addition of 90nM purified WT protein S to protein S-deficient plasma in the presence of 9nM APC appreciably attenuated thrombin generation (Figure 3D), replicating the findings with WT protein S in conditioned media. Moreover, addition



**Figure 3. Effect of WT protein S, D95A, D95N, D78A, and Q79A variants on thrombin generation.** Thrombin generation was measured in protein S-deficient plasma supplemented with 9nM APC, 100nM antibodies against TFPI, and 0 to 120nM WT protein S (A), protein S D95A (B), protein S D95N (C), or 90nM purified WT (dashed line) or purified protein S D95A (dotted line; D). Protein S concentrations are positioned adjacent to the peaks to which they refer. The cofactor activity of 60nM WT protein S and protein S variants D95A, D78A, and Q79A was compared at 9nM APC (E). Typical experiments are shown (n = 3). Whereas the cofactor activity of WT protein S is highly dependent on the APC concentration used (Figure 1B), that of protein S D95A is not, explaining the difference in fold activity between WT protein S and protein S D95A in Figures 2 and 3. Dose-response data from titrations with WT protein S, protein S D95A, and protein S D95N in the presence of 9nM APC are shown in panel F (data are expressed as mean ± S0 of 2 independent experiments performed in duplicate). Inset in panel B shows recognition of WT protein S and protein S D95A in media by polyclonal antibodies and a ± S0 of 2 lindependent experiments performed in duplicate). Inset in panel D shows the SeeBlue-prestained marker, plasma-purified protein S from Enzyme Research Laboratories Ltd (lane 1), purified recombinant WT protein S (lane 2), and purified protein S D95A (lane 3) visualized with silver staining.

of 90nM purified protein S D95A to APC produced only a minimal effect on thrombin generation (Figure 3D).

When analyzing a model<sup>35</sup> of Gla-TSR-EGF1 domains of protein S, 2 residues with a solvent-exposed R residue in close proximity to that of Asp95 were identified, Asp78 and Gln79. These were also mutated to alanine and expressed. When protein S D78A and Q79A were analyzed in the thrombin generation assay alongside WT protein S, a severely reduced APC cofactor activity was also observed (Figure 3E). This further confirms the importance of this region of EGF1 in APC cofactor activity of protein S. Dose-response experiments of WT protein S, protein S D95A, and protein S D95N, shown in Figure 3F, confirm reduced APC cofactor function of these variants.

#### Phospholipid binding of WT protein S and protein S D95A

As phospholipid binding is a prerequisite for protein S function, the binding of WT protein S and of protein S D95A to phospholipids was evaluated. Partially purified (using barium citrate precipitation) protein S, 0 to 120nM, was incubated for 2 hours on a plate coated with 25  $\mu$ g/mL phospholipids and was detected by an HRP-conjugated antibody against protein S. As shown in Figure 4, both WT protein S and protein S D95A were able to bind phospholipids, with apparent dissociation constant (Kd<sub>app</sub>) of 5.69 plus or minus 1.24 and 9.54 plus or minus 2.26nM, respectively

(Table 2). Kd<sub>app</sub> values were analyzed by the Mann-Whitney test and were found not to be statistically different (P > .05). Binding was observed in the presence of Ca<sup>2+</sup>, but not in the presence of EDTA or in the absence of phospholipids, as expected. These



Figure 4. Binding of protein S to phospholipid surfaces. Protein S (0-120nM) was incubated in a plate coated with 25  $\mu$ g/mL phospholipids. Bound protein S was detected with an HRP-conjugated polyclonal antibody against protein S. A representative experiment is shown. The apparent Kd values, 5.69  $\pm$  1.24 and 9.54  $\pm$  2.26nM for WT protein S and protein S D95A, respectively, were obtained by calculating the mean  $\pm$  SD of 3 independent experiments performed in duplicate. PL indicates phospholipids.

Table 2. Binding	g of pro	otein S to	phospholipids	and domain-spec	ific monoclona	l antibodies
------------------	----------	------------	---------------	-----------------	----------------	--------------

	Phospholipid vesicles	MK21 (Gla)	MK54 (EGF1)	MK61 (SHBG)
WT protein S	5.69 ± 1.24	2.18 ± 0.97	0.81 ± 0.11	5.31 ± 1.11
Asp95 protein S mutant	9.54 ± 2.26	$2.53 \pm 1.06$	$0.83\pm0.03$	$5.39\pm0.54$

Kd<sub>app</sub> values (nM) of WT protein S and the Asp95 protein S mutant for phospholipid vesicles and conformational domain-specific monoclonal antibodies are expressed as mean plus or minus SD of 3 independent experiments performed in duplicate. Binding of WT protein S and the Asp95 mutant to phospholipids and domain-specific monoclonal antibodies was analyzed by Mann-Whitney test; results were not statistically significant (*P* > .05).

findings are broadly consistent with previously reported Kd values for protein S binding to phospholipid surfaces.<sup>17-19,36,37</sup> Our results suggest the loss of APC cofactor activity observed for protein S D95A is not due to loss of binding to phospholipid surfaces.

#### Binding to domain-specific monoclonal antibodies

Binding of protein S to conformational domain-specific antibodies was evaluated to assess the integrity of the domain structure and folding. Protein S in concentrated conditioned media was incubated in a plate coated with monoclonal antibodies recognizing either the Gla domain (MK21), the EGF1 domain (MK54), or the C-terminal SHBG domain (MK61). Bound protein S was detected with polyclonal antibodies. Binding curves were fitted with a one-site binding equation, and the Kd<sub>app</sub> values were obtained. Kd<sub>app</sub> values were analyzed by the Mann-Whitney test and were found not to be statistically different (P > .05). Table 2 represents the mean plus or minus SD Kd<sub>app</sub> values of 3 independent experiments performed in duplicate. The results suggest that mutation of Asp95 does not result in a significant change in the domain structure of protein S.

## Protein S enhancement of APC-mediated cleavage of FVa at Arg306

The FVa variant, R506Q/R679Q, was used to evaluate protein S–enhanced APC-mediated cleavage at Arg306 in FVa. FVa inactivation was performed in the presence or absence of 0.5nM APC and a titration of purified protein S (0-100nM) was performed. After 10 minutes, the remaining FVa activity was measured in the prothrombinase assay. WT protein S efficiently enhanced APC-mediated cleavage at Arg306 of FVa, in contrast to the D95A mutant (Figure 5A).

To quantify the rate of cleavage at FVa Arg306 in the presence and absence of protein S, time course experiments were performed (Figure 5B). In the absence of protein S, or in the presence of protein S D95A, 3nM APC was used and aliquots were quenched at different time points (0-20 minutes). In the presence of WT protein S, the APC concentration was lowered to 0.5nM, as APC is efficiently enhanced by protein S. The remaining FVa activity at all time points was measured in the prothrombinase assay. Using the FVa inactivation curves obtained, the apparent pseudo-first-order rate constants were calculated and corrected for the APC concentrations used. Under these experimental conditions, protein S had no effect on FVa activity in the absence of APC (data not shown). APC-mediated cleavage at FVa Arg306 was enhanced by WT protein S by 13.9-fold plus or minus 3.6-fold, whereas protein S D95A was only able to enhance APC by 1.8-fold plus or minus 0.4-fold.

#### Discussion

We initially constructed and expressed 27 protein S variants with mutations in the Gla, TSR, EGF1, and EGF2 domains and

evaluated their ability to enhance APC-mediated anticoagulant activity in plasma. The anticoagulant function of these 27 variants, and the subsequently expressed variants D78A, D79A, and D95N, were assessed by a thrombin generation assay, the specificity of which was demonstrated using polyclonal antibodies against either protein S or protein C that completely inhibited the anticoagulant response observed when adding protein S in the presence of APC. The advantage of the thrombin generation assay conducted in plasma is that it shows very clearly how APC is heavily dependent on protein S.<sup>38</sup> In the absence of protein S, 10nM APC has no effect on thrombin generation. In contrast, in the presence of protein S, maximal anticoagulant effects with near ablation of thrombin generation are obtainable with APC concentrations less than 10nM.

Our initial screening results identified 2 protein S variants with appreciable reduction in APC cofactor function. These were the already reported GLA2/Face2 variant (a 7-residue composite mutant in the Gla domain) and the novel D95A point variant in the EGF1 domain. The former variant, reported by Saller et al, retains binding to phospholipids and yet has appreciably reduced APC cofactor activity.<sup>19</sup> It is suggested that the 7 residues substituted in this variant collectively present a face of the Gla domain to APC and form a potential contact region. It was selected for investigation here because it is a well-characterized protein S variant with



Figure 5. Protein S enhancement of APC-mediated cleavage of FVa in Arg306. Protein S (0-120nM) in the presence of 0.5nM APC was incubated with 0.8nM FVa R506Q/R679Q in the presence of phospholipids for 10 minutes. The remaining FVa activity was measured with a prothrombinase assay. Results are plotted as mean  $\pm$  SD from 3 independent experiments performed in duplicate (A). A time course experiment was performed to calculate the apparent pseudo-first-order rate constants of WT protein S and protein S D95A. It is observed that approximately 6-fold more APC is needed in the presence of protein S D95A to obtain a similar amount of APC-mediated FVa R506Q/R679Q inactivation as with WT protein S (B).



Figure 6. Location of Asp78, GIn79, and Asp95 within the protein S Gla-TSR-EGF1 model. Domains are labeled on the right side of the model. Residues mutated in the GLA2 variant, Asp78, GIn79, and Asp95, are in light gray on the left side surface model. Residues Asp78, GIn79, and Asp95 are highlighted by the box to show their proximal spatial location. The model is taken from Villoutreix et al.<sup>35</sup>

substantially reduced APC cofactor activity and could therefore act as a representative control for a dysfunctional protein S. Saller et al,19 however, did not assess GLA2/Face2 APC cofactor activity in the CAT assay. We can confirm a reduction in APC cofactor activity in plasma, assessed by CAT. The important novel results here, however, concerned the protein S D95A variant, which was almost devoid of APC cofactor activity in plasma. The natural protein S variants T103N<sup>39</sup> and K155E<sup>40,41</sup> have been reported to have reduced APC cofactor activity. We could not observe any significant reduction in APC cofactor activity of protein S T103A in the CAT assay. It is, however, important to point out that defective APC cofactor activity of protein S T103N was not observed in the APC-mediated inactivation of WT-FVa.39 In addition, at higher concentrations of APC (8nM), no difference was observed between WT protein S and protein S T103N in the FVIIIa degradation assay.<sup>39</sup> We did, however, observe a decrease in K155A cofactor activity toward APC, although this was quite moderate under conditions used in the screening assay.

Analysis of protein S function is complicated by its propensity to multimerize upon purification.<sup>42,43</sup> Accordingly, we have performed APC cofactor assays with unpurified protein S in concentrated conditioned media as well as with fully purified preparations. In the event, our results were all consistent. Both forms of the protein S D95A variant, purified or in concentrated conditioned media, had an appreciable reduction in APC cofactor activity compared with the respective WT protein S preparation. Although it has not been formally assessed, in all comparison experiments carried out, the protein S D95A variant had less APC cofactor than the GLA2/Face2 variant. This suggests that Asp95 in protein S may occupy a pivotal position with respect to its interaction with APC. The importance of Asp95 in APC cofactor activity was further confirmed by the severely reduced activity observed when mutating to alanine the 2 residues in close proximity to Asp95, Asp78, and Gln79. Asp78, Gln79, and Asp95 are conserved across species with the exception of Asp78, which is replaced by the structurally similar Asn78 in birds. Using a working model of the Gla-TSR-EGF1 domains of protein S, we demonstrate the likely proximal spatial location of these 3 residues (Figure 6) and their relationship to GLA2/Face2 residues.

We have investigated the mechanism of reduction of activity of protein S Asp95 variants. Whereas EGF1 of protein S has not been shown to contain a calcium-binding site (in contrast to EGF2, EGF3, and EGF4),<sup>44</sup> Asp95 could be part of a partially conserved calcium-binding motif. We therefore mutated Asp95 to Asn rather than Ala. Asn is structurally similar to Asp, can be  $\beta$ -hydroxylated, and is able to coordinate a calcium ion (as has been observed in EGF2, EGF3, and

EGF4). As the protein S D95N variant also had reduced activity, however, it is unlikely that a calcium-binding site is disrupted. Furthermore, whereas Asp95 is known to be  $\beta$ -hydroxylated, it has previously been shown that  $\beta$ -hydroxylation itself is not a requirement for anticoagulant activity of protein S.<sup>45</sup> Our evidence favors a direct effect of Asp95 in APC cofactor activity. Alternatively, the dramatic effect on protein S activity of the Asp95 residue substitution could be through a conformational repositioning of other functional domains. To assess this, we performed binding to phospholipids, as this property underpins all protein S function. Plate binding assays, however, indicated no major functional defect on phospholipid binding. Furthermore, domainspecific monoclonal antibody binding to WT and mutant protein S (against Gla domain, EGF1 domain, and the C-terminal SHBG domain) was also normal, suggesting that substitution of Asp95 does not reduce APC cofactor function by disrupting adjacent domain structure.

A current view of protein S–enhanced APC cofactor activity suggests a functional repositioning of the APC cleavage site away from FVa Arg506 toward FVa Arg306.<sup>9</sup> We therefore performed a FVa inactivation assay using a FVa variant, FVa R506Q/R679Q, that cannot be cleaved at position 506 and 679. This allowed us to analyze directly cleavage at Arg306 by APC, which is at the site enhanced mainly by protein S. Using both concentration-dependent and time course assays, we were able to confirm reduced cleavage of this variant by APC in the presence of protein S D95A.

The available results therefore suggest that Asp95 of protein S may play an important and a direct role in APC recognition, resulting in enhanced APC function. Our results are compatible with the study performed by Hackeng et al,<sup>27</sup> who used the isolated EGF1 domain of protein S to functionally disrupt the protein S and APC interaction. They suggested EGF1 as a potentially important APC contact site on protein S. Our results suggest that Asp95 constitutes a critical residue within EGF1, mediating the APC cofactor function. Together with Asp78 and Gln79, Asp95 could form the principal functional interaction site for APC.

#### Acknowledgments

The authors are grateful to Dr José Renan da Cunha Melo, Dr José Nélio Januário, and Dr Marcos Vinícius Andrade for their kind support on the sharing of cloning and tissue culture room. We are also grateful to Jamil Silvano de Oliveira for his helpful advice on purification techniques.

This work was supported by the International Research Development Award from Wellcome Trust, British Heart Foundation (PG/07/005), an unrestricted educational grant from Novartis, a Swedish council grant (07143), and Fundação de Amparo à Pesquisa do Estado de Minas Gerais (PPM 2007).

#### Authorship

Contribution: H.M.A., D.A.L., S.M.R., B.D., and J.T.B.C. designed the research, analyzed the results, and wrote the paper; and H.M.A., M.J.A., S.M.R., B.M.L., and S.T. performed the experiments.

Conflict-of-interest disclosure: The authors declare no competing financial interests.

Correspondence: David Lane, Department of Haematology, Faculty of Medicine, Imperial College London, Hammersmith Hospital Campus, Du Cane Rd, London W12 0NN, United Kingdom; e-mail: d.lane@imperial.ac.uk.

#### References

- Dahlbäck B, Villoutreix BO. The anticoagulant protein C pathway. FEBS Lett. 2005;579(15): 3310-3316.
- Hackeng TM, Sere KM, Tans G, Rosing J. Protein S stimulates inhibition of the tissue factor pathway by tissue factor pathway inhibitor. *Proc Natl Acad Sci U S A*. 2006;103(9):3106-3111.
- Ndonwi M, Broze G Jr. Protein S enhances the tissue factor pathway inhibitor inhibition of factor Xa but not its inhibition of factor VIIa-tissue factor. *J Thromb Haemost*. 2008;6(6):1044-1046.
- Gandrille S, Borgel D, Sala N, et al. Protein S deficiency: a database of mutations: summary of the first update. *Thromb Haemost.* 2000;84(5):918.
- Makris M, Leach M, Beauchamp NJ, et al. Genetic analysis, phenotypic diagnosis, and risk of venous thrombosis in families with inherited deficiencies of protein S. *Blood*. 2000;95(6):1935-1941.
- Burstyn-Cohen T, Heeb MJ, Lemke G. Lack of Protein S in mice causes embryonic lethal coagulopathy and vascular dysgenesis. *J Clin Invest.* 2009;119(10):2942-2953.
- Saller F, Brisset AC, Tchaikovski SN, et al. Generation and phenotypic analysis of protein Sdeficient mice. *Blood*. 2009;114(11):2307-2314.
- Rosing J, Hoekema L, Nicolaes GA, et al. Effects of protein S and factor Xa on peptide bond cleavages during inactivation of factor Va and factor VaR506Q by activated protein C. J Biol Chem. 1995;270(46):27852-27858.
- Yegneswaran S, Smirnov MD, Safa O, et al. Relocating the active site of activated protein C eliminates the need for its protein S cofactor: a fluorescence resonance energy transfer study. *J Biol Chem.* 1999;274(9):5462-5468.
- Tran S, Dahlback B. Novel APC-cleavage sites in FVa providing insights into mechanisms of action of APC and its cofactor protein S. *J Thromb Haemost.* 2010;8(1):129-136.
- Mann KG, Hockin MF, Begin KJ, Kalafatis M. Activated protein C cleavage of factor Va leads to dissociation of the A2 domain. *J Biol Chem.* 1997; 272(33):20678-20683.
- O'Brien LM, Mastri M, Fay PJ. Regulation of factor VIIIa by human activated protein C and protein S: inactivation of cofactor in the intrinsic factor Xase. *Blood.* 2000;95(5):1714-1720.
- Gershagen S, Fernlund P, Lundwall A. A cDNA coding for human sex hormone binding globulin. Homology to vitamin K-dependent protein S. *FEBS Lett.* 1987;220(1):129-135.
- Lundwall A, Dackowski W, Cohen E, et al. Isolation and sequence of the cDNA for human protein S, a regulator of blood coagulation. *Proc Natl Acad Sci U S A*. 1986;83(18):6716-6720.
- Rezende SM, Simmonds RE, Lane DA. Coagulation, inflammation, and apoptosis: different roles for protein S and the protein S-C4b binding protein complex. *Blood*. 2004;103(4):1192-1201.
- Maurissen LF, Thomassen MC, Nicolaes GA, et al. Re-evaluation of the role of the protein S-C4b binding protein complex in activated protein Ccatalyzed factor Va-inactivation. *Blood.* 2008; 111(6):3034-3041.
- 17. Rezende SM, Lane DA, Mille-Baker B, et al. Protein S Gla-domain mutations causing impaired

Ca(2+)-induced phospholipid binding and severe functional protein S deficiency. *Blood.* 2002; 100(8):2812-2819.

- Baroni M, Mazzola G, Kaabache T, et al. Molecular bases of type II protein S deficiency: the I203-D204 deletion in the EGF4 domain alters GLA domain function. J Thromb Haemost. 2006;4(1): 186-191.
- Saller F, Villoutreix BO, Amelot A, et al. The gamma-carboxyglutamic acid domain of anticoagulant protein S is involved in activated protein C cofactor activity, independently of phospholipid binding. *Blood*. 2005;105(1):122-130.
- Chang GT, Aaldering L, Hackeng TM, et al. Construction and characterization of thrombinresistant variants of recombinant human protein S. *Thromb Haemost.* 1994;72(5):693-697.
- Heeb MJ, Griffin JH. Activated protein C-dependent and -independent anticoagulant activities of protein S have different structural requirements. *Blood Cells Mol Dis.* 2002;29(2):190-199.
- Long GL, Lu D, Xie RL, Kalafatis M. Human protein S cleavage and inactivation by coagulation factor Xa. *J Biol Chem.* 1998;273(19):11521-11526.
- Dahlbäck B, Hildebrand B, Malm J. Characterization of functionally important domains in human vitamin K-dependent protein S using monoclonal antibodies. J Biol Chem. 1990;265(14):8127-8135.
- Giri TK, Villoutreix BO, Wallqvist A, Dahlback B, de Frutos PG. Topological studies of the amino terminal modules of vitamin K-dependent protein S using monoclonal antibody epitope mapping and molecular modeling. *Thromb Haemost*. 1998; 80(5):798-804.
- He X, Shen L, Villoutreix BO, Dahlback B. Amino acid residues in thrombin-sensitive region and first epidermal growth factor domain of vitamin K-dependent protein S determining specificity of the activated protein C cofactor function. J Biol Chem. 1998;273(42):27449-27458.
- Stenberg Y, Drakenberg T, Dahlback B, Stenflo J. Characterization of recombinant epidermal growth factor (EGF)-like modules from vitamin-Kdependent protein S expressed in Spodoptera cells: the cofactor activity depends on the Nterminal EGF module in human protein S. *Eur J Biochem.* 1998;251(3):558-564.
- Hackeng TM, Yegneswaran S, Johnson AE, Griffin JH. Conformational changes in activated protein C caused by binding of the first epidermal growth factor-like module of protein S. *Biochem* J. 2000;349(pt 3):757-764.
- Mille-Baker B, Rezende SM, Simmonds RE, et al. Deletion or replacement of the second EGF-like domain of protein S results in loss of APC cofactor activity. *Blood.* 2003;101(4):1416-1418.
- Rezende SM, Razzari C, Simmonds RE. In vitro high level protein S expression after modification of protein S cDNA. *Thromb Haemost.* 2003;90(6): 1214-1215.
- Komoriya A, Hortsch M, Meyers C, et al. Biologically active synthetic fragments of epidermal growth factor: localization of a major receptorbinding region. *Proc Natl Acad Sci U S A*. 1984; 81(5):1351-1355.
- 31. Persson KE, Stenflo J, Linse S, et al. Binding of

calcium to anticoagulant protein S: role of the fourth EGF module. *Biochemistry*. 2006;45(35): 10682-10689.

- Lorand L, Mann KG. Proteolytic Enzymes in Coagulation, Fibrinolysis, and Complement Activation, Part A: Mammalian Blood Coagulation Factors and Inhibitors. San Diego, CA: Academic Press; 1993.
- Tran S, Norstrom E, Dahlback B. Effects of prothrombin on the individual activated protein Cmediated cleavages of coagulation factor Va. *J Biol Chem.* 2008;283(11):6648-6655.
- Norstrøm EA, Steen M, Tran S, Dahlback B. Importance of protein S and phospholipid for activated protein C-mediated cleavages in factor Va. *J Biol Chem.* 2003;278(27):24904-24911.
- Villoutreix BO, Teleman O, Dahlback B. A theoretical model for the Gla-TSR-EGF-1 region of the anticoagulant cofactor protein S: from biostructural pathology to species-specific cofactor activity. J Comput Aided Mol Des. 1997;11(3): 293-304.
- Saller F, Kaabache T, Aiach M, Gandrille S, Borgel D. The protein S thrombin-sensitive region modulates phospholipid binding and the gammacarboxyglutamic acid-rich (Gla) domain conformation in a non-specific manner. J Thromb Haemost. 2006;4(3):704-706.
- Heeb MJ, Schuck P, Xu X. Protein S multimers and monomers each have direct anticoagulant activity. J Thromb Haemost. 2006;4(2):385-391.
- Seré KM, Rosing J, Hackeng TM. Inhibition of thrombin generation by protein S at low procoagulant stimuli: implications for maintenance of the hemostatic balance. *Blood*. 2004;104(12): 3624-3630.
- Giri TK, Garcia de Frutos P, Dahlback B. Protein S Thr103Asn mutation associated with type II deficiency reproduced in vitro and functionally characterised. *Thromb Haemost*. 2000;84(3):413-419.
- Hayashi T, Nishioka J, Shigekiyo T, Saito S, Suzuki K. Protein S Tokushima: abnormal molecule with a substitution of Glu for Lys-155 in the second epidermal growth factor-like domain of protein S. *Blood.* 1994;83(3):683-690.
- Tsuda H, Urata M, Tsuda T, et al. Four missense mutations identified in the protein S gene of thrombosis patients with protein S deficiency: effects on secretion and anticoagulant activity of protein S. *Thromb Res.* 2002;105(3):233-239.
- Seré KM, Janssen MP, Willems GM, et al. Purified protein S contains multimeric forms with increased APC-independent anticoagulant activity. *Biochemistry*. 2001;40(30):8852-8860.
- Seré KM, Willems GM, Rosing J, Hackeng TM. Protein S multimers are generated in vitro and affect protein S structure-function analyses. Semin Hematol. 2006;43(1 suppl 1):S111-S120.
- Stenflo J, Stenberg Y, Muranyi A. Calcium-binding EGF-like modules in coagulation proteinases: function of the calcium ion in module interactions. *Biochim Biophys Acta*. 2000;1477(1-2):51-63.
- Nelson RM, VanDusen WJ, Friedman PA, Long GL. Beta-Hydroxyaspartic acid and betahydroxyasparagine residues in recombinant human protein S are not required for anticoagulant cofactor activity or for binding to C4b-binding protein. J Biol Chem. 1991;266(31):20586-20589.